UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): January 20, 2012

LEE ENTERPRISES, INCORPORATED

(Exact name of Registrant as specified in its charter)

Commission File Number 1-6227

Delaware (State of Incorporation)

42-0823980 (I.R.S. Employer Identification No.)

201 N. Harrison Street, Davenport, Iowa 52801 (Address of Principal Executive Offices)

(563) 383-2100 Registrant's telephone number, including area code

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

Drovisions:	
1 White a second string amount to Bull 425 and but to Compile A + (47 CFD 220 425)	

- [] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- [] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- [] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- [] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.03. Bankruptcy or Receivership.

As previously disclosed on December 12, 2011 (the "Petition Date"), Lee Enterprises, Incorporated (the "Company") and certain of its subsidiaries (together with the Company, the "Debtors") filed voluntary petitions in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court") for relief under Chapter 11 of Title 11 of the United States Code (the "Bankruptcy Code" and the "Ch. 11 Proceedings"). The Debtors also filed on the Petition Date their Amended Joint Prepackaged Plan of Reorganization dated December 2, 2011 and the Debtors' Amended Solicitation and Disclosure Statement dated December 2, 2011 (the "Disclosure Statement"). On January 19, 2012, the Debtors filed the Second Amended Joint Prepackaged Plan of Reorganization for Lee Enterprises, Incorporated and its Debtor Subsidiaries (the "Plan"), which reflected certain non-material modifications to the plan of reorganization filed on the Petition Date.

On January 23, 2012 (the "Confirmation Date"), the Bankruptcy Court held a hearing and entered an order (the "Confirmation Order"), confirming the Plan. The Debtors expect to emerge from the Ch. 11 Proceedings after satisfying the remaining conditions to effectiveness contemplated under the Plan (the date that all conditions to the effectiveness of the Plan have been satisfied or waived, which is expected to be January 30, 2012, the "Effective Date"). Copies of the Confirmation Order and Plan are filed as **Exhibits 2.1** and **2.2** hereto, respectively, and are incorporated herein by reference.

The following is a summary of the material features of the Plan, as confirmed by the Bankruptcy Court. The following summary highlights only certain provisions of the Plan and is not a complete description of that document. This summary is qualified in its entirety by reference to the Plan. Capitalized terms used but not defined in this Form 8-K have the meanings set forth in the Plan. The Plan provides that on the Effective Date:

- The Company will enter into agreements to amend and extend its prepetition borrowings under its revolving credit facility and A Term Loan Facility beyond their current maturity of April 2012 in a structure of first and second lien debt. The first lien debt consists of a term loan of \$689.5 million, as well as a new \$40 million revolving credit facility that is not expected to be drawn at closing, both of which mature in December 2015. The first lien debt carries interest at LIBOR plus 6.25%, with a LIBOR floor of 1.25%. Interest on the \$40 million revolver is at LIBOR plus 5.5%, with a LIBOR floor of 1.25%. Mandatory amortization payments for the first lien debt total \$5 million for the remainder of the Company's 2012 fiscal year, \$11 million in 2013, \$12.75 million in 2014, and \$13.5 million prior to the final maturity in 2015. In addition, the first lien debt requires principal payments based on calculated excess cash flow and allows for optional repayments.
- The second lien debt consists of a \$175 million term loan with an interest rate of 15% maturing in April 2017. There are no scheduled mandatory amortization payments required, except as may be required in 2017 for income tax purposes. Second lien creditors will receive their pro rata share of the Company's issuance of 6,743,640 shares of its Common Stock, par value \$0.01 per share, amounting to approximately 13% of outstanding shares on a pro forma basis as of the Effective Date.
- The Company will enter into agreements regarding its remaining prepetition debt, the Pulitzer Notes, to extend their maturities to December 2015. The debt will carry an interest rate of 10.55%, increasing 0.75% in January 2013 and each year thereafter. Adjusted for principal payments of \$5.0 million and non-cash fees in the amount of \$3.5 million to be paid to Noteholders, the amended Pulitzer Notes will have a balance of \$126.4 million. Mandatory amortization payments total \$1.4 million in 2012 and \$6.4 million annually thereafter prior to the final maturity in December 2015. The Pulitzer Notes also require principal payments based on calculated excess cash flow and allow for optional repayments. Each Holder

of a Claim arising under the prepetition Pulitzer Notes will receive its pro rata share of new Pulitzer Notes issued by the Company's indirect subsidiary, St. Louis Post-Dispatch LLC.

- The Company will adopt an Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation") and Amended and Restated By-Laws. The Company's Certificate of Incorporation will reduce the par value of the Company's Common Stock from \$2.00 per share to \$0.01 per share.
- The Company's Board of Directors will have the same eleven (11) directors who were serving as directors on the Petition Date. Also, the Company expects its management to consist of its officers as of the Petition Date.
- All of the Debtors' Executory Contracts, except certain exceptions specified in the Plan, will be deemed assumed as of the Effective
 Date, including but not limited to, the Employee Benefit Plans, Collective Bargaining Plans and the Company's preferred stock
 purchase rights agreement.
- The provisions of the Plan will be binding on the Debtors and any creditor or equity interest holder of the Debtors. Also, the Confirmation Order discharges the Debtors from any and all debts, Claims, and Interests that arose on or prior to the Effective Date (other than Claims and Interests that are being reinstated pursuant to the Plan) and substitutes for such debt, Claims and Interests the obligations specified under the Plan.
- Subject to the terms of the Plan, the Debtors will on the Effective Date be discharged and released from any and all Claims that are not reinstated under the Plan. All Entities that held such claims will be enjoined pursuant to the Plan and the Confirmation Order from asserting any such claims against the Debtors or the Reorganized Debtors and their respective assets and property. In addition, the Debtors will release certain claims against the "Released Parties" identified in the Plan, which include the Consenting Lenders, Backstop Lenders, Consenting Noteholders, the DIP Revolving Facility Lenders and certain agents thereof and their respective officers, directors, affiliates and representatives. The Holders of Claims under the Company's existing prepetition credit facility and the holders of the Pulitzer Notes have also granted certain releases in favor of the Debtors and certain parties related to the Debtors contained in the Plan, subject to the rights of certain of those holders under the Plan to opt out of such releases. The Plan and the Confirmation Order also contain broad exculpation provisions in favor of the Debtors and the other Released Parties under the Plan, pursuant to which the Debtors and the other Released Parties shall not have nor incur liability to, or be subject to actions by, any Holder of a Claim against or Interest in the Debtors, or other parties relating thereto, for any act or omission in connection with, relating to, or arising out of the Ch. 11 Proceedings, the Plan and Disclosure Statement, and certain other documents and actions relating to the Ch. 11 Proceedings and the restructuring contemplated by the Plan.

As of the Confirmation Date, the Company had 44,957,601 shares of Common Stock, par value \$2.00 per share, issued and outstanding. In addition, as of the Confirmation Date, the Company's stockholders have previously authorized the Company to reserve for issuance (i) 1,400,000 shares of Common Stock under the Company's 1977 Employee Stock Purchase Plan, as amended (of which 270,104 shares remain unissued); (ii) 135,000 shares of Common Stock under the Company's Supplemental Employee Stock Purchase Plan, as amended (of which 8,717 shares remain unissued); (iii) 6,250,000 shares of Common Stock under the Company's Amended and Restated 1990 Long-Term Incentive Plan (of which 3,268,425 shares remain unissued); and (iv) 450,000 shares of Common Stock under the Company's Amended and Restated 1996 Stock Plan for Non-Employee Directors (of which 182,455 shares remain unissued). The Company has an unused balance of Common Stock of 71,312,698 shares.

As of the Effective Date, under the Company's Certificate of Incorporation, the total number of shares of all

classes of the reorganized Company's capital stock which the Company will have authority to issue will be 150,500,000, consisting of 500,000 shares of Serial Convertible Preferred Stock, without par value, 120,000,000 shares of Common Stock, par value \$0.01 per share, and 30,000,000 shares of Class B Common Stock, par value \$2.00 per share. As of the Effective Date, the Company will not have any shares of Class B Common Stock or Serial Convertible Preferred Stock outstanding.

Information regarding the assets and liabilities of the Company as of September 25, 2011 is incorporated herein by reference to the Company's annual report on Form 10-K for the fiscal year ended September 25, 2011 filed with the Securities and Exchange Commission ("SEC") on December 9, 2011.

On January 23, 2012, the Company issued a news release announcing that the Bankruptcy Court had confirmed the Plan. A copy of the news release is attached hereto as **Exhibit 99.1** and incorporated herein by reference.

Item 7.01. Regulation FD Disclosure.

On January 20, 2012, the Debtors filed their unaudited Monthly Operating Report for the Filing Period ended December 25, 2011 with the Bankruptcy Court, which is available on the Company's refinancing website located at www.leerefinancing.com.

The Monthly Operating Report is limited in scope, covers a limited time period and has been prepared solely for the purpose of complying with reporting requirements of the Bankruptcy Court and the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"). The financial information contained in the Monthly Operating Report is preliminary and unaudited and does not purport to show the financial statements of any of the Debtors in accordance with accounting principles generally accepted in the United States of America ("GAAP") and, therefore, may exclude items required by GAAP, such as certain reclassifications, eliminations, accruals and disclosure items. In addition, certain of the financial information contained in the Monthly Operating Report also excludes certain non-Debtor entities of the Company that are consolidated for GAAP purposes. The Monthly Operating Report may be subject to revision. The Monthly Operating Report is in a format required by the Bankruptcy Court and the Bankruptcy Rules and should not be used for investment purposes. The information in the Monthly Operating Report should not be viewed as indicative of future results. The Company cautions readers not to place undue reliance on the Monthly Operating Reports or other operating reports filed with the Bankruptcy Court.

Limitation on Incorporation by Reference

The Monthly Operating Report shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended. Registration statements or other documents filed with the SEC shall not incorporate the Monthly Operating Report by reference, except as otherwise expressly stated in such filing. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information in the Monthly Operating Report that is required to be disclosed solely by Regulation FD.

Forward-looking Statements

The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" for forward-looking statements. This Current Report contains information that may be deemed forward-looking that is based largely on our current expectations, and is subject to certain risks, trends and uncertainties that could cause actual results to differ materially from those anticipated. Among such risks, trends and other uncertainties, which in some instances are beyond our control, are our ability to generate cash flows and maintain liquidity sufficient to

service our debt, comply with or obtain amendments or waivers of the financial covenants contained in our credit facilities, if necessary, and to refinance our debt as it comes due. Other risks and uncertainties include the impact and duration of continuing adverse economic conditions, changes in advertising demand, potential changes in newsprint and other commodity prices, energy costs, interest rates, availability of credit, labor costs, legislative and regulatory rulings, difficulties in achieving planned expense reductions, maintaining employee and customer relationships, increased capital costs, maintaining our listing status on the NYSE, competition and other risks detailed from time to time in our publicly filed documents. Any statements that are not statements of historical fact (including statements containing the words "may", "will", "would", "could", "believe", "expect", "anticipate", "intend", "plan", "project", "consider" and similar expressions) generally should be considered forward-looking statements. Readers are cautioned not to place undue reliance on such forward-looking statements, which are made as of the date of this report. We do not undertake to publicly update or revise our forward-looking statements.

Item 8.01. Other Events.

On January 23, 2012, the Company's Chairman, President and Chief Executive Officer, Mary E. Junck, sent a letter to the Company's stockholders and employees. A copy of the this letter is furnished with this report as **Exhibit 99.2**.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits	
2.1	Confirmation Order dated January 23, 2012
2.2	Second Amended Joint Prepackaged Plan of Reorganization for Lee Enterprises, Incorporated and its Debtor Subsidiaries, as confirmed by the Bankruptcy Court on January 23, 2012
99.1	News Release of Lee Enterprises, Incorporated dated January 23, 2012
99.2	Letter, dated January 23, 2012, to the stockholders and employees of Lee Enterprises, Incorporated from its Chairman, President and Chief Executive Officer

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LEE ENTERPRISES, INCORPORATED

Date: January 25, 2012

Carl G. Schmidt

Vice President, Chief Financial Officer,

and Treasurer

INDEX TO EXHIBITS

Exhibit No.	Description
2.1	Confirmation Order dated January 23, 2012
2.2	Second Amended Joint Prepackaged Plan of Reorganization for Lee Enterprises, Incorporated and its Debtor Subsidiaries, as confirmed by the Bankruptcy Court on January 23, 2012
99.1	News Release of Lee Enterprises, Incorporated dated January 23, 2012
99.2	Letter, dated January 23, 2012, to the stockholders and employees of Lee Enterprises, Incorporated from its Chairman, President and Chief Executive Officer

UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

	Ref Docket No. 159
Debtors.	Jointly Administered
LEE ENTERPRISES, INCORPORATED, et al., ¹	Case No. 11-13918 (KG)
In re:	Chapter 11

FINDINGS OF FACT, CONCLUSIONS OF LAW
AND ORDER (I) APPROVING (A) SOLICITATION AND
DISCLOSURE STATEMENT AND (B) PREPETITION SOLICITATION
AND VOTING PROCEDURES AND (II) CONFIRMING SECOND
AMENDED JOINT PREPACKAGED PLAN OF REORGANIZATION
FOR LEE ENTERPRISES, INCORPORATED AND ITS DEBTOR SUBSIDIARIES

WHEREAS, Lee Enterprises, Incorporated ("<u>Lee Enterprises</u>") and certain of its direct and indirect subsidiaries, as debtors and debtors in possession in the above-captioned, jointly administered cases (collectively, the "<u>Debtors</u>" and, as reorganized entities after emergence, the "<u>Reorganized Debtors</u>"), filed voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") on December 12, 2011 (the "<u>Petition Date</u>");

WHEREAS, on the Petition Date, the Debtors filed the Affidavit of Carl G. Schmidt, Vice President, Chief Financial Officer and Treasurer of Lee Enterprises, Incorporated, in Support of First Day Motions [Docket No. 3] (the "First Day Affidavit") setting forth, among other things, the circumstances surrounding the commencement of these Chapter 11 Cases;

¹ The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, where applicable, are: Lee Enterprises, Incorporated (3980); Accudata, Inc. (3648); Fairgrove LLC (N/A); Flagstaff Publishing Co. (4796); Hanford Sentinel, Inc. (0775); HomeChoice, LLC (3701); HSTAR LLC (N/A); INN Partners, L.C. (9478); Journal-Star Printing Co. (6412); Kauai Publishing Co. (7036); K. Falls Basin Publishing, Inc. (3285); Lee Consolidated Holdings Co. (2158); Lee Procurement Solutions Co. (6292); Lee Publications, Inc. (1886); Napa Valley Publishing Co. (7802); NIPC, Inc. (9541); NLPC LLC (7381); Northern Lakes Publishing Co. (9679); NVPC LLC (N/A); Pantagraph Publishing Co. (7058); Pulitzer Missouri Newspapers, Inc. (1960); Pulitzer Network Systems LLC (5359); Pulitzer Newspapers, Inc. (1560); Pulitzer Technologies, Inc. (8892); Pulitzer Utah Newspapers, Inc. (1884); Pulitzer Inc. (9711); Santa Maria Times, Inc. (3801); SHTP LLC (N/A); Sioux City Newspapers, Inc. (9191); SOPC LLC (5551); Southwestern Oregon Publishing Co. (0741); St. Louis Post-Dispatch LLC (5357); STL Distribution Services LLC (0922); Suburban Journals of Greater St. Louis LLC (6217); and Ynez Corporation (5443). The Debtors' corporate headquarters and the service address for each Debtor is 201 N. Harrison Street, Suite 600, Davenport, IA 52801.

WHEREAS, the Debtors have proposed the Second Amended Joint Prepackaged Plan of Reorganization for Lee Enterprises and Its Debtor Subsidiaries, dated January 19, 2012 [Docket No. 159] (as initially proposed on November 7, 2011, and as the same has been or may be further amended, supplemented or modified, the "Plan");

WHEREAS, prior to the Petition Date, Holders of Prepetition Credit Agreement Claims executed and delivered copies of that certain Support Agreement dated as of August 11, 2011 and effective as of September 8, 2011 (as amended by the First Amendment to the Support Agreement dated December 2, 2011 and as may be further amended, supplemented or modified, the "Lee Support Agreement"), pursuant to which more than 50% of the Holders of Prepetition Credit Agreement Claims in number and more than 66.7% of the Holders of Prepetition Credit Agreement Claims in amount committed to, among other things, vote to accept the Plan, support Confirmation of the Plan, and not opt out of the releases set forth in the Plan, subject to the terms and conditions thereof;

WHEREAS, prior to the Petition Date, Holders of PD LLC Notes Claims executed and delivered copies of that certain Support Agreement dated as of December 2, 2011 (as the same may be further amended, supplemented or modified, the "Pulitzer Support Agreement"), pursuant to which more than 50% of the Holders of PD LLC Notes Claims in number and more than 66.7% of the Holders of PD LLC Notes Claims in amount committed to, among other things, vote to accept the Plan, support Confirmation of the Plan, and not opt out of the releases set forth in the Plan, subject to the terms and conditions thereof;

² Capitalized terms and phrases used but not otherwise defined herein have the meaning assigned to such terms in the Plan. The rules of interpretation set forth in Article I.B of the Plan apply to this Confirmation Order. In addition, in accordance with Article I.A of the Plan, any term used in the Plan or this Confirmation Order that is not defined in the Plan or this Confirmation Order, but that is used in the Bankruptcy Code or the Bankruptcy Rules, has the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable. A copy of the Plan (without the Exhibits thereto) is attached to this Confirmation Order as Exhibit A and is incorporated herein by reference.

WHEREAS, prior to the Petition Date, the Debtors solicited votes to accept or reject the Plan from the Holders of Claims entitled to vote on the Plan pursuant to sections 1126 and 1129 of the Bankruptcy Code; and, specifically, (i) on November 7, 2011, the Debtors (a) caused The Garden City Group, Inc. (the "Voting Agent"), the claims, noticing, and voting agent in these Chapter 11 Cases, to distribute copies of the Solicitation and Disclosure Statement, dated November 7, 2011 (as subsequently amended, supplemented or modified, the "Solicitation and Disclosure Statement"), the Plan, and the relevant Ballots via overnight mail to each Holder of Prepetition Credit Agreement Claims in Classes 1C through 9C under the Plan as of November 4, 2011 (the "Lee Voting Record Date") and (b) made the Plan and the Solicitation and Disclosure Statement available to all Holders of Prepetition Credit Agreement Claims as of the Lee Voting Record Date via Intralinks; and (ii) on December 3, 2011, the Debtors caused the Voting Agent to distribute copies of the Solicitation and Disclosure Statement, dated December 2 2011, the Plan, and the relevant Ballots via overnight mail to each Holder of PD LLC Notes Claims in Classes 10C through 35C under the Plan as of November 30, 2011;

WHEREAS, the Debtors established December 9, 2011 at 4:00 p.m., prevailing Eastern Time, as the deadline by which Holders of Prepetition Credit Agreement Claims and PD LLC Notes Claims were required to have returned their completed Ballots to the Voting Agent (the "Voting Deadline");

³ Copies of the forms of Ballots were attached to the Motion of the Debtors for an Order (I) Scheduling a Combined Hearing to (A) Approve Adequacy of Solicitation and Disclosure Statement, (B) Approve Prepetition Solicitation Procedures and (C) Confirm the Joint Prepackaged Plan of Reorganization, (II) Establishing Deadlines and Procedures for Filing Objections to Approval of Solicitation and Disclosure Statement, Approval of Prepetition Solicitation Procedures and Confirmation of Plan, and (III) Approving Form and Manner of Notice of Objection Deadline and Confirmation Hearing [Docket No. 11] (the "Scheduling Motion").

⁴ As reflected in the Second Amended Declaration of Craig E. Johnson of The Garden City Group, Inc. Certifying (I) Methodology for the Tabulation of Votes on Amended Joint Prepackaged Plan of Reorganization for Lee Enterprises, Incorporated and Its Debtor Subsidiaries and (II) Voting Results [Docket No. 157], the Voting Deadline was extended for certain parties.

WHEREAS, on the Petition Date, Craig E. Johnson of the Voting Agent, executed an affidavit of service (which the Voting Agent filed with the Court on the Petition Date [Docket No. 16]) regarding the mailing of a notice that provided recipients with information concerning, among other things: (a) the filing of the Debtors' voluntary chapter 11 petitions on the Petition Date, (b) the date, time, and place of the Confirmation Hearing, (c) the deadline and procedures for filing objections to the adequacy of the Solicitation and Disclosure Statement, the Prepetition Solicitation Procedures, and/or confirmation of the Plan, and (d) how to obtain copies of the Solicitation and Disclosure Statement and the Plan (the "Confirmation Hearing Notice") in accordance with the terms of the Scheduling Order (as amended on December 12, 2011 [Docket No. 30] and January 18, 2012 [Docket No. 157], the "Scheduling/Voting Affidavit");

WHEREAS, on December 13, 2011, the Court entered the Order (I) Scheduling a Combined Hearing to (A) Approve Adequacy of Solicitation and Disclosure Statement, (B) Approve Prepetition Solicitation Procedures and (C) Confirm the Joint Prepackaged Plan of Reorganization, (II) Establishing Deadlines and Procedures for Filing Objections to Approval of Solicitation and Disclosure Statement, Approval of Prepetition Solicitation Procedures and Confirmation of Plan, and (III) Approving Form and Manner of Notice of Objection Deadline and Confirmation Hearing [Docket No. 57] (the "Scheduling Order"), pursuant to which the Court (i) scheduled a combined hearing on January 23, 2012 (the "Confirmation Hearing") to, among other things, consider (a) approval of the Solicitation and Disclosure Statement pursuant to section 1125 of the Bankruptcy Code, (b) approval of the prepetition solicitation and voting procedures described in the Scheduling/Voting Affidavit (the "Prepetition Solicitation Procedures"), and (c) confirmation of the Plan and (ii) established January 13, 2012 at 4:00 p.m. (prevailing Eastern Time) as the deadline by which all objections to (a) approval of the Prepetition Solicitation Procedures, (b) approval of the

Solicitation and Disclosure Statement, and/or (c) confirmation of the Plan must have been filed with the Court and received by the parties listed in the Scheduling Motion (the "Objection Deadline");

WHEREAS, on January 9, 2012, the Voting Agent filed an affidavit of publication, attesting that a publication version of the Confirmation Hearing Notice (which the Court approved pursuant to the Scheduling Order) was published in (a) the national edition of <a href="https://doi.org/10.2011/j.com/hem-10.2011/j.com/hem

WHEREAS, on January 19, 2012, the Debtors filed the affidavits of (i) Carl G. Schmidt, Vice President, Chief Financial Officer and Treasurer of Lee Enterprises [Docket No. 162] (the "Schmidt Confirmation Affidavit" and, together with the First Day Affidavit, the "Schmidt Affidavits") and (ii) Jordan Steinberg, a Vice President of The Blackstone Group [Docket No. 163] (the "Steinberg Affidavit" and, together with the Schmidt Affidavits, the "Supporting Affidavits") in support of confirmation of the Plan;

WHEREAS, on January 19, 2012, the Debtors filed a memorandum of law in support of confirmation of the Plan (the "Memorandum of Law") [Docket No. 164];

WHEREAS, in the Scheduling/Voting Affidavit, the Voting Agent certified that, as of the Voting Deadline, (i) 100% of the Holders of Prepetition Credit Agreement Claims that voted on the Plan, totaling \$841,908,887.37 in amount of such Claims, had voted to accept the Plan, and (ii) 100% of the Holders of PD LLC Notes Claims that voted on the Plan, holding \$138,000,000.00 in amount of such Claims, had voted to accept the Plan;

WHEREAS, (i) no objections to (a) approval of the Prepetition Solicitation Procedures and (b) approval of the Solicitation and Disclosure Statement were filed or received by the Debtors

before the Objection Deadline; (ii) one objection to confirmation of the Plan was filed before the Objection Deadline by the Internal Revenue Service (the "IRS") [Docket No. 156] (the "IRS Objection"); and (iii) one informal objection to the confirmation of the Plan was received by the Debtors before the Objection Deadline from Earnest L. Haynes, Jr. (the "Haynes Objection" and, together with the IRS Objection, the "Plan Objections");

WHEREAS, the Court held the Confirmation Hearing on January 23, 2012;

WHEREAS, the Court has reviewed the Plan, the Solicitation and Disclosure Statement, the Scheduling Order, the Scheduling/Voting Affidavit, the Publication Affidavit, the Memorandum of Law, the Supporting Affidavits, the Plan Objections, and all other materials submitted to the Court in connection with the Confirmation Hearing; and

WHEREAS, the Court has considered (i) the statements made by counsel at the Confirmation Hearing and (ii) all evidence proffered or adduced at the Confirmation Hearing.

NOW, THEREFORE, after due deliberation and good cause appearing therefor, the Court hereby makes and issues the following Findings of Fact, Conclusions of Law and Order with respect to approval of the Solicitation and Disclosure Statement, approval of the Prepetition Solicitation Procedures, and confirmation of the Plan (the "Confirmation Order"):

I. FINDINGS OF FACT AND CONCLUSIONS OF LAW⁵

A. <u>Jurisdiction and Venue</u>. On the Petition Date, the Debtors commenced these Chapter 11 Cases in good faith by filing voluntary petitions for relief under chapter 11 of the Bankruptcy Code. The Debtors are (and were as of the Petition Date) qualified to be debtors under section 109 of the Bankruptcy Code. The Court has jurisdiction over the Chapter 11 Cases pursuant

⁵ This Confirmation Order constitutes the Bankruptcy Court's findings of fact and conclusions of law under Federal Rule of Civil Procedure 52, as made applicable herein by Bankruptcy Rules 7052 and 9014. Any and all findings of fact herein shall constitute findings of fact even if stated as conclusions of law, and any and all conclusions of law herein shall constitute conclusions of law even if stated as findings of fact.

to 28 U.S.C. § 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(L). The Court has subject matter jurisdiction over the Chapter 11 Cases pursuant to 28 U.S.C. §§ 157(b)(1) and 1334. Venue was proper as of the Petition Date and continues to be proper before this Court as of the date hereof pursuant to 28 U.S.C. §§ 1408 and 1409.

- B. <u>Judicial Notice</u>. The Court takes judicial notice of the dockets of these Chapter 11 Cases maintained by the Clerk of the Court, including, without limitation, all pleadings and other documents filed, all orders entered, all hearing transcripts and all evidence and arguments made, proffered or adduced at the hearings held before the Court during these Chapter 11 Cases, including, without limitation, the Confirmation Hearing.
- C. <u>Burden of Proof.</u> The Debtors, as proponents of the Plan, have the burden of proving the elements of sections 1129(a) and 1129(b) of the Bankruptcy Code by a preponderance of the evidence and have met that burden as further found and determined herein.
- D. Adequacy of the Information Contained in the Solicitation and Disclosure Statement. The Solicitation and Disclosure Statement (i) contains "adequate information" as such term is defined in section 1125 of the Bankruptcy Code, thereby satisfying sections 1125 and 1126(b) of the Bankruptcy Code, (ii) contains sufficient information of a kind necessary to satisfy the disclosure requirements of any applicable non-bankruptcy law, and (iii) is approved in all respects.
- E. <u>Transmittal of Solicitation Materials</u>. To obtain the requisite acceptances of the Plan, (i) on or about November 7, 2011, the Debtors caused the Voting Agent to transmit the Plan, the Solicitation and Disclosure Statement, and relevant Ballots to Holders of Prepetition Credit Agreement Claims in accordance with the Prepetition Solicitation Procedures and (ii) on or about December 3, 2011, the Debtors caused the Voting Agent to transmit the Plan, the Solicitation and Disclosure Statement, and relevant Ballots to Holders of PD LLC Notes Claims in accordance

with the Prepetition Solicitation Procedures. Such transmittal and service of the Plan, the Solicitation and Disclosure Statement, and the Ballots constitutes due and sufficient notice of the Solicitation and Disclosure Statement, the Plan, and the Voting Deadline. The procedures by which the Ballots for acceptance or rejection of the Plan were solicited and tabulated were fair, properly conducted, and in accordance with sections 1125 and 1126 of the Bankruptcy Code, Bankruptcy Rules 3017 and 3018, and any and all other applicable rules, laws and regulations.

- F. Good Faith Solicitation. Votes to accept or reject the Plan were solicited fairly, in good faith, and in compliance with the Bankruptcy Code, the Bankruptcy Rules, and all other applicable rules, laws and regulations. Based on the record of these Chapter 11 Cases, the Debtors and the other Released Parties have acted in "good faith" within the meaning of section 1125(e) of the Bankruptcy Code and in compliance with the applicable provisions of the Bankruptcy Code and the Bankruptcy Rules in connection with the solicitation of votes to accept the Plan, and otherwise. Accordingly, the Released Parties are entitled to the protections afforded by section 1125(e) of the Bankruptcy Code and Sections 10.3 and 10.4 of the Plan.
- G. Technical Modifications to the Plan. Any modifications embodied in the Plan or the Exhibits to the Plan (collectively, the "Plan Modifications") constitute technical or non-material changes and do not materially and adversely affect or change the treatment of any Claims or Interests. Accordingly, pursuant to Bankruptcy Rule 3019, the Plan Modifications do not require additional disclosure under section 1125 of the Bankruptcy Code or the re-solicitation of votes under section 1126 of the Bankruptcy Code, nor do they require that the Holders of Claims or Interests be afforded an opportunity to change any previously-cast acceptances or rejections of the Plan
 - II. <u>Notice of Confirmation Hearing</u>. The Confirmation Hearing Notice was

transmitted and served as set forth in the Scheduling Order (as evidenced by the Scheduling/Voting Affidavit), and such transmittal and service of Confirmation Hearing Notice constitutes due and sufficient notice of the Plan, the Confirmation Hearing, the Voting Deadline, and the Objection Deadline. In addition, in accordance with Bankruptcy Rules 2002(1) and 3017(d) and the Scheduling Order, and as set forth in the Publication Affidavit, the Debtors caused a publication version of the Confirmation Hearing Notice (which was attached to the Scheduling Motion as Exhibit C thereto) to be published in (a) the national edition of The Wall Street Journal on December 22, 2011, (b) the national edition of USA Today on December 22, 2011, and (c) the daily edition of the St. Louis Post-Dispatch on December 19, 2011. Such notice was adequate and sufficient under the circumstances, was given in compliance with Bankruptcy Rules 2002, 3017 and 3020 and the Scheduling Order, and no other or further notice shall be required. All parties in interest in the Chapter 11 Cases had a full and fair opportunity to appear and be heard at the Confirmation Hearing.

- I. Propriety of Various Agreements and Documents Relating to the Plan. In light of all of the circumstances and the record in these Chapter 11 Cases, each transaction contemplated in connection with the Plan is integral to the terms and conditions set forth in the Plan and is critical to the implementation of the Plan. All contracts, instruments, releases, agreements and documents related to, or necessary to implement, effectuate and consummate the Plan are valid, proper and reasonable under the circumstances and due and sufficient notice thereof has been provided in connection with approval of the Solicitation and Disclosure Statement and Confirmation of the Plan.
- J. <u>Releases, Discharges, Exculpations and Injunctions</u>. Based upon the facts and circumstances of these Chapter 11 Cases, the releases, discharges, exculpations,

indemnifications and injunctions set forth in Article X of the Plan are (a) supported by good and valuable consideration, the adequacy of which is hereby confirmed, (b) fair, equitable, and reasonable, (c) supported by the record of these Chapter 11 Cases, and (d) in the best interests of the Debtors, their Estates, and the Holders of Claims against and Interests in the Debtors.

- K. Assumption of Executory Contracts and Unexpired Leases. The assumption by the applicable Debtors and assignment to the applicable Reorganized Debtors, as of the Effective Date, of (i) all executory contracts and unexpired leases to which any of the Debtors is a party, (ii) all Collective Bargaining Agreements, (iii) all insurance policies issued to, or insurance agreements entered into by, the Debtors prior to the Petition Date (to the extent that such insurance policies or agreements are considered to be executory contracts), (iv) all contracts, agreements and leases that the Debtors entered into or assumed after the Petition Date, and (v) the Employee Benefit Plans and Pension Plans, satisfies the applicable provisions of the Bankruptcy Code, is supported by the Debtors' sound business judgment, and is in the best interests of the Debtors and their Estates. The assumption or rejection of any executory contract or unexpired lease pursuant to the Plan shall be legal, valid, and binding as of the Effective Date upon the applicable Debtor or Reorganized Debtor and all non-Debtor parties to such executory contract or unexpired lease.
- L. <u>Exemption from Securities Laws</u>. The offer, issuance, distribution, transfer or exchange of the New Common Stock and New PD LLC Notes pursuant to the Plan shall be, and shall be deemed to be, exempt from registration under any applicable federal or state securities laws to the fullest extent permissible under applicable non-bankruptcy law and under bankruptcy law, including, without limitation, section 1145(a) of the Bankruptcy Code.
- M. <u>Compliance with Requirements of Section 1129 of the Bankruptcy Code</u>. The Plan satisfies all requirements for confirmation under section 1129 of the Bankruptcy Code, as fol

lows:

- i. Section 1129(a)(1) Compliance with Applicable Provisions of the Bankruptcy Code. The Plan complies with all applicable provisions of the Bankruptcy Code as required by section 1129(a)(1) of the Bankruptcy Code, including, without limitation, sections 1122 and 1123 of the Bankruptcy Code and Bankruptcy Rule 3016.
- ii. Section 1122 Classification of Claims and Interests. The Plan constitutes a separate plan of reorganization for each of the Debtors. In accordance with section 1122(a) of the Bankruptcy Code, each Class of Claims against and Interests in the Debtors contains only Claims or Interests that are substantially similar to the other Claims or Interests within that Class. Valid business, factual and legal reasons exist for separately classifying the various Classes of Claims against and Interests in the Debtors under the Plan.
- iii. <u>Section 1123(a) Mandatory Provisions of the Plan</u>. The Plan fully complies with each requirement of section 1123(a) of the Bankruptcy Code, as set forth below:
- (a) Section 1123(a)(1). Article III of the Plan properly designates all Claims and Interests that require classification, as required by section 1123(a)(1) of the Bankruptcy Code. Pursuant to section 1123(a)(1) of the Bankruptcy Code, Administrative Expense Claims, DIP Revolving Facility Claims, Priority Tax Claims, and Section 507(b) Claims are not required to be designated into Classes.
- (b) <u>Section 1123(a)(2)</u>. In accordance with section 1123(a)(2) of the Bankruptcy Code, Article III of the Plan specifies each Class of Claims or Interests that is Unimpaired under the Plan. In particular, Article III of the Plan provides that Classes 1A through 35A (Priority Non-Tax Claims against the Debtors), 1B through 35B (Other Secured Claims against the Debtors), 1D through 35D (General Unsecured Claims against the Debtors), Classes 1E through

35E (Intercompany Claims against the Debtors), 12F and 14F (Herald Claim against PD LLC and DS LLC), and 1G through 35G (Interests in the Debtors) are Unimpaired under the Plan.

- (c) Section 1123(a)(3). In accordance with section 1123(a)(3) of the Bankruptcy Code, Article III of the Plan specifies the treatment of Classes 1C through 9C (Prepetition Credit Agreement Claims against the Lee Debtors) and Classes 10C through 35C (PD LLC Notes Claims against the Pulitzer Debtors), the only Classes of Claims that are Impaired under the Plan.
- (d) Section 1123(a)(4). In accordance with section 1123(a)(4) of the Bankruptcy Code, Article III of the Plan provides the same treatment for each Claim or Interest in a given Class unless the Holder of such Claim or Interest agrees to less favorable treatment, in which case such less favorable treatment shall control.
- (e) Section 1123(a)(5). In accordance with section 1123(a)(5) of the Bankruptcy Code, Article V of the Plan provides adequate means for the Plan's implementation. On the Effective Date of the Plan, the Reorganized Debtors will have sufficient Cash to make all payments required to be made on the Effective Date pursuant to the terms of the Plan. Moreover, Article V and various other provisions of the Plan specifically provide adequate and proper means for the Plan's implementation including, without limitation: (i) the continued corporate existence and vesting of assets in the Reorganized Debtors; (ii) the issuance and distribution of the New Common Stock; (iii) the cancellation of the Prepetition Credit Agreement Documents and the PD LLC Notes Documents; (iv) the conversion of the DIP Revolving Facility into the New First Lien Revolving Credit Facility; (v) the entry by the Reorganized Debtors into the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents; and (vi) the continuation in their positions of the directors of Reorganized Lee

Enterprises and the other Reorganized Debtors.

- (f) Section 1123(a)(6). In accordance with section 1123(a)(6) of the Bankruptcy Code, the Certificate of Incorporation and the organizational documents for the Reorganized Debtors contain provisions prohibiting the issuance of nonvoting equity securities, provide for the appropriate distribution of voting power among all classes of equity securities authorized for issuance, and otherwise comply with the requirements of section 1123(a)(6) of the Bankruptcy Code.
- (g) Section 1123(a)(7). In accordance with section 1123(a)(7) of the Bankruptcy Code, the provisions of the Plan regarding the manner of selection of officers and directors of the Reorganized Debtors are consistent with the interests of creditors and equity security holders and with public policy.
- iv. <u>Section 1123(b) Discretionary Provisions of the Plan</u>. In accordance with section 1123(b)(6) of the Bankruptcy Code, the Plan does not contain any discretionary provision that is inconsistent with the applicable provisions of the Bankruptcy Code.
- v. <u>Bankruptcy Rule 3016 Identification of the Plan; Conspicuous Injunction</u>. In accordance with Bankruptcy Rule 3016(a), the Plan is dated and identifies that the Debtors are submitting it as proponents. Other than conduct otherwise enjoined by the Bankruptcy Code, the Plan describes in specific and conspicuous language all acts to be enjoined and identifies the Entities subject to the injunctions set forth in the Plan, in accordance with Bankruptcy Rule 3016(c).
- vi. <u>Section 1129(a)(2) Compliance by the Debtors with Applicable Provisions of the Bankruptcy Code</u>. The Debtors, as the proponents of the Plan, have complied with all applicable provisions of the Bankruptcy Code, including, without limitation, sections 1123, 1125

and 1126 of the Bankruptcy Code and Bankruptcy Rules 3017, 3018 and 3019.

- vii. Section 1129(a)(3) The Debtors Have Proposed the Plan in Good Faith. The Debtors have proposed the Plan in good faith and not by any means forbidden by law. Consistent with the overriding purpose of chapter 11 of the Bankruptcy Code, the Plan is designed to allow each of the Debtors to reorganize on a going concern basis while maximizing recoveries to the Holders of Claims against and Interests in the Debtors. In determining that the Plan has been proposed in good faith, the Court has examined the totality of the circumstances surrounding the filing of the Chapter 11 Cases, the Plan itself, and the process leading to its formulation. The Chapter 11 Cases were filed, and the Plan was proposed, with the legitimate purpose of reorganizing the Debtors in order to maximize the value of their respective Estates.
- viii. Section 1129(a)(4) Court Approval of Certain Administrative Expense Payments. In accordance with section 1129(a)(4) of the Bankruptcy Code, any payment made or promised by the Debtors or by any Person issuing securities or acquiring property under the Plan, for services rendered or expenses incurred in connection with the Chapter 11 Cases, or in connection with the Plan and incident to the Chapter 11 Cases, has been disclosed to this Court. The procedures set forth in the Plan for the Court's review and allowance of the fees and expenses to be paid by the Debtors satisfy the objectives of section 1129(a)(4) of the Bankruptcy Code. Accordingly, the Plan complies with section 1129(a)(4) of the Bankruptcy Code.
- ix. Section 1129(a)(5) Disclosure of Identity of Directors and Officers and the Compensation of Insiders. In accordance with section 1129(a)(5) of the Bankruptcy Code, the Debtors have disclosed the identity and affiliations of the directors and officers of the Reorganized Debtors. The appointment to, or continuance in, such offices by such Persons is consistent with the interests of the Holders of Claims and Interests and with public policy. The Debtors have also

disclosed the identity and affiliations of any insiders that will be employed or retained by the Reorganized Debtors and the nature of any compensation for such insiders.

- x. <u>Section 1129(a)(6) Approval of Rate Changes</u>. The Plan does not provide for any rate changes subject to the jurisdiction of any governmental regulatory commission. Accordingly, section 1129(a)(6) of the Bankruptcy Code is inapplicable to the Plan.
- xi. Section 1129(a)(7) Best Interests of Holders of Claims and Interests. The Plan satisfies the "best interests of creditors" test under section 1129(a)(7) of the Bankruptcy Code. Each Impaired Class of Claims (i.e., Prepetition Credit Agreement Claims against the Lee Debtors and PD LLC Notes Claims against the Pulitzer Debtors) has accepted the Plan. Further, the liquidation analysis annexed to the Solicitation and Disclosure Statement as Exhibit D (the "Liquidation Analysis"), including the methodology used and estimations and assumptions made therein, and the evidence related thereto that was proffered at the Confirmation Hearing, (a) is persuasive and credible as of the dates such evidence was prepared, presented or proffered, (b) either has not been controverted by other persuasive evidence or has not been challenged, (c) is based upon reasonable and sound assumptions, and (d) provide a reasonable estimate of the liquidation value of the Estates upon conversion of the Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code. Accordingly, the Plan satisfies the requirements of section 1129(a)(7) of the Bankruptcy Code.
- xii. Section 1129(a)(8) Acceptance of the Plan by Each Impaired Voting Class. As indicated in Article III of the Plan and as set forth in the Scheduling/Voting Affidavit, Classes 1C through 9C (Prepetition Credit Agreement Claims against the Lee Debtors) and Classes 10C through 35C (PD LLC Claims against the Pulitzer Debtors) are Impaired under the Plan. The Holders of Claims in Classes 1C through 9C (Prepetition Credit Agreement Claims against the Lee

Debtors) and Classes 10C through 35C (PD LLC Notes Claims against the Pulitzer Debtors) have voted to accept the Plan by majorities in excess of one-half in number and two-thirds in amount of the Claims in such Classes, as required by section 1126(c) of the Bankruptcy Code.

- xiii. Section 1129(a)(9) Treatment of Claims Entitled to Priority. The Plan provides for the treatment of Claims entitled to priority pursuant to sections 507(a)(2)-(8) of the Bankruptcy Code in the manner required by section 1129(a)(9) of the Bankruptcy Code.
- xiv. Section 1129(a)(10) Acceptance by at Least One Impaired, Non-Insider Class. In accordance with the requirements of section 1129(a)(10) of the Bankruptcy Code and as evidenced by the Scheduling/Voting Affidavit, at least one Impaired Class of Claims against each Debtor that has any Impaired Classes of Claims against it has accepted the Plan, excluding the votes of insiders. Accordingly, the Plan satisfies section 1129(a)(10) of the Bankruptcy Code.
- xv. Section 1129(a)(11) Feasibility of the Plan. The Plan is feasible. The evidence proffered or adduced at the Confirmation Hearing (a) is persuasive and credible, (b) either has not been contravened by other persuasive evidence or has not been challenged, and (c) establishes that the Plan has a reasonable likelihood of success and that confirmation of the Plan is not likely to be followed by the liquidation, or the need for further financial reorganization, of the Reorganized Debtors. Accordingly, the Plan satisfies the requirements of section 1129(a)(11) of the Bankruptcy Code.
- xvi. Section 1129(a)(12) Payment of Statutory Bankruptcy Fees. In accordance with section 1129(a)(12) of the Bankruptcy Code, Section 12.7 of the Plan provides for payment of the fees payable pursuant to 28 U.S.C. § 1930 as and when due or as otherwise agreed by the United States Department of Justice, Office of the United States Trustee. Accordingly, the Plan satisfies the requirements of section 1129(a)(12) of the Bankruptcy Code.

- xvii. Section 1129(a)(13) Treatment of Retiree Benefits. As of the date hereof, no modifications have been made to retiree benefits pursuant to subsection (e)(1)(B) or (g) of section 1114 of the Bankruptcy Code. In accordance with section 1129(a)(13) of the Bankruptcy Code, Section 6.5 of the Plan provides for the continuation after the Effective Date of all retiree benefits, as that term is defined in section 1114 of the Bankruptcy Code. Accordingly, the Plan satisfies the requirements of section 1129(a)(13) of the Bankruptcy Code.
- xviii. Sections 1129(a)(14)-(16) Domestic Obligations, Individual Debtors, or Nonprofit Corporation or Trusts.

 None of the Debtors are (a) required to pay any domestic support obligations, (b) individuals, or (c) nonprofit corporations or trusts.

 Accordingly, sections 1129(a)(14) through (16) of the Bankruptcy Code are not applicable to the Plan.
- (xix) <u>Section 1129(d) Principal Purpose of the Plan</u>. The principal purpose of the Plan is not the avoidance of taxes or the avoidance of the application of section 5 of the Securities Act of 1933 (the "<u>Securities Act</u>"), and no party in interest has filed an objection alleging otherwise. Accordingly, the Plan satisfies the requirements of section 1129(d) of the Bankruptcy Code.
- N. <u>Satisfaction of Conditions Precedent</u>. The condition precedent to the confirmation of the Plan, as set forth in Section 9.1 of the Plan, has been satisfied or waived in accordance with such section of the Plan.

II. ORDER

NOW THEREFORE, BASED ON THE FOREGOING FINDINGS OF FACT AND CONCLUSIONS OF LAW, IT IS HEREBY ORDERED, ADJUDGED, AND DECREED THAT:

- 1. <u>Solicitation and Disclosure Statement</u>. The Solicitation and Disclosure Statement contains adequate information in accordance with sections 1125 and 1126 of the Bankruptcy Code and is hereby approved in all respects.
- 2. <u>Prepetition Solicitation Procedures</u>. The Prepetition Solicitation Procedures complied with sections 1125 and 1126 of the Bankruptcy Code, Bankruptcy Rules 3017 and 3018, all other applicable provisions of the Bankruptcy Code, and all other applicable rules, laws and regulations. Such procedures are hereby approved in all respects.
- 3. <u>Plan Objections</u>. All objections to the Plan not otherwise withdrawn or consensually resolved at or prior to the Confirmation Hearing are hereby overruled for the reasons set forth in the Memorandum of Law and/or stated by counsel at the Confirmation Hearing.
- 4. <u>Confirmation</u>. The Plan (a copy of which, without the Exhibits thereto, is attached hereto as <u>Exhibit A</u>) is approved and confirmed in all respects pursuant to section 1129 of the Bankruptcy Code. The terms of the Plan and the Exhibits thereto are incorporated by reference into this Confirmation Order. The terms of the Plan and the Exhibits thereto shall be effective and binding as of the Effective Date. Notwithstanding the foregoing, if there is any inconsistency between the terms of the Plan or the Exhibits thereto and the terms of this Confirmation Order, the terms of this Confirmation Order shall control.
- 5. <u>Plan Classification Controlling</u>. The classification of Claims and Interests for purposes of distributions under the Plan shall be governed solely by the terms of the Plan.
- 6. <u>Provisions of Plan and Confirmation Order Non-Severable and Mutually Dependent</u>. The provisions of the Plan and this Confirmation Order, including the findings of fact and conclusions of law set forth herein, are non-severable and mutually dependent.

7. Solicitation of the Plan and Exemption from Securities Laws. The Debtors, the Prepetition Administrative Agent, the Backstop Lenders, the DIP Revolving Facility Lenders, the Consenting Lenders, the Consenting Noteholders, the PD LLC Notes Collateral Agent, and each of their respective Related Persons have solicited votes on the Plan and/or have participated in the activities described in section 1125 of the Bankruptcy Code, as applicable, in good faith, within the meaning of section 1125(e) of the Bankruptcy Code, and in compliance with the applicable provisions of the Scheduling Order, the Bankruptcy Code, the Bankruptcy Rules, and all other applicable rules, laws and regulations and, accordingly, such parties are entitled to the protections afforded by section 1125(e) of the Bankruptcy Code and the exculpation provisions set forth in Article X of the Plan and shall not be liable at any time for the violation of any applicable law, rule, or regulation governing the solicitation of acceptances or rejections of the Plan or the distributions made pursuant to the Plan. Specifically, pursuant to section 1125(e) of the Bankruptcy Code, none of the parties listed above that participated in (i) the transmittal of the Plan, the Solicitation and Disclosure Statement, the Ballots and any related solicitation materials, (ii) the solicitation of votes to accept or reject the Plan, or (iii) the offer, issuance, sale or purchase of the New Common Stock and the New PD LLC Notes offered under or in connection with the Plan, shall be liable, on account of such actions or such participation, for any violation of any applicable law, rule or regulation governing the solicitation of votes to accept or reject the Plan or the offer, issuance, sale or purchase of any securities. The issuance and distribution of the New Common Stock and the New PD LLC Notes under or in connection with the Plan shall be, and shall be deemed to be, exempt from registration under any applicable federal or state securities laws to the fullest extent permissible under applicable bankruptcy law and non-bankruptcy law, including, without limitation, section 1145(a) of the Bankruptcy Code.

- 8. Authorization and Direction to Act. Pursuant to section 1142(b) of the Bankruptcy Code, no action of the stockholders, creditors or directors of the Debtors or the Reorganized Debtors shall be required to (i) authorize the Debtors and the Reorganized Debtors (or any of their respective officers, employees or agents acting on their behalf) to effectuate and carry out the Plan or any order of this Court relating thereto; (ii) consummate the transactions contemplated by the Plan, the Solicitation and Disclosure Statement, or any order of this Court; or (iii) take any other action contemplated by the Plan, the Solicitation and Disclosure Statement, or any order of this Court as may be necessary or appropriate to implement the Plan, and all such actions are hereby authorized, approved, and to the extent taken prior to the Effective Date, ratified in all respects as if they had been authorized or approved by the unanimous actions of the directors and stockholders of the Debtors and the Reorganized Debtors. This Confirmation Order constitutes the Court's approval of the Plan and the transactions and distributions contemplated thereby, including, without limitation, those transactions contemplated under Article V of the Plan, the Exhibits to the Plan, and/or this Confirmation Order.
- 9. Continued Corporate Existence and Vesting of Assets in the Reorganized Debtors. On and after the Effective Date, the Reorganized Debtors shall continue to exist as separate entities in accordance with the applicable law in the respective jurisdiction in which they are formed and pursuant to their respective certificates or articles of incorporation (or similar organizational documents) and by-laws in effect prior to the Effective Date, except to the extent such certificates or articles of incorporation (or similar organizational documents) and by-laws are to be amended and/or restated pursuant to the terms of the Plan. Notwithstanding anything to the contrary in the Plan, the Reinstated Claims against and Interests in a particular Debtor or Reorganized Debtor shall remain the obligations solely of such Debtor or Reorganized Debtor

following the Effective Date and shall not become obligations of any other Debtor or Reorganized Debtor by virtue of the Plan, the Chapter 11 Cases, or otherwise. Except as otherwise provided in the Plan, all property of the respective Estate of each Debtor, including all claims, rights and causes of action and any property acquired by the Debtors or the Reorganized Debtors under or in connection with the Plan, together with any property of each Debtor that is not property of its Estate and that is not specifically disposed of pursuant to the Plan, shall revest in the applicable Reorganized Debtor on the Effective Date free and clear of all Claims, Liens, charges, other encumbrances and Interests, provided that such revesting shall not relieve any Reorganized Debtor from such Reorganized Debtor's obligations under and the liens created pursuant to the applicable terms and provisions of the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents. Thereafter, the Reorganized Debtors may operate their businesses and may use, acquire and dispose of property free of any restrictions of the Bankruptcy Code and the Bankruptcy Rules. As of the Effective Date, all property of each Reorganized Debtor shall be free and clear of all Liens and non-Reinstated Claims and Interests, except as specifically provided in the Plan or this Confirmation Order. For the avoidance of doubt, on the Effective Date, the property of each Reorganized Debtor shall be subject to new liens pursuant to the terms of the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents.

10. <u>Cancellation of Certain Credit and Debt Documents</u>. On the Effective Date, in consideration for the distributions to be made on the Effective Date pursuant to the Plan and except as otherwise provided in the Plan, (i) all Prepetition Credit Agreement Documents, PD LLC Notes, PD LLC Notes Documents, and any other instruments, documents, plans or agreements evidencing or creating any indebtedness or obligations of a Debtor related thereto shall be

cancelled, (ii) the obligations of any of the Debtors under any Prepetition Credit Agreement Documents, PD LLC Notes, PD LLC Notes Documents, or any other agreements evidencing or creating any indebtedness or obligations of a Debtor that relate to Claims or Interests related thereto shall be discharged pursuant to Section 10.2 of the Plan, and (iii) the DIP Revolving Facility shall convert into the New First Lien Revolving Credit Facility; <u>provided</u>, <u>however</u>, that this paragraph shall not apply to Reinstated Claims and Interests, including, but not limited to, Reinstated Intercompany Claims.

11. Cancellation of Liens. Except as otherwise provided in the Plan, on the Effective Date, in consideration for the distributions to be made on the Effective Date pursuant to the Plan, (a) all Liens, charges, encumbrances and rights related to any Claim or Interest, including, without limitation, those existing under the Prepetition Credit Agreement Documents and the PD LLC Notes Documents, but excluding any Lien securing an Other Secured Claim that is Reinstated pursuant to the Plan, shall be terminated, null and void and of no effect and (b) all Liens, charges, encumbrances and rights related to any Claim or Interest under the DIP Revolving Facility shall secure the obligations under the New First Lien Revolving Credit Facility. The Holders of Secured Claims (other than Other Secured Claims that are Reinstated pursuant to the Plan) shall be authorized and directed to release any collateral or other property of any Debtor (including any Cash collateral) held by such Holder and to take such actions as may be requested by the Debtors (or the Reorganized Debtors, as the case may be) to evidence the release of any Liens, including the execution, delivery, and filing or recording of such release documents as may be requested by the Debtors (or the Reorganized Debtors, as the case may be).

The future holders of liens under the New First Lien Credit Facility Documents, the New Second Lien Credit Facility Documents, and the New PD LLC Notes Documents (collectively, the "New Liens") are hereby authorized to file,

with the appropriate authorities, financing statements and other documents (the "Perfection Documents") in order to evidence the New Liens, subject, in all cases, to the New Lee Intercreditor Agreement and the New PD LLC Intercreditor Agreement (as applicable). Notwithstanding anything in the Plan or this Confirmation Order to the contrary, if such Perfection Documents are filed prior to the Effective Date, then upon cancellation of the prior liens on the collateral subject to the New Liens, (i) such Perfection Documents shall be valid, binding, and in full force and effect, and (ii) the New Liens shall become obligations of the Reorganized Debtors, subject, in all cases, to the New Lee Intercreditor Agreement and the New PD LLC Intercreditor Agreement (as applicable).

Entry into New Credit Facilities. On the Effective Date, without any requirement of further action by stockholders, directors, members or managers of the Reorganized Debtors, the Reorganized Debtors are authorized to (i) enter into the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith; (ii) grant such liens and security interests as necessary to provide security for the applicable Debtors' obligations under the (a) New First Lien Credit Agreement, in accordance with the terms of the New First Lien Credit Facility Documents, (b) New Second Lien Term Loan Agreement, in accordance with the terms of the New Second Lien Term Loan Documents, and (c) New PD LLC Notes Agreement, New Pulitzer Guaranty Agreement, and New PD LLC Notes Subsidiary Guaranty Agreement, in accordance with the terms of the New PD LLC Notes Documents, which liens shall be valid, binding and enforceable on the collateral specified in the relevant agreement executed by the respective Reorganized Debtors, and, with respect to the liens granted to provide security for the applicable Debtors' obligations pursuant to the New PD

LLC Notes Documents and the New First Lien Credit Facility Documents, other than as expressly set forth in the New PD LLC Notes Documents and the New First Lien Credit Facility Documents (as applicable), such liens shall be senior to any other liens with respect to the applicable Debtors, including any liens relating to Reinstated Claims; (iii) perform all of their obligations under the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents; and (iv) take all such other actions as any of the Reorganized Debtors may determine are necessary, appropriate or desirable in connection with the consummation of the transactions contemplated by the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents. On the Effective Date, (i) the parties to each New First Lien Credit Facility Document shall be deemed to have executed such New First Lien Credit Facility Document and shall become bound thereto, regardless of whether any party actually executes such New Second Lien Term Loan Document and shall become bound thereto, regardless of whether any party actually executes such New Second Lien Term Loan Document, and (iii) the parties to each New PD LLC Notes Document shall be deemed to have executed such New PD LLC Notes Document and shall become bound thereto, regardless of whether any party actually executes such New PD LLC Notes Document and shall become bound thereto,

13. In addition, on the Effective Date, the Reorganized Lee Debtors shall be authorized to enter into the New First Lien Credit Agreement as well as any notes, documents or agreements delivered in connection therewith, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith. The New First Lien Lenders shall have valid, binding and enforceable liens on the collateral specified in the

New First Lien Credit Facility Documents. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New First Lien Credit Facility Documents are granted in good faith as an inducement to the New First Lien Lenders to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New First Lien Credit Facility Documents.

The New First Lien Revolving Credit Facility shall be a superpriority revolving credit facility with payment priority over the New First Lien Term Loan, and the provisions of the New First Lien Credit Facility Documents setting forth the payment priority of the New First Lien Revolving Credit Facility shall be fully enforceable in accordance with their terms and the parties to the New First Lien Credit Facility Documents shall be enjoined from avoiding, circumventing, recharacterizing or otherwise frustrating the enforcement of the priority provisions thereof.

Moreover, on the Effective Date, the Reorganized Lee Debtors shall be authorized to enter into the New Second Lien Term Loan Agreement, and the Reorganized Lee Debtors and the Reorganized Pulitzer Debtors shall be authorized to enter into any notes, guarantees, documents or agreements delivered, executed, or entered in connection therewith, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith including, without limitation, the New Second Lien Term Loan Documents. The New Second Lien Lenders shall have valid, binding and enforceable liens on the collateral specified in the New Second Lien Term Loan Documents. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New Second Lien Term Loan Documents are granted in good faith as an inducement to the New Second Lien Lenders to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent

transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New Second Lien Term Loan Documents.

- Agreement and issue the New PD LLC Notes, and the Reorganized PD LLC shall be authorized to enter into the New PD LLC Notes Agreement and issue the New PD LLC Notes, and the Reorganized Pulitzer Debtors shall be authorized to enter into any documents or agreements delivered in connection therewith, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith. Other than as expressly set forth in the New PD LLC Notes Documents, the New PD LLC Notes Documents, which shall be senior to any other liens with respect to the Pulitzer Debtors. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New PD LLC Notes Documents and payments made (including, but not limited to, payments of principal of the PD LLC Notes and the Lee Closing Date Payment) with respect to the restructuring of the PD LLC Notes pursuant to, and identified in, the Pulitzer Support Agreement are granted and made in good faith as an inducement to the Holders of the New PD LLC Notes to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New PD LLC Notes Documents.
- 16. Conversion of DIP Revolving Facility Claims. On the Effective Date, in accordance with and on the terms set forth in the DIP Revolving Facility Agreement, the DIP Revolving Facility Claims shall either be (i) indefeasibly paid in full in Cash from the Lee Debtors (and the "Commitments" (as defined in the DIP Revolving Facility Agreement) shall be irrevocably cancelled and the "Letters of Credit" (as defined in the DIP Revolving Credit Facility Agreement)

shall be returned undrawn) in accordance with Section 2.2 of the Plan or (ii) converted into revolving loans and commitments outstanding under the New First Lien Revolving Credit Facility pursuant to the New First Lien Credit Facility Documents.

- 17. Issuance of New Common Stock and New PD LLC Notes. On the Effective Date, (i) as authorized by the Certificate of Incorporation, Reorganized Lee Enterprises shall issue shares of New Common Stock equal to fifteen percent (15%) of the total number of issued and outstanding shares of common stock of Reorganized Lee Enterprises, computed on a pre-issuance basis, together with any and all instruments, certificates and other documents required to be issued or distributed pursuant to the Plan in order to effect such issuance of New Common Stock without further act or action under applicable law, regulation, order or rule, and (ii) Reorganized PD LLC shall issue the New PD LLC Notes for distribution in accordance with the terms of the New PD LLC Notes Documents and the Plan. The issuance and distribution of the New Common Stock and the New PD LLC Notes under or in connection with the Plan shall be, and shall be deemed to be, exempt from registration under any applicable federal or state securities laws to the fullest extent permissible under applicable bankruptcy law and non-bankruptcy law, including, without limitation, section 1145(a) of the Bankruptcy Code. Without limiting the effect of section 1145 of the Bankruptcy Code, all documents, agreements and instruments entered into on or as of the Effective Date contemplated by or in furtherance of the Plan shall become effective and binding in accordance with their respective terms and conditions upon the parties thereto. In addition, all of the shares of New Common Stock and the New PD LLC Notes issued under or in connection with the Plan shall be fully paid and non-assessable and freely tradable under section 1145 of the Bankruptcy Code.
 - 18. Corporate Governance Certificate of Incorporation and By-Laws. On the

Effective Date, the Certificate of Incorporation and the By-Laws for Reorganized Lee Enterprises shall be substantially in the form of Exhibit 1.18 and Exhibit 1.16 to the Plan, respectively. To the extent required by section 1123(a)(6) of the Bankruptcy Code, on the Effective Date, the Certificate of Incorporation shall be amended to prohibit the issuance of non-voting equity securities. The certificates or articles of incorporation, by-laws, certificates of formation, limited liability company agreements, or similar governing documents, as applicable, of the Debtors or Reorganized Debtors other than Reorganized Lee Enterprises shall be amended as necessary to satisfy the provisions of the Plan and the Bankruptcy Code. After the Effective Date, the Reorganized Debtors may amend and restate their certificates or articles of incorporation, by-laws, certificates of formation, limited liability company agreements, partnership agreements or similar governing documents, as applicable, as permitted by applicable law.

19. Corporate Action. On the Effective Date, the amendment of the Certificate of Incorporation, the selection of directors and officers for Reorganized Lee Enterprises and each other Reorganized Debtor, and all other actions contemplated by the Plan shall be deemed authorized and approved in all respects (subject to the provisions of the Plan). All matters provided for in the Plan involving the corporate structure of the Debtors or the Reorganized Debtors, and any corporate action required by the Debtors or the Reorganized Debtors in connection with the Plan, shall be deemed to have timely occurred in accordance with applicable law and shall be in effect, without any requirement of further action by the security holders or directors of the Debtors or the Reorganized Debtors. On and after the Effective Date, the appropriate officers of Reorganized Lee Enterprises and/or the other Reorganized Debtors shall be authorized and directed to issue, execute and deliver the agreements, documents, securities and instruments contemplated

by the Plan in the name of and on behalf of Reorganized Lee Enterprises and/or the other Reorganized Debtors.

- Executory Contracts and Unexpired Leases. As set forth in Article VI of the Plan, on the Effective Date, all executory contracts or unexpired leases of the Debtors shall be deemed assumed in accordance with, and subject to, the provisions and requirements of Sections 365 and 1123 of the Bankruptcy Code, unless such executory contract or unexpired lease (i) was previously assumed or rejected by the Debtors or (ii) previously expired or terminated pursuant to its own terms. Each executory contract and unexpired lease assumed pursuant to Article VI of the Plan shall revest in and be fully enforceable by the respective Reorganized Debtor in accordance with its terms, except as modified by the provisions of the Plan, any order of the Court authorizing and providing for its assumption, or applicable federal law.
- 21. <u>Post-Petition Contracts and Leases</u>. All contracts, agreements and leases that were entered into by the Debtors or assumed by the Debtors after the Petition Date shall be deemed assumed by the applicable Debtor and assigned to the applicable Reorganized Debtor on the Effective Date.
- Distributions and Claims Reconciliation. The provisions of Article VII and Article VIII of the Plan regarding the distributions under the Plan and the reconciliation of Disputed Claims are hereby approved in all respects. In addition, no party shall be required to file a proof of claim on account of Claims held against the Debtors. On the Effective Date, any and all proofs of claim submitted to the Voting Agent and/or filed with the Bankruptcy Court shall be deemed expunged from the record of these Chapter 11 Cases, and the Debtors shall not be required to file any response to such proofs of claim with the Bankruptcy Court. The Debtors are hereby authorized to make distributions on account of Claims in accordance with the books and records of

the Debtors. To the extent the Holder of a Claim disagrees with the Debtors' books and records with respect to the Allowed amount of such Holder's Claim, then such Holder must advise the Debtors of such disagreement in writing, in which event such Holder's Claim shall become a Disputed Claim. From and after the Effective Date, the Reorganized Debtors shall have sole responsibility and authority for disputing, objecting to, compromising and settling, or otherwise resolving and making distributions (if any) with respect to all Claims, including all Disputed Claims and Administrative Expense Claims. For the avoidance of doubt, nothing in this paragraph shall relieve any party of its obligation to comply with the procedures set forth in Section 12.3 of the Plan with respect to the payment of Administrative Expense Claims.

- Binding Effect of the Plan. On the Effective Date, except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, all provisions of the Plan, including all agreements, instruments and other documents filed in connection with the Plan and executed by the Debtors or the Reorganized Debtors in connection with the Plan, shall be binding upon the Debtors, the Reorganized Debtors, each Holder of a Claim against or Interest in each of the Debtors and such Holder's respective successors and assigns, whether or not the Claim or Interest of such Holder is Impaired under the Plan and whether or not such Holder has accepted the Plan, and all other parties that are affected in any manner by the Plan. All agreements, instruments and other documents filed in connection with the Plan shall be given full force and effect, and shall bind all parties referred to therein as of the Effective Date, whether or not such agreements are actually issued, delivered or recorded on the Effective Date or thereafter and whether or not a party has actually executed such agreement.
- 24. <u>Discharge of Non-Reinstated Claims</u>. On the Effective Date, the Debtors shall be deemed discharged and released under section 1141(d)(l)(A) of the Bankruptcy Code from

any and all Claims not Reinstated under the Plan, including, but not limited to, demands and liabilities that arose before the Effective Date, and all debts of the kind specified in sections 502(g), 502(h) or 502(i) of the Bankruptcy Code. Except as otherwise provided in the Plan (including with respect to all Claims that are Reinstated under the Plan) or in this Confirmation Order, all consideration distributed under the Plan shall be in exchange for, and in complete satisfaction, settlement, discharge and release of, all Claims of any nature whatsoever against the Debtors or any of their Estates, assets, properties or interests in property. Notwithstanding anything in Section 10.1.1 of the Plan or this Confirmation Order to the contrary, the DIP Revolving Facility shall be converted into the New First Lien Revolving Credit Facility, subject to the terms thereof, including the satisfaction of the conversion conditions set forth therein.

- 25. <u>Non-Discharge of Interests</u>. No Interests in the Debtors shall be discharged by the Plan or this Confirmation Order.
- Discharge Injunction. As of the Effective Date, except as otherwise expressly provided in the Plan or this Confirmation Order, all Entities shall be precluded from asserting against the Debtors or the Reorganized Debtors and their respective assets, property and Estates, any other or further Claims (other than those Reinstated under the Plan), or any other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities of any nature whatsoever, relating to any of the Debtors or Reorganized Debtors or any of their respective assets, property and Estates, based upon any act, omission, transaction or other activity of any nature that occurred prior to the Effective Date. In accordance with the foregoing, except as expressly provided in the Plan or this Confirmation Order, this Confirmation Order shall constitute a judicial determination, as of the Effective Date, of the discharge of all non-Reinstated Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities,

pursuant to sections 524 and 1141 of the Bankruptcy Code, and such discharge shall void and extinguish any judgment obtained against the Debtors, the Reorganized Debtors, or any of their respective assets, property and Estates at any time, to the extent such judgment is related to a discharged Claim, debt or liability.

Releases by the Debtors. Except as otherwise expressly provided in the Plan or this Confirmation

Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each of the Debtors and Reorganized Debtors on its own behalf and as a representative of its respective Estate, shall, and shall be deemed to, completely and forever release, waive, void, extinguish and discharge unconditionally, each and all of the Released Parties of and from any and all Claims and causes of action (including, without limitation, Avoidance Actions), any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to any of the Debtors, the Reorganized Debtors, their respective assets, property and Estates, the Chapter 11 Cases, the Lee Support Agreement, the Pulitzer Support Agreement, the Plan, or the Solicitation and Disclosure Statement that may be asserted by or on behalf of any of the Debtors, the Reorganized Debtors or their respective Estates against any of the Released Parties; provided, however, that nothing in Section 10.2 of the Plan or in this Confirmation Order shall be construed to release any Released Party from willful misconduct or gross negligence as

determined by a Final Order.

28. Releases by Certain Holders of Claims. Except as otherwise expressly provided in the Plan or this Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, (i) each Holder of a Claim entitled to vote on the Plan that did not elect, by submitting a Ballot and checking the appropriate box thereon, not to grant the releases set forth in Section 10.3 of the Plan and (ii) each Holder of Claim bound by the Lee Support Agreement or the Pulitzer Support Agreement to grant the releases set forth in Section 10.3 of the Plan (regardless of whether such Holder submitted a Ballot containing an election not to grant the releases set forth in Section 10.3 of the Plan) shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the Debtors and their respective Related Persons of and from any and all Claims, any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to any of the Debtors, the Reorganized Debtors or their respective assets, property and Estates, the Chapter 11 Cases, the Plan, and/or the Disclosure Statement; provided, however, that nothing in Section 10.3 of the Plan shall be construed to release the Debtors or their respective Related Persons from willful misconduct or gross negligence as determined by a Final Order. For the avoidance of doubt, nothing in this paragraph shall release the

Reorganized Debtors and their respective Related Persons from their obligations under the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, or the New PD LLC Notes Documents.

- Exculpation. From and after the Effective Date, the Released Parties shall neither have nor incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, formulating, negotiating or implementing the Lee Support Agreement, the Pulitzer Support Agreement, the Plan, and/or the Solicitation and Disclosure Statement, the solicitation of acceptances of the Plan, the pursuit of Confirmation of the Plan, the Confirmation of the Plan, the consummation of the Plan, the administration of the Plan, the property to be distributed under the Plan, or any other act taken or omitted to be taken in connection with or in contemplation of the Chapter 11 Cases or implementation of the Plan; provided, however, that Section 10.4 of the Plan shall not apply to release (x) obligations under the Plan, and obligations under the contracts, instruments, releases, agreements, and documents delivered, Reinstated or assumed under the Plan, and (y) any claims or causes of action arising out of willful misconduct or gross negligence as determined by a Final Order. Any of the Released Parties shall be entitled to rely, in all respects, upon the advice of counsel with respect to their duties and responsibilities under the Plan.
- 30. <u>Injunction Related to Exculpation</u>. Except as expressly provided in the Plan or this Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, cause of

action or liability of any nature whatsoever, of the types described in Section 10.4 of the Plan and relating to any of the Debtors or the Reorganized Debtors or any of their respective assets, property and/or Estates, are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any Released Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit, action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 10.1 of the Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is inconsistent with the provisions of the Plan or this Confirmation Order.

31. <u>Survival of Indemnification Obligations</u>. The obligations of the Debtors to indemnify any past and present directors, officers, agents, employees and representatives, pursuant to certificates or articles of incorporation, by-laws, contracts and/or applicable statutes, in respect of all actions, suits and proceedings against any of such officers, directors, agents, employees and representatives, based upon any act or omission related to service with, for or on behalf of the Debtors, shall not be discharged or impaired by this Confirmation Order or consummation of the

Plan and shall be assumed by the Reorganized Debtors.

- 32. <u>Term of Bankruptcy Injunction or Stays</u>. All injunctions or stays in effect in the Chapter 11 Cases under sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.
- 33. Exemption from Transfer Taxes. Pursuant to section 1146(a) of the Bankruptcy Code, (a) the issuance, transfer or exchange of notes or equity securities under the Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; (c) the making or assignment of any lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under the Plan, including, without limitation, merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any stamp tax or other similar tax.
- 34. <u>Surrender of Instruments</u>. As a condition to participation in distributions under the Plan, unless the Debtors shall otherwise agree, each PD LLC Noteholder and the Holder(s) of any evidence of indebtedness of the Debtors relating to a non-Reinstated Claim that desires to receive the property to be distributed on account of an Allowed non-Reinstated Claim based on such PD LLC Note or evidence of indebtedness shall surrender such PD LLC Note or evidence of indebtedness to the Debtors, or their designee, and shall execute and deliver such other documents as are necessary to effectuate the Plan. Except as otherwise provided in Section 12.1 of the Plan (or except as the Debtors otherwise agree), if no surrender of a PD LLC Note or evidence of indebtedness relating to a non-Reinstated Claim occurs and a claimant does not provide an affidavit and indemnification agreement, in form and substance satisfactory to the Debtors, that such PD LLC Note or evidence of indebtedness was lost, then no distribution may be made to any claimant whose Claim is based on such security, note, debenture or evidence of indebtedness thereof. The

Debtors shall make subsequent distributions only to the persons who surrender PD LLC Notes or evidence of indebtedness, as applicable, for exchange (or their assignees) and the record holders of such PD LLC Notes or other indebtedness shall be those holders of record as of the Distribution Record Date.

- 35. <u>Post-Confirmation Date Retention of Professionals</u>. On the Effective Date, any requirement that Professionals employed by the Reorganized Debtors comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date will terminate, and the Reorganized Debtors will be authorized to employ and compensate professionals in the ordinary course of business and without the need for application to or approval by the Bankruptcy Court.
- Bar Date for Certain Administrative Expense Claims. All applications for final allowance of compensation or reimbursement of expenses incurred by any Professional, and all other requests for the payment of Administrative Expense Claims, including all requests for the allowance of any Administrative Expense Claim pursuant to section 503(b)(3)(D) of the Bankruptcy Code for substantial contributions made in the Chapter 11 Cases (but excluding all requests for the payment of obligations incurred by the Debtors in the ordinary course of their business operations after the Petition Date), must be filed with the Bankruptcy Court and served on the Reorganized Debtors and their counsel at the addresses set forth in Section 12.14 of the Plan not later than thirty (30) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court. Any request for the payment of an Administrative Expense Claim that is not timely filed and served shall be discharged and forever barred and the Holder of such Administrative Expense Claim shall be enjoined from commencing or continuing any action, process, or act to collect, offset or recover such Claim. The Debtors and the Reorganized Debtors shall have sole responsibility for

filing objections to and resolving all requests for the allowance of Administrative Expense Claims.

- 37. <u>Effectuating Documents and Further Transactions</u>. Each of the Debtors and the Reorganized Debtors is authorized to execute, deliver, file or record such contracts, instruments, certificates, notes, releases and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement and further evidence the terms and conditions of the Plan and the New Common Stock and New PD LLC Notes issued under or in connection with the Plan.
- Resolution of the IRS Objection. Notwithstanding any provision to the contrary in the Plan, this Confirmation Order, and any implementing Plan documents, nothing shall: (1) affect the ability of the IRS to pursue any non-Debtors to the extent allowed by non-bankruptcy law for any liabilities that may be related to any federal tax liabilities owed by the Debtors; (2) affect the rights of the IRS to assert setoff and recoupment and such rights are expressly preserved; (3) discharge any claim of the IRS described in section 1141(d)(6) of the Bankruptcy Code; or (4) require the IRS to file an administrative expense claim in order to receive payment for any liability described in sections 503(b)(1)(B) and 503(b)(1)(C) of the Bankruptcy Code in accordance with section 503(b)(1)(D) of the Bankruptcy Code.
- 39. <u>Payment of Statutory Fees</u>. All fees payable pursuant to section 1930 of title 28 of the United States Code shall be paid as and when due or as otherwise agreed by the United States Department of Justice, Office of the United States Trustee.
- 40. <u>Payment of Consenting Lender Fees, Consenting Noteholder Fees, and Backstop Lender Fees.</u>

 Notwithstanding anything to the contrary in the Plan, the Consenting Lender Fees, the Consenting Noteholder Fees, and the Backstop Lender Fees incurred in connection with the Lee Support Agreement, the Pulitzer Support Agreement, and Backstop Commitment

Letters, respectively, shall be paid in full, in Cash, on the Effective Date to the Consenting Lenders, the Consenting Noteholders, and the Backstop Lenders, as applicable, and the Debtors shall pay such other fees and expenses on the Effective Date as may have been agreed to by the Debtors prior to the Effective Date, including, without limitation, the PD LLC Noteholder Fees and any fees or expenses required to be paid under any Backstop Commitment Letter.

- 41. <u>Termination of Lee Support Agreement and Pulitzer Support Agreement</u>. On the Effective Date, the Lee Support Agreement and the Pulitzer Support Agreement shall terminate in accordance with Sections 2.1(b)(i) and 2.1(i) thereof, respectively.
- 42. <u>Substantial Consummation</u>. The substantial consummation of the Plan, within the meaning of section 1127 of the Bankruptcy Code, shall be deemed to occur on the Effective Date.
- 43. Retention of Jurisdiction. Except as expressly provided in the Plan, pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of this Confirmation Order and the occurrence of the Effective Date, the Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, exclusive jurisdiction for the purposes set forth in Article XI of the Plan, subject to the provisions of Section 11.2 of the Plan.
- 44. Reference to and Validity and Enforceability of Plan Provisions. The failure to reference any particular provision of the Plan in this Confirmation Order shall have no effect on the binding effect, enforceability or legality of such provisions and such provisions shall have the same binding effect, enforceability or legality as every other provision of the Plan. Each term and provision of the Plan is valid and enforceable pursuant to its terms and is specifically approved pursuant to this Confirmation Order even if not referenced herein.

- Notice of Entry of this Confirmation Order. Pursuant to Bankruptcy Rules 2002(f)(7) and 3020(c), no later than ten (10) Business Days after the entry of this Confirmation Order, the Debtors shall cause a notice of the entry of this Confirmation Order to be served on all Holders of Claims or Interests and any other Persons on whom the Confirmation Hearing Notice was served. In addition, pursuant to Bankruptcy Rule 2002(l), the Debtors shall publish a notice of the entry of this Confirmation Order once in the national editions of The Wall Street Journal and USA Today, and once in the daily edition of the St. Louis Post-Dispatch, not later than ten (10) business days after the occurrence of the Effective Date. Such actual and publication notice of entry of this Confirmation Order shall have the effect of an order of the Court and shall constitute good and sufficient notice of the entry of this Confirmation Order and of the relief granted herein pursuant to Bankruptcy Rules 2002(f)(7) and 3020(c), and no other or further notice of entry of this Confirmation Order need be given.
- A6. Reversal or Vacatur. If any or all of the provisions of this Confirmation Order are hereafter reversed, modified or vacated by subsequent order of this Court or any other court, such reversal, modification or vacatur shall not affect the validity of the acts or obligations incurred or undertaken under or in connection with the Plan or the validity, priority or enforceability of any liens securing such obligations or intercreditor arrangements entered into in connection with the Plan prior to the Debtors' receipt of written notice of any such order. Notwithstanding any such reversal, modification, or vacatur of this Confirmation Order, any such act or obligation incurred or undertaken pursuant to, and in reliance on, this Confirmation Order prior to the effective date of such reversal, modification or vacatur shall be governed in all respects by the provisions of this Confirmation Order and the Plan or any amendments or modifications thereto.

Authorization to Consummate the Plan. This Confirmation Order is a final order and the period in which an appeal must be filed shall commence upon the entry hereof. The 14-day stay of this Confirmation Order under Bankruptcy Rule 3020(e) is hereby waived. The Debtors are authorized to consummate the Plan at any time after the entry of this Confirmation Order, subject to the satisfaction or waiver of the conditions precedent to occurrence of the Effective Date as set forth in Section 9.2 of the Plan.

Dated: Wilmington, Delaware January 23, 2012

/s/ Kevin Gross

THE HONORABLE KEVIN GROSS
CHIEF UNITED STATES BANKRUPTCY JUDGE

UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	Chapter 11
LEE ENTERPRISES, INCORPORATED,	
<u>et al.</u> , ¹	Case No. 11-13918 (KG)
Debtors.	Jointly Administered

SECOND AMENDED JOINT PREPACKAGED PLAN OF REORGANIZATION FOR LEE ENTERPRISES, INCORPORATED AND ITS DEBTOR SUBSIDIARIES

SIDLEY AUSTIN LLP One South Dearborn Street Chicago, Illinois 60603 Larry J. Nyhan Kenneth P. Kansa Bojan Guzina Kerriann S. Mills

Telephone: (312) 853-7000 Facsimile: (312) 853-7036

YOUNG CONAWAY STARGATT & TAYLOR, LLP The Brandywine Building 1000 West Street, 17th Floor P.O. Box 391 Wilmington, Delaware 19899-0391 Robert S. Brady (No. 2847) Edwin J. Harron (No. 3396) Edmon L. Morton (No. 3856)

Telephone: (302) 571-6600 Facsimile: (302) 571-1253

Counsel to the Debtors and Debtors in Possession Dated: January 19, 2012

¹ The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, where applicable, are: Lee Enterprises, Incorporated (3980); Accudata, Inc. (3648); Fairgrove LLC (N/A); Flagstaff Publishing Co. (4796); Hanford Sentinel, Inc. (0775); HomeChoice, LLC (3701); HSTAR LLC (N/A); INN Partners, L.C. (9478); Journal-Star Printing Co. (6412); Kauai Publishing Co. (7036); K. Falls Basin Publishing, Inc. (3285); Lee Consolidated Holdings Co. (2158); Lee Procurement Solutions Co. (6292); Lee Publications, Inc. (1886); Napa Valley Publishing Co. (7802); NIPC, Inc. (9541); NLPC LLC (7381); Northern Lakes Publishing Co. (9679); NVPC LLC (N/A); Pantagraph Publishing Co. (7058); Pulitzer Missouri Newspapers, Inc. (1960); Pulitzer Network Systems LLC (5359); Pulitzer Newspapers, Inc. (1560); Pulitzer Technologies, Inc. (8892); Pulitzer Utah Newspapers, Inc. (1884); Pulitzer Inc. (9711); Santa Maria Times, Inc. (3801); SHTP LLC (N/A); Sioux City Newspapers, Inc. (9191); SOPC LLC (5551); Southwestern Oregon Publishing Co. (0741); St. Louis Post-Dispatch LLC (5357); STL Distribution Services LLC (0922); Suburban Journals of Greater St. Louis LLC (6217); and Ynez Corporation (5443). The Debtors' corporate headquarters and the service address for each Debtor is 201 N. Harrison Street, Suite 600, Davenport, IA 52801.

TABLE OF CONTENTS

PAGE

31

DEFINED TERMS AND RULES OF INTERPRETATION	1
TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS, DIP REVOLVING FACILITY CLAIMS, PRIORITY TAX CLAIMS, AND SECTION 507(b) CLAIMS	16
Administrative Expense Claims	16
DIP Revolving Facility Claims	17
Priority Tax Claims	17
Section 507(b) Claims	17
CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS	18
Summary of Classification and Treatment of Classified Claims and Interests.	18
Classification and Treatment of Claims Against and Interests in the Lee Debtors (Debtors 1-9).	19
Classification and Treatment of Claims Against and Interests in the Pulitzer Debtors (Debtors 10-35)	21
Unimpaired Claims and Interests	24
ACCEPTANCE OR REJECTION OF THE PLAN	24
Impaired Classes of Claims Entitled to Vote	24
Acceptance by an Impaired Class	24
Presumed Acceptances by Unimpaired Classes	24
Confirmability and Severability of this Plan	24
MEANS FOR IMPLEMENTATION OF THE PLAN	25
Non-Substantive Consolidation	25
Corporate Governance, Directors, Officers and Corporate Action	25
Issuance and Distribution of New Securities	26
Reporting Requirements Under Securities Exchange Act of 1934	27
New Debt Documents	27
Conversion or Repayment of DIP Revolving Facility Claims	28
New Second Lien Term Loan Election	28
New First Lien Credit Facility	28
New Second Lien Term Loan	28
	29
	29
	29
Cancellation of Certain Credit and Debt Documents	30
	30
Additional Transactions Authorized Under this Plan	31
	TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS, DIP REVOLVING FACILITY CLAIMS, PRIORITY TAX CLAIMS, AND SECTION 507(b) CLAIMS Administrative Expense Claims DIP Revolving Facility Claims Priority Tax Claims Section 507(b) Claims CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS Summary of Classification and Treatment of Classified Claims and Interests. Classification and Treatment of Claims Against and Interests in the Lee Debtors (Debtors 1-9). Classification and Treatment of Claims Against and Interests in the Pulitzer Debtors (Debtors 10-35) Unimpaired Claims and Interests ACCEPTANCE OR REJECTION OF THE PLAN Impaired Classes of Claims Entitled to Vote Acceptance by an Impaired Classe Presumed Acceptances by Unimpaired Classes Confirmability and Severability of this Plan MEANS FOR IMPLEMENTATION OF THE PLAN Non-Substantive Consolidation Corporate Governance, Directors, Officers and Corporate Action Issuance and Distribution of New Securities Reporting Requirements Under Securities Exchange Act of 1934 New Debt Documents Conversion or Repayment of DIP Revolving Facility Claims New Second Lien Term Loan Election New First Lien Credit Facility New Second Lien Term Loan New PD LLC Notes Binding Effect Continued Corporate Existence and Vesting of Assets in the Reorganized Debtors

i

EMPLOYEE BENEFIT PLANS

ARTICLE VI:

TREATMENT OF EXECUTORY CONTRACTS, UNEXPIRED LEASES, INSURANCE POLICIES AND

6.1.	Assumption of Executory Contracts and Unexpired Leases	31
6.2.	Cure of Defaults Under Assumed Executory Contracts and Unexpired Leases	31
6.3.	Assumption of Collective Bargaining Agreements	31
6.4.	Insurance Policies and Agreements	32
6.5.	Employee Compensation and Benefit Plans	32
6.6.	Retiree Benefits and Pension Plans	32
6.7.	Post-Petition Contracts and Leases	32
ARTICLE VII:	PROVISIONS GOVERNING DISTRIBUTIONS	32
7.1.	Distributions on Account of Claims Allowed as of the Effective Date	33
7.2.	Distributions on Account of Claims that Become Allowed after the Effective Date	33
7.3.	Distribution Date	33
7.4.	Interest on Claims	33
7.5.	Distributions by Disbursing Agent(s)	33
7.6.	Delivery of Distributions and Undeliverable or Unclaimed Distributions	33
7.7.	Record Date for Distributions	34
7.8.	Allocation of Plan Distributions Between Principal and Interest	34
7.9.	Means of Cash Payment	34
7.10.	Withholding and Reporting Requirements	34
7.11.	Setoff and Recoupment	35
7.12.	Fractional Shares	35
ARTICLE VIII:	PROCEDURES FOR RESOLVING DISPUTED CLAIMS	35
8.1.	Objection to and Estimation of Claims	35
8.2.	No Distributions Pending Allowance	35
8.3.	Distributions on Account of Disputed Claims Once They Are Allowed	35
8.4.	Reinstated Claims and Interests	36
ARTICLE IX:	CONFIRMATION AND CONSUMMATION OF THE PLAN	36
9.1.	Condition to Confirmation	36
9.2.	Conditions to Effective Date	36
9.3.	Waiver of Conditions	37
9.4.	Vacation of Confirmation Order	37
9.5.	Notice of Effective Date	37
ARTICLE X:	EFFECT OF PLAN CONFIRMATION	37
10.1.	Discharge	37
10.2.	Releases by the Debtors	38
10.3.	Releases by Certain Holders of Claims	38
10.4.	Exculpation	39
10.5.	Injunction Related to Exculpation	39
10.6.	Survival of Indemnification Obligations	40

40

Term of Bankruptcy Injunction or Stays

10.7.

ARTICLE XI:	RETENTION OF JURISDICTION	40
ARTICLE XII:	MISCELLANEOUS PROVISIONS	42
12.1.	Surrender of Instruments	42
12.2.	Post-Confirmation Date Retention of Professionals	42
12.3.	Bar Date for Certain Administrative Expense Claims	42
12.4.	Effectuating Documents and Further Transactions	43
12.5.	Corporate Action	43
12.6.	Exemption from Transfer Taxes	43
12.7.	Payment of Statutory Fees	43
12.8.	Payment of Consenting Lender Fees, Consenting Noteholder Fees, and Backstop Lender Fees	43
12.9.	Support Agreements	44
12.10.	Amendment or Modification of this Plan	44
12.11.	Severability of Plan Provisions	44
12.12.	Successors and Assigns	44
12.13.	Revocation, Withdrawal or Non-Consummation	44
12.14.	Notice	45
12.15.	Governing Law	45
12.16.	Tax Reporting and Compliance	45
12.17.	Exhibits	46
12.18.	Filing of Additional Documents	46
12.19.	Reservation of Rights	46

EXHIBITS

Exhibit 1.16	-	Amended and Restated By-Laws of Reorganized Lee Enterprises
Exhibit 1.18	_	Amended and Restated Certificate of Incorporation of Reorganized Lee Enterprises
Exhibit 1.73	_	New First Lien Credit Agreement (inclusive of all exhibits thereto)
Exhibit 1.79	_	New Lee Intercreditor Agreement
Exhibit 1.80	_	New PD LLC Intercreditor Agreement
Exhibit 1.83	_	New PD LLC Notes Agreement (inclusive of all exhibits thereto)
Exhibit 1.85	_	New PD LLC Notes Subsidiary Guaranty Agreement
Exhibit 1.86	_	New Pulitzer Guaranty Agreement
Exhibit 1.89	_	New Second Lien Term Loan Agreement (inclusive of all exhibits thereto)
Exhibit 5.2.2	_	Directors and Officers of Reorganized Lee Enterprises
Exhibit 5.2.3	_	Directors and Officers of the Reorganized Debtors other than Reorganized Lee Enterprises

INTRODUCTION

Lee Enterprises, Incorporated ("<u>Lee Enterprises</u>") and those Affiliates of Lee Enterprises listed in footnote 1 hereto hereby propose the following joint plans of reorganization for their reorganization cases under Chapter 11 of the Bankruptcy Code for the resolution of the outstanding Claims against and Interests in each of the Debtors. Capitalized terms used but not defined in this paragraph have the meanings assigned to them in Article I of this Plan. The classification and treatment of Claims against and Interests in the Debtors is set forth in Article II and Article III of this Plan. The Debtors are the proponents of this Plan within the meaning of section 1129 of the Bankruptcy Code. Reference is made to the Disclosure Statement, distributed contemporaneously herewith, for a discussion of the Debtors' history, business, properties and operations, projections for those operations, risk factors, a summary and analysis of this Plan, and related matters.

ARTICLE I: DEFINED TERMS AND RULES OF INTERPRETATION

- A. <u>Defined Terms</u>. As used in this Plan, capitalized terms shall have the meanings set forth in this Article I. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.
- 1.1 Administrative Expense Claim means a Claim for costs and expenses of administration of the Chapter 11 Cases arising on or after the Petition Date and prior to the Effective Date under sections 328, 330, 363, 365, 503(b), or 507(a)(2) of the Bankruptcy Code, including, without limitation, (a) any actual and necessary costs and expenses of preserving the Estates and operating the businesses of the Debtors from and after the Petition Date (such as wages, salaries and commissions for services and payments for inventory, leased equipment and premises) and Claims of governmental units for taxes (including tax audit Claims) related to tax years commencing after the Petition Date, but excluding Claims related to tax periods, or portions thereof, ending on or before the Petition Date; (b) all compensation for actual and necessary legal, financial, advisory, accounting and other services provided by the Professionals and the reimbursement of actual and necessary expenses incurred by the Professionals pursuant to sections 328 or 330 of the Bankruptcy Code; (c) with the exception of Section 507(b) Claims, any indebtedness or obligations incurred or assumed by the Debtors during the Chapter 11 Cases; (d) any payment to be made under this Plan or otherwise to cure a default under an executory contract or unexpired lease that has been or will be assumed by any of the Debtors; or (e) any fees and charges assessed against the Estates under section 1930, Chapter 123, of Title 28 of the United States Code.
- 1.2 <u>Affiliate</u> has the meaning assigned to such term in section 101(2) of the Bankruptcy Code and when used in this Plan with reference to any Debtor shall include, but not be limited to, each of the other Debtors.
- 1.3 <u>Allowed</u> means, with respect to any Claim or Interest, or any portion thereof, in any Class or category specified, a Claim or Interest arising on or before the Effective Date (a) (i) as to which no objection to allowance has been interposed in accordance with Section 8.1 of this

Plan or (ii) as to which any objection has been settled, waived, withdrawn or denied by a Final Order; or (b) that is expressly allowed (i) by a Final Order, (ii) by an agreement between the Holder of such Claim or Interest and the Debtors or the Reorganized Debtors, or (iii) pursuant to the terms of this Plan; <u>provided</u>, <u>however</u>, that, as to subparagraph (a) only, "Allowed" shall not include any Claim subject to disallowance in accordance with section 502(d) of the Bankruptcy Code.

- 1.4 <u>Allowed Claim or Interest means a Claim or Interest in a particular Class or of a particular type that is also an Allowed Claim or Interest.</u> For example, an Allowed Administrative Expense Claim is an Administrative Expense Claim that is also an Allowed Claim.
- 1.5 <u>Avoidance Actions</u> means causes of action arising under sections 542, 544, 545, 547, 548, 549, 550, 551 or 553(b) of the Bankruptcy Code, or under similar or related state or federal statutes or common law, including fraudulent transfer laws, in each case whether or not litigation to prosecute such causes of action was commenced prior to the Effective Date.
- 1.6 <u>Backstop Cash</u> means an amount of Cash, up to an aggregate amount of \$10,000,000, that one or more Backstop Lenders may pay to Reorganized Lee Enterprises on the Effective Date to acquire the portion of the New Second Lien Term Loan that such Backstop Lender has committed to acquire pursuant to the terms and conditions of the applicable Backstop Commitment Letter.
- 1.7 <u>Backstop Commitment Amount(s)</u> means, individually or collectively, the maximum amount of Prepetition Credit Agreement Claims that each Backstop Lender has committed, pursuant to the terms and conditions of the relevant Backstop Commitment Letter, to convert (and, if necessary, pay Backstop Cash) to acquire up to \$175,000,000, which amount includes an original issue discount of 5%, of the New Second Lien Term Loan Facility.
- 1.8 Backstop Commitment Letter(s) means, individually or collectively, (a) that certain Amended and Restated Backstop Commitment Letter between Lee Enterprises and Goldman Sachs Lending Partners LLC, dated December 2, 2011, (b) that certain Amended and Restated Backstop Commitment Letter between Lee Enterprises and Mutual Quest Fund, dated December 2, 2011, (c) that certain Amended and Restated Backstop Commitment Letter between Lee Enterprises and Monarch Master Funding Ltd, dated December 2, 2011, (d) that certain Backstop Commitment Letter between Lee Enterprises and Mudrick Distressed Opportunity Fund Global, LP, dated December 2, 2011, and (e) that certain Backstop Commitment Letter between Lee Enterprises and Blackwell Partners, LLC, dated December 2, 2011.
- 1.9 <u>Backstop Lender(s)</u> means, individually or collectively, the Initial Backstop Lenders and any assignee thereof, in accordance with the terms of the applicable Backstop Commitment Letter.
- 1.10 <u>Backstop Lender Fees</u> means commitment fees equal to 0.5% of the Backstop Lenders' aggregate Backstop Commitment Amounts, incurred by the Lee Debtors in connection with the transactions contemplated by the Backstop Commitment Letters.

- 1.11 <u>Ballot</u> means the ballot form for accepting or rejecting this Plan and making certain elections under this Plan, distributed with the Disclosure Statement to the Holders of Claims that are Impaired under this Plan and entitled to vote to accept or reject this Plan pursuant to <u>Article III</u> and <u>Article IV</u> hereof.
- 1.12 <u>Bankruptcy Code</u> means Title 11 of the United States Code, 11 U.S.C. §§ 101 through 1532, as in effect on the Petition Date, together with any amendments and modifications thereto that may subsequently be made applicable to the Chapter 11 Cases.
- 1.13 <u>Bankruptcy Court</u> means the United States Bankruptcy Court for the District of Delaware or any other court with jurisdiction over the Chapter 11 Cases.
- 1.14 <u>Bankruptcy Rules</u> means the Federal Rules of Bankruptcy Procedure promulgated under section 2075 of Title 28 of the United States Code and any local rules of the Bankruptcy Court, as in effect on the Petition Date, together with any amendments and modifications thereto that may subsequently be made applicable to the Chapter 11 Cases.
 - 1.15 <u>Business Day</u> means any day other than a Saturday, a Sunday or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).
- 1.16 <u>By-Laws</u> means the Amended By-Laws of Lee Enterprises, effective May 17, 2007, together with any amendments thereto, which shall be substantially in the form of <u>Exhibit 1.16</u> to this Plan.
 - 1.17 Cash means legal tender of the United States of America.
- 1.18 <u>Certificate of Incorporation</u> means the Restated Certificate of Incorporation of Lee Enterprises, as of March 3, 2005, as amended in accordance with Section 5.2.1 of this Plan. The Certificate of Incorporation shall be substantially in the form of <u>Exhibit 1.18</u> to this Plan.
- 1.19 <u>Chapter 11 Cases</u> means the voluntary cases under Chapter 11 of the Bankruptcy Code commenced by the Debtors in the Bankruptcy Court on the Petition Date.
 - 1.20 <u>Claim</u> means a "claim," as defined in section 101(5) of the Bankruptcy Code.
- 1.21 <u>Class</u> means each category of Holders of Claims or Interests established under <u>Article III</u> of this Plan pursuant to sections 1122 and 1123(a)(1) of the Bankruptcy Code.
- 1.22 <u>Collective Bargaining Agreements</u> means all collective bargaining agreements to which any of the Debtors is a party on the Confirmation Date.
 - 1.23 <u>Confirmation</u> means the entry of the Confirmation Order by the Bankruptcy Court.
- 1.24 <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the Bankruptcy Court's docket.

- 1.25 <u>Confirmation Hearing</u> means the hearing held by the Bankruptcy Court on confirmation of this Plan, as such hearing may be continued from time to time.
- 1.26 <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming this Plan pursuant to section 1129 of the Bankruptcy Code, which order shall be in form and substance reasonably satisfactory to the Prepetition Administrative Agent, the Initial Backstop Lenders, and the Required Consenting Noteholders.
- 1.27 <u>Consenting Lenders</u> means (a) those certain Holders of Prepetition Credit Agreement Claims that are signatories to the Lee Support Agreement and (b) any Joining Lender Party.
- 1.28 <u>Consenting Lender Fees</u> means the balance of the consent fees not paid to the Consenting Lenders prior to the Petition Date, as set forth in section 4.6 of the Lee Support Agreement and incurred by the Lee Debtors in connection with the transactions contemplated by the Lee Support Agreement.
- 1.29 <u>Consenting Noteholders</u> means (a) those certain Holders of PD LLC Notes Claims that are signatories to the Pulitzer Support Agreement and (b) any Joining Noteholder Party.
- 1.30 <u>Consenting Noteholder Fees</u> means the balance of the consent fees not paid to the Consenting Noteholders prior to the Petition Date, as set forth in section 4.6 of the Pulitzer Support Agreement and incurred by the Pulitzer Debtors in connection with the transactions contemplated by the Pulitzer Support Agreement.
- 1.31 <u>Converting Lenders</u> means those certain Holders (or an advisor, nominee, or investment manager for beneficial holder(s)) of Prepetition Credit Agreement Claims that elect to convert all or a portion of their funded Prepetition Credit Agreement Claims into an applicable share of the New Second Lien Term Loan by making the New Second Lien Term Loan Election and, for the avoidance of doubt, includes each Backstop Lender.
 - 1.32 <u>DBSI</u> means Deutsche Bank Securities Inc.
 - 1.33 <u>DBTCA</u> means Deutsche Bank Trust Company Americas.
 - 1.34 <u>Debtor(s)</u> means, individually or collectively, the debtors and debtors in possession identified in footnote 1 hereto.
- 1.35 <u>DIP Revolving Facility Agent</u> means DBTCA in its capacity as the administrative agent under the DIP Revolving Facility Agreement.
- 1.36 <u>DIP Revolving Facility Agreement</u> means that certain Credit and Guaranty Agreement, by and among Lee Enterprises, as borrower, the Lee Subsidiary Debtors, as guarantors, the DIP Revolving Facility Agent, and the DIP Revolving Facility Lenders, as the same may be amended, modified, or supplemented from time to time.

- 1.37 <u>DIP Revolving Facility Claims</u> means all Claims against the Lee Debtors held by the DIP Revolving Facility Agent and the DIP Revolving Facility Lenders pursuant to the DIP Revolving Facility Documents and the interim order and Final Order approving the Lee Debtors' entry into and performance of their obligations under the DIP Revolving Facility Documents.
- 1.38 <u>DIP Revolving Facility Commitment Letter</u> means that certain commitment letter, entered into by and among Lee Enterprises, DBTCA, DBSI, and the other "Commitment Parties" (as defined therein), including all exhibits and attachments thereto.
- 1.39 <u>DIP Revolving Facility Documents</u> means collectively, the DIP Revolving Facility Agreement and all other agreements (including, without limitation, any guaranty agreements, pledge and collateral agreements, and intercompany subordination agreements), documents and instruments delivered in connection therewith.
- 1.40 <u>DIP Revolving Facility Lenders</u> means the "Commitment Parties" (and their successors and assigns) as defined in and pursuant to the DIP Revolving Facility Commitment Letter.
- 1.41 <u>Disallowed Claim</u> means any Claim, including any portion thereof, that has been disallowed, denied, dismissed, expunged, or overruled pursuant to a Final Order of the Bankruptcy Court or any other court of competent jurisdiction.
 - 1.42 <u>Disbursing Agent(s)</u> means any Entity in its capacity as a disbursing agent under this Plan.
- 1.43 <u>Disclosure Statement</u> means the disclosure statement relating to this Plan, including, without limitation, all exhibits and schedules thereto, as the same may be amended, supplemented or otherwise modified from time to time, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code.
- 1.44 <u>Disputed Claim</u> means any Claim, including any portion thereof, (a) that is neither an Allowed Claim nor a Disallowed Claim, or (b) to the extent the Debtors or any party in interest have interposed a timely objection or request for estimation of such Claim, which objection or request for estimation has not been withdrawn or determined pursuant to a Final Order.
- 1.45 <u>Distribution Record Date</u> means the Confirmation Date or such other date as may be designated in the Confirmation Order.
 - 1.46 <u>DS LLC</u> means STL Distribution Services LLC.
 - 1.47 <u>DTC</u> means The Depository Trust Company.
- 1.48 Effective Date means, and shall occur on, the Business Day on which (i) each of the conditions precedent to the occurrence of the Effective Date set forth in Article IX of this Plan has been satisfied or waived in accordance with the terms thereof and (ii) the Debtors file with the Bankruptcy Court a notice indicating the same and transmit a copy of such notice to (a) counsel to the Prepetition Administrative Agent and the Backstop Lenders and (b) counsel to the Required

Consenting Noteholders.

- 1.49 <u>Employee Benefit Plans</u> means any employment, tax qualified or non-tax qualified Pension Plan, Multiemployer Pension Plan, welfare, healthcare, bonus, incentive compensation, sick leave and other leave, vacation pay, business expense reimbursement, dependent care, retirement, savings, deferred compensation, supplemental pension, workers compensation, life insurance, disability, dependent care, dependent healthcare, education, severance or other compensation or benefit plans, agreements or arrangements for the benefit of the current or former directors, officers or employees (whether salaried or hourly, active or retired) of the applicable Debtor.
 - 1.50 Entity means an entity as defined in section 101(15) of the Bankruptcy Code.
- 1.51 <u>Estate(s)</u> means, individually or collectively, the estate or estates of the Debtors created in the Chapter 11 Cases under section 541 of the Bankruptcy Code.
 - 1.52 <u>Exhibit(s)</u> means, individually or collectively, the exhibits to this Plan.
- 1.53 Final Order means an order or judgment of the Bankruptcy Court (or other court of competent jurisdiction) entered by the Clerk of the Bankruptcy Court on the docket in the Chapter 11 Cases (or on the docket of any other court of competent jurisdiction), which has not been reversed, vacated or stayed and as to which (a) the time to appeal, petition for certiorari or move for a new trial, reargument or rehearing has expired and as to which no appeal, petition for certiorari or other proceedings for a new trial, reargument or rehearing shall then be pending, or (b) if an appeal, writ of certiorari, new trial, reargument or rehearing thereof has been sought, such order or judgment of the Bankruptcy Court shall have been affirmed by the highest court to which such order was appealed, or certiorari shall have been denied or a new trial, reargument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for certiorari or move for a new trial, reargument or rehearing shall have expired; provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed relating to such order, shall not cause such order not to be a Final Order.
- 1.54 <u>General Unsecured Claim</u> means a Claim against any Debtor that is not an Administrative Expense Claim, a DIP Revolving Facility Claim, a Priority Tax Claim, a Section 507(b) Claim, a Priority Non-Tax Claim, an Other Secured Claim, a Prepetition Credit Agreement Claim, a PD LLC Notes Claim, an Intercompany Claim, or the Herald Claim.
- 1.55 <u>Herald Claim</u> means the Claim of The Herald Publishing Company LLC against PD LLC and DS LLC arising under and pursuant to that certain Redemption Agreement dated as of February 18, 2009, which Claim is, <u>inter alia</u>, subordinated in all respects to the PD LLC Notes Claims.
 - 1.56 <u>Holder</u> means an Entity holding a Claim against, or Interest in, any Debtor.
 - 1.57 <u>Impaired</u> means "impaired" within the meaning of section 1124 of the Bankruptcy Code.
 - 1.58 <u>Initial Backstop Lender(s)</u> means, individually or collectively, (a) Goldman Sachs

Lending Partners LLC, (b) Mutual Quest Fund, (c) Monarch Master Funding Ltd, (d) Mudrick Distressed Opportunity Fund Global, LP, and (e) Blackwell Partners, LLC, each party to its own Backstop Commitment Letter.

- 1.59 <u>Intercompany Claim</u> means any Claim against a Debtor that is held by another Debtor or a non-Debtor Affiliate of any Debtor.
- 1.60 Interest means the interest of any Holder of equity securities in any Debtor that is represented by any issued and outstanding common stock, preferred stock, limited liability company interest, partnership interest, or any other instrument evidencing an ownership interest in such Debtor prior to the Effective Date (including prior to the Petition Date), whether or not transferable, and any restricted stock units, calls, rights, puts, awards, commitments, repurchase rights, unvested or unexercised options, rights of conversion, warrants, unvested common interests, unvested preferred interests or any other agreements of any character related to the common or preferred interests of any such Debtor, obligating any such Debtor to issue, transfer, purchase, redeem, or sell any equity interests or other equity securities, any rights under any equity incentive plans, voting agreements and registration rights agreements regarding equity securities of any such Debtor, any claims arising from the rescission of a purchase, sale or other acquisition of any outstanding common stock, preferred stock or other equity securities (or any right, claim, or interest in and to any common stock, preferred stock or other equity securities of such Debtor, any Claims for the payment of any distributions with respect to any common stock, preferred stock, or other equity interests in or securities of such Debtor, and any claims for damages or any other relief arising from the purchase, sale, or other acquisition of any such Debtor's outstanding common stock, preferred stock, or other equity interests in or securities of such Debtor, and any claims for damages or any other relief arising from the purchase, sale, or other acquisition of any such Debtor's outstanding common stock, preferred stock, or other equity interests or securities.
- 1.61 <u>Joining Lender Party</u> means any Person or Entity that (a) receives or acquires a portion of the Prepetition Credit Agreement Claims pursuant to a sale, assignment, transfer, hypothecation or other disposition of such Claims by a Consenting Lender and (b) executes a Lender Joinder, in each case in accordance with the terms of the Lee Support Agreement.
- 1.62 <u>Joining Noteholder Party</u> means any Person or Entity that (a) receives or acquires a portion of the PD LLC Notes Claims pursuant to a sale, assignment, transfer, hypothecation or other disposition of such Claims by a PD LLC Noteholder and (b) executes a Noteholder Joinder, in each case in accordance with the terms of the Pulitzer Support Agreement.
- 1.63 <u>Lee Closing Date Payment</u> means a payment of \$5,000,000 to be made by Reorganized PD LLC, utilizing funds made available to it by Reorganized Lee Enterprises, to the Holders of the New PD LLC Notes on the Effective Date, concurrently with the other transactions that will be authorized on the Effective Date pursuant to Article V of this Plan, which payment shall be deemed a prepayment of the New PD LLC Notes in the amount of \$5,000,000, in accordance with the terms of the New PD LLC Notes Agreement.
 - 1.64 <u>Lee Debtors</u> means, collectively, Lee Enterprises and the Lee Subsidiary Debtors.
 - 1.65 <u>Lee Enterprises</u> has the meaning set forth in the Introduction to this Plan.
 - 1.66 Lee Subsidiary Debtors means, collectively, Accudata, Inc.; INN Partners, L.C.;

Journal-Star Printing Co.; Klamath Falls Basin Publishing, Inc.; Lee Consolidated Holdings Co.; Lee Publications, Inc.; Lee Procurement Solutions Co.; and Sioux City Newspapers, Inc.

- 1.67 <u>Lee Support Agreement</u> means that certain Support Agreement, entered into by and among the Debtors and the Consenting Lenders, dated as of August 11, 2011 (including all exhibits and attachments thereto) and which became effective on September 8, 2011, as amended by the First Amendment to Support Agreement dated December 2, 2011 and as it may be further amended or otherwise modified from time to time in accordance with the terms thereof. A copy of the Lee Support Agreement is attached to the Disclosure Statement as Exhibit B.
- 1.68 <u>Lender Joinder</u> means a joinder to the Lee Support Agreement that complies with the requirements set forth in section 7.1(c) of the Lee Support Agreement and is executed by a Joining Lender Party and delivered to counsel to the Debtors and counsel to the Prepetition Administrative Agent.
- 1.69 <u>Lien</u> means, with respect to any interest in property, any mortgage, lien, pledge, charge, security interest, easement or encumbrance of any kind whatsoever affecting such interest in property.
- 1.70 <u>Maximum Exchange Amount</u> means the maximum amount of funded Prepetition Credit Agreement Claims that a particular Holder of Prepetition Credit Agreement Claims can convert into loans under the New Second Lien Term Loan, which shall be equal to \$166,250,000 multiplied by the amount of such Holder's funded Prepetition Credit Agreement Claims divided by the aggregate amount of all funded Prepetition Credit Agreement Claims, with such calculation determined as of (a) with respect to Consenting Lenders, the date such Consenting Lender executed its signature page to the Lee Support Agreement or Lender Joinder, as applicable, or (b) with respect to Holders of funded Prepetition Credit Agreement Claims that are not Consenting Lenders, the date such Holder executed and delivered its Ballot; <u>provided</u>, <u>however</u>, that the Maximum Exchange Amount shall not apply to the funded Prepetition Credit Agreement Claims converted by the Backstop Lenders into loans under the New Second Lien Term Loan.
- 1.71 <u>Multiemployer Pension Plan</u> means a plan (i) to which more than one employer is required to contribute, (ii) which is maintained pursuant to one or more collective bargaining agreements between one or more employee organizations and more than one employer, and (iii) which satisfies such other requirements contained in regulations promulgated by the United States Department of Labor.
- 1.72 New Common Stock means the approximately 6,743,640 shares of duly and validly issued new common stock in Reorganized Lee Enterprises, which will be issued (or shall be authorized to be issued) by Reorganized Lee Enterprises on, or as soon as reasonably practicable after, the Effective Date pursuant to this Plan and distributed to the Converting Lenders as set forth in Sections 3.2.3(c) and 5.3 of this Plan.
- 1.73 New First Lien Credit Agreement means that certain credit agreement, effective as of the Effective Date, by and among Reorganized Lee Enterprises, as borrower, DBTCA, as administrative agent and sole collateral agent, and the New First Lien Lenders, which shall be substantially in the form of Exhibit 1.73 to this Plan.

- 1.74 New First Lien Credit Facility means, collectively, the New First Lien Term Loan and the New First Lien Revolving Credit Facility.
- 1.75 <u>New First Lien Credit Facility Documents</u> means, collectively, the New First Lien Credit Agreement, the New Lee Intercreditor Agreement, and all other agreements (including, without limitation, any guaranty agreements, pledge and collateral agreements, and intercompany subordination agreements), documents and instruments delivered in connection therewith.
- 1.76 <u>New First Lien Lenders</u> means, as of the Effective Date, all Holders of Prepetition Credit Agreement Claims and DIP Revolving Facility Claims that receive loans under the New First Lien Term Loan or hold commitments or loans under the New First Lien Revolving Credit Facility and, after the Effective Date, their permitted assigns.
- 1.77 <u>New First Lien Revolving Credit Facility</u> means that certain superpriority revolving facility in an aggregate principal amount of \$40,000,000 provided under the New First Lien Credit Agreement.
- 1.78 New First Lien Term Loan means that certain term loan facility in an aggregate principal amount of up to \$689,510,000 provided under the New First Lien Credit Agreement.
- 1.79 New Lee Intercreditor Agreement means that certain intercreditor agreement, effective as of the Effective Date, by and among DBTCA, as collateral agent under the New First Lien Credit Facility Documents, Wilmington Trust, N.A. as collateral agent under the New Second Lien Term Loan Documents, and Reorganized Lee Enterprises, as the borrower thereunder, and each of the other loan parties party thereto, which shall be substantially in the form of Exhibit 1.79 to this Plan.
- 1.80 <u>New PD LLC Intercreditor Agreement</u> means that certain Intercreditor Agreement, effective as of the Effective Date, by and among Wilmington Trust, N.A. as collateral agent under the New Second Lien Term Loan Documents, certain of the Reorganized Debtors, and the PD LLC Notes Collateral Agent, which shall be substantially in the form of <u>Exhibit 1.80</u> to this Plan.
 - 1.81 New PD LLC Noteholder means the Holder of a New PD LLC Note.
- 1.82 New PD LLC Notes means the secured promissory notes that will be issued by Reorganized PD LLC on the Effective Date pursuant to the New PD LLC Notes Agreement and this Plan, in an aggregate principal amount equal to (i) the principal amount of the PD LLC Notes outstanding on the Effective Date minus the Lee Closing Date Payment plus (ii) a non-refundable fee in the amount of \$3,500,000, which shall be earned in full on the Effective Date and payable-in-kind (such that, for the avoidance of doubt, the aggregate principal amount of the New PD LLC Notes on the Effective Date shall be \$126,355,000), the terms of which are set forth in the New PD LLC Notes Agreement.
- 1.83 New PD LLC Notes Agreement means that certain amended and restated PD LLC Notes Agreement, effective as of the Effective Date, by and among PD LLC and the New PD LLC Noteholders party thereto on the Effective Date, pursuant to which the New PD LLC Notes will be issued. The New PD LLC Notes Agreement shall be substantially in the form of Exhibit 1.83 to this

Plan.

- 1.84 New PD LLC Notes Documents means collectively, the New PD LLC Notes, the New PD LLC Notes Agreement, the New PD LLC Intercreditor Agreement, the New PD LLC Notes Subsidiary Guaranty Agreement, the New Pulitzer Guaranty Agreement, and all agreements (including, without limitation, any pledge and collateral agreements, and intercompany subordination agreements), documents and instruments delivered in connection therewith.
- 1.85 New PD LLC Notes Subsidiary Guaranty Agreement means that certain Guaranty Agreement, effective as of the Effective Date, made by (i) the Pulitzer Subsidiary Debtors other than PD LLC and (ii) Star Publishing Company, an Arizona corporation, in favor of the holders of the New PD LLC Notes, which shall be in substantially the form of Exhibit 1.85 to this Plan.
- 1.86 New Pulitzer Guaranty Agreement means that certain Guaranty Agreement, effective as of the Effective Date, made by Pulitzer in favor of the holders of the New PD LLC Notes, which shall be in substantially the form of Exhibit 1.86 to this Plan.
- 1.87 <u>New Second Lien Lenders</u> means, as of the Effective Date, all Holders of Prepetition Credit Agreement Claims that receive loans under the New Second Lien Term Loan Agreement and, after the Effective Date, their permitted assigns.
- 1.88 New Second Lien Term Loan means that certain second lien term loan in an aggregate principal amount of \$175,000,000 provided under the New Second Lien Term Loan Agreement.
- 1.89 New Second Lien Term Loan Agreement means that certain second lien term loan agreement, effective as of the Effective Date, by and among Reorganized Lee Enterprises, as borrower, Wilmington Trust, N.A., as administrative agent and collateral agent, and the New Second Lien Lenders. The New Second Lien Term Loan Agreement shall be substantially in the form of Exhibit 1.89 to this Plan, with such modifications, if any, that are reasonably satisfactory to the Prepetition Administrative Agent and the Initial Backstop Lenders.
- 1.90 New Second Lien Term Loan Documents means, collectively, the New Second Lien Term Loan Agreement, the New Lee Intercreditor Agreement, the New PD LLC Intercreditor Agreement, and all other agreements (including, without limitation, any guaranty agreements and pledge and collateral agreements), documents and instruments delivered in connection therewith, in each case with such modifications, if any, that are reasonably satisfactory to the Prepetition Administrative Agent and the Initial Backstop Lenders.
- 1.91 New Second Lien Term Loan Election means an irrevocable election made by certain Holders of Allowed Prepetition Credit Agreement Claims (or an advisor, nominee, or investment manager for beneficial holder(s)) to convert all or a portion of such Holder's funded Prepetition Credit Agreement Claims into a pro rata share of the New Second Lien Term Loan. Each such Holder's converted amount shall be (a) as set forth on Annex II to the Lee Support Agreement or, in the case of a Backstop Lender, as set forth on Annex II to the Lee Support Agreement to such Backstop Lender's commitments under the relevant Backstop Commitment Letter, (b) as set forth on the annex to the Lender Joinder executed by such

Holder if such Holder is a Joining Lender Party, or (c) to the extent such Holder has not executed the Lee Support Agreement or a Lender Joinder, as set forth on the Ballot completed by such Holder; <u>provided</u>, <u>however</u>, that for each Holder of an Allowed Prepetition Credit Agreement Claim that executed and delivered the Lee Support Agreement and made the New Second Lien Term Loan Election, the converted amount shall also appear on such Holder's Ballot; <u>provided further</u> that, with respect to any Holder other than a Backstop Lender, in no event shall such Holder's New Second Lien Term Election exceed such Holder's Maximum Exchange Amount; and <u>provided further</u> that each Backstop Lender shall be obligated (on the terms and subject to the conditions set forth in the applicable Backstop Commitment Letter) to convert into the New Second Lien Term Loan the amount of such Backstop Lender's funded Prepetition Credit Agreement Claims that is necessary to satisfy such Backstop Lender's commitments under the applicable Backstop Commitment Letter.

- 1.92 <u>Noteholder Joinder</u> means a joinder to the Pulitzer Support Agreement that complies with the requirements set forth in section 7.1(c) of the Pulitzer Support Agreement and is executed by a Joining Noteholder Party and delivered to counsel to the Debtors.
- 1.93 <u>Other Secured Claim</u> means any Secured Claim other than a DIP Revolving Facility Claim, a Prepetition Credit Agreement Claim or a PD LLC Notes Claim.
 - 1.94 PD LLC means St. Louis Post-Dispatch LLC.
 - 1.95 PD LLC Noteholder means the Holder of a PD LLC Note.
- 1.96 <u>PD LLC Noteholders' Fees</u> means the reasonable fees, expenses, disbursements and indemnity claims, including, without limitation, attorney and agent fees, expenses and disbursements, incurred by the PD LLC Noteholders, whether prior to or after the Petition Date, and until the Effective Date, as required to be paid by PD LLC pursuant to the PD LLC Notes Agreement.
- 1.97 <u>PD LLC Notes</u> means the notes issued and outstanding under the PD LLC Notes Documents, which have been issued Private Placement Number 85229*AB2 by Standard & Poor's CUSIP Service Bureau.
- 1.98 PD LLC Notes Agreement means the PD LLC Note Agreement, dated as of May 1, 2000, as amended by (i) Amendment No. 1 to Note Agreement dated as of November 23, 2004, (ii) Amendment No. 2 to Note Agreement dated as of February 1, 2006, (iii) Amendment No. 3 to Note Agreement dated as of November 19, 2008, (iv) the Limited Waiver to Note Agreement and Guaranty Agreement (as amended), dated as of December 26, 2008, (v) Amendment No. 4 and First Amendment to Limited Waiver to Note Agreement and Guaranty Agreement, dated as of January 16, 2009, (vi) Limited Waiver and Amendment No. 5 to Note Agreement, dated as of February 18, 2009, (vii) Amendment No. 6 to Note Agreement dated as of April 6, 2011; and (viii) and Limited Amendment No. 7 to Note Agreement dated as of November 7, 2011, as such agreement may be further amended, modified, or supplemented from time to time.
- 1.99 <u>PD LLC Notes Claims</u> means all Claims against the Debtors arising under, evidenced by, or secured pursuant to, the PD LLC Notes Documents.
 - 1.100 PD LLC Notes Collateral Agent means The Bank of New York Mellon Trust

Company, N.A., in its capacity as collateral agent under the PD LLC Notes Security Agreement.

- 1.101 <u>PD LLC Notes Documents</u> means, collectively, the PD LLC Notes Agreement and all other agreements (including, without limitation, any guaranty agreements, pledge agreements, and collateral agreements), documents, and instruments delivered in connection therewith.
- 1.102 PD LLC Notes Guaranty Agreements means, collectively, (i) that certain Guaranty Agreement, dated as of May 1, 2000, as amended by (a) Amendment No. 1 to Guaranty Agreement, dated as of August 7, 2000, (b) Amendment No. 2 to Guaranty Agreement, dated as of November 23, 2004, (c) Amendment No. 3 to Guaranty Agreement, dated as of June 3, 2005, (d) Amendment No. 4 to Guaranty Agreement, dated as of February 1, 2006, and (e) Limited Waiver and Amendment No. 5 to Guaranty Agreement, dated as of February 18, 2009, in each case between Pulitzer and the PD LLC Noteholders then party to the PD LLC Notes Agreement and (ii) that certain Subsidiary Guaranty Agreement, dated as of February 18, 2009, between the Pulitzer Debtors other than Pulitzer, on the one hand, and the PD LLC Noteholders then party to the PD LLC Notes Agreement, on the other hand.
- 1.103 <u>PD LLC Notes Security Agreement</u> means that certain Security Agreement dated as of February 18, 2009 (as amended, restated, supplemented, or otherwise modified) by Pulitzer and certain of its subsidiaries in favor of the PD LLC Notes Collateral Agent, on behalf of and for the benefit of the PD LLC Noteholders.
- 1.104 <u>Pension Plan</u> means an employee pension benefit plan within the meaning of section (2)(A) of the Employee Retirement Income Security Act of 1974, as amended, but excludes Multiemployer Pension Plans.
- 1.105 <u>Person</u> means any individual, corporation, partnership, association, joint stock company, joint venture, limited liability company, limited liability partnership, trust, estate, unincorporated organization or other entity, or any domestic or foreign government, governmental agency, or any subdivision, department or other instrumentality thereof.
 - 1.106 Petition Date means the date on which the Debtors commenced the Chapter 11 Cases.
- 1.107 <u>Plan</u> means this Chapter 11 plan of reorganization, including all Exhibits, supplements, appendices and schedules hereto, either in its present form or as the same may be altered, amended or modified from time to time in accordance with the provisions of the Bankruptcy Code, the terms hereof, the Lee Support Agreement, and the Pulitzer Support Agreement.
- 1.108 <u>Prepetition Administrative Agent</u> means DBTCA, in its capacity as administrative agent under the Prepetition Credit Agreement.
- 1.109 <u>Prepetition Credit Agreement</u> means that certain Amended and Restated Credit Agreement, dated as of December 21, 2005, by and among Lee Enterprises, the lenders from time to time party thereto, DBTCA, as Administrative Agent, DBSI and SunTrust Capital Markets, Inc., as Joint Lead Arrangers, DBSI, as Book Running Manager, SunTrust Bank, as Syndication Agent, and Bank of America, N.A., The Bank of New York and The Bank of Tokyo-Mitsubishi, Ltd., Chicago Branch, as Co-Documentation Agents, as amended by that First Amendment dated as of September

- 29, 2008, that Second Amendment dated as of October 29, 2008, that Third Amendment dated as of February 18, 2009, and that Fourth Amendment dated as of December 2, 2011, as such agreement may be further amended, modified, or supplemented from time to time.
- 1.110 <u>Prepetition Credit Agreement Claims</u> means all Claims arising under, evidenced by, or secured pursuant to, the Prepetition Credit Agreement Documents.
- 1.111 <u>Prepetition Credit Agreement Documents</u> means, collectively, the Prepetition Credit Agreement and all other agreements (including, without limitation, any guaranty agreements, pledge and collateral agreements, intercreditor agreements, and intercompany subordination agreements), documents and instruments delivered in connection therewith.
- 1.112 <u>Priority Non-Tax Claim</u> means any Claim entitled to priority in payment pursuant to section 507(a) of the Bankruptcy Code, other than an Administrative Expense Claim or a Priority Tax Claim.
- 1.113 <u>Priority Tax Claim</u> means any Claim of a governmental unit of the kind entitled to priority in payment as specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.114 <u>Professional</u> means any Person retained by the Debtors or a statutory committee, if any, pursuant to a Final Order of the Bankruptcy Court entered pursuant to sections 327, 328 or 1103 of the Bankruptcy Code.
- 1.115 <u>Pro Rata</u> or <u>Pro Rata Share</u> means the proportion that the amount of any Allowed Claim in a particular Class bears to the aggregate amount of all Allowed Claims in such Class, except in reference to a specific type of Claim, in which case Pro Rata or Pro Rata Share means the proportion that an Allowed Claim of such type bears to the aggregate amount of all Allowed Claims of such type.
 - 1.116 <u>Pulitzer</u> means Pulitzer Inc.
 - 1.117 <u>Pulitzer Debtors</u> means, collectively, Pulitzer and the Pulitzer Subsidiary Debtors.
- 1.118 Pulitzer Subsidiary Debtors means, collectively, Pulitzer Technologies, Inc.; PD LLC; Fairgrove LLC; DS LLC; Suburban Journals of Greater St. Louis LLC; Pulitzer Network Systems LLC; Pulitzer Newspapers, Inc.; Flagstaff Publishing Co.; Hanford Sentinel Inc.; HomeChoice, LLC; Kauai Publishing Co.; Napa Valley Publishing Co.; NIPC, Inc. f/k/a Northern Illinois Publishing Co., Inc.; NVPC LLC; Northern Lakes Publishing Co.; NLPC LLC; Pantagraph Publishing Co.; HSTAR LLC; Pulitzer Missouri Newspapers, Inc.; Pulitzer Utah Newspapers, Inc.; Santa Maria Times, Inc.; SHTP LLC; Southwestern Oregon Publishing Co.; SOPC LLC; and Ynez Corporation.
- 1.119 <u>Pulitzer Support Agreement</u> means that certain Support Agreement, entered into by and among PD LLC, Star Publishing Company, the Guarantors (as defined in the Pulitzer Support Agreement), the Lee Debtors, and the Consenting Noteholders, including all exhibits and attachments thereto, which became effective on December 2, 2011, as amended or otherwise modified from time to time in accordance with the terms thereof.

- 1.120 Reinstate, Reinstated or Reinstatement means (a) leaving unaltered the legal, equitable and contractual rights to which a Claim or Interest entitles the Holder of such Claim or Interest, or (b) notwithstanding any contractual provision or applicable law that entitles the Holder of such Claim or Interest to demand or receive accelerated payment of such Claim or Interest after the occurrence of a default, (i) curing any such default that occurred before or after the Petition Date, other than a default of a kind specified in section 365(b)(2) of the Bankruptcy Code; (ii) reinstating the maturity of such Claim or Interest as such maturity existed before such default; (iii) compensating the Holder of such Claim or Interest for any damages incurred as a result of any reasonable reliance by such Holder on such contractual provision or such applicable law; (iv) if such Claim or Interest arises from any failure to perform a nonmonetary obligation other than a default arising from failure to operate under a nonresidential real property lease subject to section 365(b)(1)(A) of the Bankruptcy Code, compensating the Holder of such Claim or Interest (other than any Debtor or an insider of any Debtor) for any pecuniary loss incurred by such Holder as the result of such failure; and (v) not otherwise altering the legal, equitable or contractual rights to which such Claim or Interest entitles the Holder thereof.
- 1.121 <u>Related Persons</u> means, with respect to any Person, such Person's predecessors, successors and assigns (whether by operation of law or otherwise) and their respective present and former affiliates and each of their respective current and former members, partners, equity holders, officers, directors, employees, managers, shareholders, partners, financial advisors, attorneys, accountants, investment bankers, consultants, agents and professionals each acting in such capacity, and any Person claiming by or through any of them (including their respective officers, directors, managers, shareholders, partners, employees, equity holders, members, and professionals).
- 1.122 <u>Released Parties</u> means (i) the Debtors, (ii) the Consenting Lenders, (iii) the Backstop Lenders, (iv) the Prepetition Administrative Agent, (v) the Consenting Noteholders, (vi) the PD LLC Notes Collateral Agent, (vii) the DIP Revolving Facility Agent, (viii) the DIP Revolving Facility Lenders, and (ix) the Related Persons of each of the foregoing.
- 1.123 Reorganized means, individually or collectively, each of the reorganized Debtors and any successors thereto by merger, consolidation, conversion or otherwise, on or after the Effective Date, after giving effect to the transactions occurring on or prior to the Effective Date in accordance with this Plan. For example, (i) Reorganized Lee Enterprises means reorganized Lee Enterprises or any successors thereto by merger, consolidation, conversion or otherwise, on or after the Effective Date, after giving effect to the transactions occurring on the Effective Date in accordance with this Plan and (ii) Reorganized Debtors means, collectively, each of the reorganized Debtors or any successors thereto by merger, consolidation, conversion or otherwise, on or after the Effective Date, after giving effect to the transactions occurring on or prior to the Effective Date in accordance with this Plan.
- 1.124 <u>Required Consenting Lenders</u> means, collectively, the Consenting Lenders holding greater than 50.0% of the aggregate outstanding amount of the funded Prepetition Credit Agreement Claims held by all of the Consenting Lenders.
- 1.125 <u>Required Consenting Noteholders</u> means (a) at any time that there are more than two non-affiliated Consenting Noteholders (which, for the avoidance of doubt, shall not include Pulitzer or any of its affiliates), the Consenting Noteholders holding greater than 60% of the aggregate

outstanding PD LLC Notes held by all Consenting Noteholders, so long as at least two Consenting Noteholders are not affiliates, and (b) otherwise, the Consenting Noteholders holding 51% of the aggregate outstanding PD LLC Notes held by the Consenting Noteholders.

- 1.126 Second Lien Pro Rata Share means (i) with respect to each Converting Lender (other than a Backstop Lender), the percentage obtained by dividing (x) the amount of New Second Lien Term Loan that such Converting Lender validly elected to receive pursuant to its New Second Lien Term Loan Election by (y) the aggregate amount of the New Second Lien Term Loan (without giving effect to any original issue discount for purpose of clauses (x) and (y)) and (ii) with respect to each Backstop Lender, the percentage obtained by dividing (x) the aggregate amount of New Second Lien Term Loan that such Backstop Lender receives in accordance with its Backstop Commitment Letter by (y) the aggregate amount of the New Second Lien Term Loan (without giving effect to any original issue discount for purpose of clauses (x) and (y)).
- 1.127 Section 507(b) Claims means any superpriority Claims pursuant to section 507(b) of the Bankruptcy Code that may be granted by the Bankruptcy Court to (i) the Prepetition Administrative Agent and the Holders of Prepetition Credit Agreement Claims under the Bankruptcy Court's interim and/or final orders approving the Lee Debtors' entry into and performance of their obligations under the DIP Revolving Facility Documents and (ii) the PD LLC Notes Collateral Agent and the Holders of PD LLC Notes Claims under the Bankruptcy Court's interim and/or final orders approving the Pulitzer Debtors' use of cash collateral.
- 1.128 Secured Claim means any Claim secured by a Lien on collateral to the extent of the value of such collateral (i) as set forth in this Plan, (ii) as agreed to by the Holder of such Claim and the relevant Debtor(s) or (iii) as determined pursuant to a Final Order of the Bankruptcy Court in accordance with section 506(a) of the Bankruptcy Code or, in the event that such Claim is subject to setoff under section 553 of the Bankruptcy Code, to the extent of such setoff.
 - 1.129 <u>Unimpaired</u> means with respect to a Claim or Interest, a Claim or Interest that is not Impaired.
 - 1.130 <u>Voting Deadline</u> means the deadline for returning Ballots to accept or reject this Plan.
- B. Rules of Interpretation. For purposes of this Plan, unless otherwise provided herein: (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural; (b) unless otherwise provided in this Plan, any reference in this Plan to a contract, instrument, release, or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions; (c) any reference in this Plan to an existing document, schedule or Exhibit filed or to be filed means such document, schedule or Exhibit, as it may have been or may be amended, modified, or supplemented pursuant to this Plan; (d) any reference to an entity as a Holder of a Claim or Interest includes that entity's successors and assigns; (e) all references in this Plan to Sections or Articles are references to Sections or Articles of or to this Plan, as the same may be amended, waived or modified from time to time; (f) the words "herein," "hereof," "hereto," "hereto," "hereunder" and other words of similar import refer to this Plan as a whole and not to any particular Section, subsection or clause contained in this Plan; (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended

to be a part of or to affect the interpretation of this Plan; (h) the rules of construction set forth in section 102 of the Bankruptcy Code will apply; and (i) in computing any period of time prescribed or allowed by this Plan, Bankruptcy Rule 9006(a) will apply.

- C. <u>Exhibits</u>. All Exhibits to this Plan are incorporated into and are a part of this Plan as if set forth in full herein. Exhibits to this Plan that are not annexed hereto shall be filed with the Clerk of the Bankruptcy Court no later than ten (10) days prior to the deadline established for objecting to Confirmation of this Plan. Holders of Claims and Interests may obtain a copy of the filed Exhibits upon written request to the Debtors. Upon their filing, the Exhibits may be inspected in the office of the Clerk of the Bankruptcy Court during normal business hours or at the Bankruptcy Court's website at http://www.deb.uscourts.gov.
- D. <u>Deemed Acts</u>. Whenever an act or event is expressed under this Plan to have been deemed done or to have occurred, it shall be deemed to have been done or to have occurred by virtue of this Plan and/or Confirmation Order without any further act by any party.

ARTICLE II:

TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS, DIP REVOLVING FACILITY CLAIMS, PRIORITY TAX CLAIMS, AND SECTION 507(b) CLAIMS

In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Expense Claims, DIP Revolving Facility Claims, Priority Tax Claims, and Section 507(b) Claims have not been classified and thus are excluded from the Classes of Claims and Interests set forth in Article III.

2.1. <u>Administrative Expense Claims</u>. Subject to the provisions of sections 328, 330, 331 and 503(b) of the Bankruptcy Code, on either: (i) the latest to occur of (x) the Effective Date, (y) the date upon which such Administrative Expense Claim becomes an Allowed Claim and (z) such other date as agreed upon by the Debtors and the Holder of such Administrative Expense Claim, or (ii) such other date as the Bankruptcy Court may order, each Holder of an Allowed Administrative Expense Claim shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Administrative Expense Claim, (a) Cash equal to the full unpaid amount of such Allowed Administrative Expense Claim, or (b) such other less favorable treatment as the applicable Debtor and such Holder shall have agreed; <u>provided</u>, <u>however</u>, that Allowed Administrative Expense Claims not yet due or that represent obligations incurred by the Debtors in the ordinary course of their business during the Chapter 11 Cases, or assumed by the Debtors during the Chapter 11 Cases, shall be paid or performed when due in the ordinary course of business and in accordance with the terms and conditions of the particular agreements governing such obligations.

- 2.2. <u>DIP Revolving Facility Claims</u>. The DIP Revolving Facility Claims shall be Allowed on the Effective Date pursuant to this Plan. On the Effective Date, all Allowed DIP Revolving Facility Claims shall, at the Debtors' option, in accordance with and on the terms set forth in the DIP Revolving Facility Agreement, either (i) convert into revolving loans and commitments outstanding under the New First Lien Revolving Credit Facility pursuant to the New First Lien Credit Facility Documents and the DIP Revolving Facility Commitment Letter; <u>provided</u>, <u>however</u>, that all accrued and unpaid interest and all fees and expenses due and payable on or prior to the Effective Date under the terms of the DIP Revolving Facility Agreement shall be indefeasibly paid in full in Cash on the Effective Date by the Lee Debtors, or (ii) be indefeasibly paid in full in Cash by the Lee Debtors, and, in each case, the "Commitments" (as defined in the DIP Revolving Facility Agreement) under the DIP Revolving Facility Agreement shall be irrevocably cancelled and the "Letters of Credit" (as defined in the DIP Revolving Facility Agreement) shall be returned undrawn. Notwithstanding anything to the contrary in this Plan, the liens and security interests securing the DIP Revolving Facility Claims shall continue in full force and effect until the DIP Revolving Facility Claims (a) convert into revolving loans and commitments outstanding under the New First Lien Revolving Credit Facility or (b) are indefeasibly paid in full in Cash by the Lee Debtors, the "Commitments" (as defined in the DIP Revolving Facility Agreement) are irrevocably cancelled, and the "Letters of Credit" (as defined in the DIP Revolving Facility Agreement) are irrevocably cancelled, and the "Letters of Credit" (as defined in the DIP Revolving Facility Agreement) are returned undrawn on the Effective Date.
- 2.3. Priority Tax Claims. Except to the extent that a Holder of an Allowed Priority Tax Claim agrees to a less favorable treatment (in which event such other agreement shall govern), each Holder of an Allowed Priority Tax Claim against any of the Debtors that is due and payable on or before the Effective Date shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Priority Tax Claim, Cash equal to the amount of such Allowed Priority Tax Claim on the later of (i) the Effective Date and (ii) the date such Priority Tax Claim becomes an Allowed Claim or as soon thereafter as practicable. All Allowed Priority Tax Claims against any of the Debtors which are not due and payable on the Effective Date shall be paid in the ordinary course of business by the Reorganized Debtors in accordance with the terms thereof.
- 2.4. Section 507(b) Claims. On the Effective Date, as a consequence of the implementation of the New First Lien Credit Facility, the Reorganized Debtors' entry into the New Second Lien Term Loan Documents, the issuance of the New Common Stock, and the treatment of PD LLC Notes Claims in accordance with section 3.3.3(c) of this Plan, all Section 507(b) Claims shall be settled, released by the Holders of such Claims, and discharged.

ARTICLE III: CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS

3.1. <u>Summary of Classification and Treatment of Classified Claims and Interests</u>.

3.1.1 <u>General</u>.

- (a) Pursuant to sections 1122 and 1123 of the Bankruptcy Code, Claims and Interests are classified for all purposes, including, without limitation, voting, Confirmation and distribution pursuant to this Plan, as set forth herein. A Claim or Interest shall be deemed classified in a particular Class only to the extent that the Claim or Interest qualifies within the description of that Class, and shall be deemed classified in a different Class to the extent that any remainder of such Claim or Interest qualifies within the description of such different Class. A Claim or Interest is in a particular Class only to the extent that such Claim or Interest is Allowed in that Class and has not been paid or otherwise settled prior to the Effective Date.
- (b) This Plan constitutes, and shall be deemed to constitute, a separate chapter 11 plan of reorganization for each Debtor. For purposes of brevity and convenience, the classification and treatment of Claims and Interests has been set forth in two (2) groups: the Lee Debtors (Debtors 1-9) and (ii) the Pulitzer Debtors (Debtors 10-35). Each group is specifically listed in Appendices A and B to this Plan, and each Debtor is ascribed a numerical designation therein that corresponds to the applicable Debtor.
- 3.1.2 <u>Identification of Classes Against the Lee Debtors</u>. The following chart assigns a letter to each Class with respect to the Lee Debtors (Debtors 1-9) for purposes of identifying each separate Class:

<u>CLASS</u>	CLAIM OR INTEREST
A	Priority Non-Tax Claims
В	Other Secured Claims
С	Prepetition Credit Agreement Claims
D	General Unsecured Claims
E	Intercompany Claims
G	Interests in the Lee Debtors

3.1.3 <u>Identification of Classes Against the Pulitzer Debtors</u>. The following chart assigns a letter to each Class with respect to the applicable Pulitzer Debtors (Debtors 10-35) for purposes of identifying each separate Class:

CLASS	<u>CLAIM OR INTEREST</u>
A	Priority Non-Tax Claims
В	Other Secured Claims
C	PD LLC Notes Claims
D	General Unsecured Claims
E	Intercompany Claims
F	Herald Claim (against PD LLC and DS LLC only)
G	Interests in the Pulitzer Debtors

3.2. <u>Classification and Treatment of Claims Against and Interests in the Lee Debtors (Debtors 1-9).</u>

3.2.1 <u>Classes 1A through 9A: Priority Non-Tax Claims</u>.

- (a) Classification: Classes 1A through 9A consist of all Priority Non-Tax Claims against the Lee Debtors.
- (b) Treatment: On the Effective Date, each Holder of an Allowed Priority Non-Tax Claim against any of the Lee Debtors shall have such Claim Reinstated.
- (c) Voting: Claims in Classes 1A through 9A are Unimpaired. Each Holder of an Allowed Claim in Classes 1A through 9A shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.2 Classes 1B through 9B: Other Secured Claims.

- (a) Classification: Classes 1B through 9B consist of all Other Secured Claims against the Lee Debtors.
- (b) Treatment: On the Effective Date, each Holder of an Allowed Other Secured Claim against any of the Lee Debtors shall have its Claim Reinstated.
- (c) Voting: Claims in Classes 1B through 9B are Unimpaired. Each Holder of an Allowed Claim in Classes 1B through 9B shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.3 Classes 1C through 9C: Prepetition Credit Agreement Claims.

- (a) Classification: Classes 1C through 9C consist of all Prepetition Credit Agreement Claims against the Lee Debtors.
- (b) Allowance: The Prepetition Credit Agreement Claims shall be Allowed on the Effective Date pursuant to this Plan in the aggregate principal amount thereof, plus

- (i) any accrued but unpaid interest thereon payable at the non-default interest rate under the Prepetition Credit Agreement through the Effective Date of this Plan and (ii) all other unpaid monetary Obligations as defined in the Prepetition Credit Agreement Documents, except to the extent that such Claims are otherwise provided in this Plan or any of the documents entered into in accordance with this Plan to be paid or satisfied, and shall not be subject to avoidance, objection, challenge, deduction, subordination, recharacterization or offset.
- (c) Treatment: On the Effective Date, each Holder of an Allowed Prepetition Credit Agreement Claim against the Lee Debtors shall receive on account of, in full and complete satisfaction, settlement, release and discharge of, and in exchange for such Claim, such Holder's Pro Rata Share of (i) the New First Lien Term Loan and (ii) any accrued and unpaid interest on the Prepetition Credit Agreement Claims, calculated at the non-default rate under the Prepetition Credit Agreement Documents, through the Effective Date, that has not been previously paid to the Holders of Prepetition Credit Agreement Claims, including, but not limited to, as adequate protection under applicable orders of the Bankruptcy Court; provided, however, that each Converting Lender shall receive, for the Prepetition Credit Agreement Claim(s) that such Converting Lender elects to convert pursuant to such Converting Lender's New Second Lien Term Loan Election (which converted amount shall be deducted both from the amount of such Converting Lender's Allowed Prepetition Credit Agreement Claims solely for purposes of determining the Holders' Pro Rata Shares of the New First Lien Term Loan), such Converting Lender's Second Lien Pro Rata Share of (i) the New Second Lien Term Loan and (ii) the New Common Stock. In addition, on the Effective Date, any unexpired letters of credit outstanding under the Prepetition Credit Agreement shall be deemed made or issued, as applicable, under the New First Lien Revolving Credit Facility.
- (d) Voting: Claims in Classes 1C through 9C are Impaired. Each Holder of an Allowed Claim in Classes 1C through 9C shall be entitled to vote to accept or reject this Plan.

3.2.4 Classes 1D through 9D: General Unsecured Claims.

- (a) Classification: Classes 1D through 9D consist of all General Unsecured Claims against the Lee Debtors.
- (b) Treatment: On or as soon as reasonably practicable after the latest to occur of (i) the Effective Date or (ii) the date on which an Allowed General Unsecured Claim against any of the Lee Debtors becomes due and payable in accordance with the applicable non-bankruptcy law governing such Claim, the Holder of such Allowed General Unsecured Claim against any of the Lee Debtors shall receive Cash from the Lee Debtors equal to the full unpaid amount of such Claim, together with interest thereon, at the applicable non-default rate, to which the Holder of such Claim may be entitled under applicable non-bankruptcy law through the Effective Date.
- (c) Voting: Claims in Classes 1D through 9D are Unimpaired. Each Holder of an Allowed Claim in Classes 1D through 9D shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

- 3.2.5 <u>Classes 1E through 9E: Intercompany Claims</u>.
 - (a) Classification: Classes 1E through 9E consist of all Intercompany Claims against the Lee Debtors.
- (b) Treatment: On the Effective Date each Holder of an Allowed Intercompany Claim against any of the Lee Debtors shall have such Claim Reinstated; <u>provided</u>, <u>however</u>, that the Intercompany Claim held by Pulitzer against Lee Enterprises shall be reduced by the amount of the Lee Closing Date Payment.
- (c) Voting: Claims in Classes 1E through 9E are Unimpaired. Each Holder of an Allowed Claim in Classes 1E through 9E shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.
 - 3.2.6 <u>Classes 1G through 9G: Interests in the Lee Debtors.</u>
 - (a) Classification: Classes 1G through 9G consist of all Interests in the Lee Debtors.
- (b) Treatment: On the Effective Date, each Holder of an Interest in the Lee Debtors shall have such Interest Reinstated.
- (c) Voting: Interests in Classes 1G through 9G are Unimpaired. Each Holder of an Interest in Classes 1G through 9G shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.
 - 3.3. Classification and Treatment of Claims Against and Interests in the Pulitzer Debtors (Debtors 10-35).
 - 3.3.1 Classes 10A through 35A: Priority Non-Tax Claims.
- (a) Classification: Classes 10A through 35A consist of all Priority Non-Tax Claims against the Pulitzer Debtors.
- (b) Treatment: On the Effective Date, each Holder of an Allowed Priority Non-Tax Claim against any of the Pulitzer Debtors shall have such Claim Reinstated.
- (c) Voting: Claims in Classes 10A through 35A are Unimpaired. Each Holder of an Allowed Claim in Classes 10A through 35A shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.3.2 Classes 10B through 35B: Other Secured Claims.

Debtors.

- (a) Classification: Classes 10B through 35B consist of all Other Secured Claims against the Pulitzer
- (b) Treatment: On the Effective Date, each Holder of an Allowed Other Secured Claim against any of the Pulitzer Debtors shall have its Claim Reinstated.
- (c) Voting: Claims in Classes 10B through 35B are Unimpaired. Each Holder of an Allowed Claim in Classes 10B through 35B shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.3.3 Classes 10C through 35C: PD LLC Notes Claims.

- (a) Classification: Classes 10C through 35C consist of all PD LLC Notes Claims against the Pulitzer Debtors.
- (b) Allowance: The PD LLC Notes Claims shall be Allowed pursuant to this Plan in the aggregate principal amount thereof as of the Effective Date, plus (i) any accrued but unpaid interest thereon at the non-default rate under the PD LLC Notes Documents through the Effective Date of this Plan and (ii) all other Obligations (as defined in the PD LLC Notes Documents), except to the extent that such Claims are otherwise provided in this Plan or any of the documents entered into in accordance with this Plan to be paid or satisfied, and shall not be subject to avoidance, objection, challenge, deduction, subordination, recharacterization, or offset.
- (c) Treatment: On the Effective Date, each Holder of an Allowed PD LLC Notes Claim shall receive on account of, in full and complete satisfaction, settlement, release and discharge of, and in exchange for such Claim, such Holder's Pro Rata Share of (i) the New PD LLC Notes, (ii) the Lee Closing Date Payment, which shall be deemed to be a prepayment of the New PD LLC Notes held by such Holder, and (iii) any accrued interest on the PD LLC Notes, calculated at the non-default rate under the PD LLC Notes Documents, through the Effective Date, that has not previously been paid to the PD LLC Noteholders, including, but not limited to, as adequate protection under applicable orders of the Bankruptcy Court.
- (d) Voting: Claims in Classes 10C through 35C are Impaired. Each Holder of an Allowed Claim in Classes 10C through 35C shall be entitled to vote to accept or reject this Plan.

3.3.4 <u>Classes 10D through 35D: General Unsecured Claims</u>.

- (a) Classification: Classes 10D through 35D consist of all General Unsecured Claims against the Pulitzer Debtors.
- (b) Treatment: On or as soon as reasonably practicable after the latest to occur of (a) the Effective Date or (b) the date on which an Allowed General Unsecured Claim against any of the Pulitzer Debtors becomes due and payable in accordance with the applicable non-

bankruptcy law governing such Claim, the Holder of such Allowed General Unsecured Claim against any of the Pulitzer Debtors shall receive Cash from the Pulitzer Debtors equal to the full unpaid amount of such Claim, together with interest thereon, at the applicable non-default rate, to which the Holder of such Claim may be entitled under applicable non-bankruptcy law through the Effective Date.

(c) Voting: Claims in Classes 10D through 35D are Unimpaired. Each Holder of an Allowed Claim in Classes 10D through 35D shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.3.5 Classes 10E through 35E: Intercompany Claims.

- (a) Classification: Classes 10E through 35E consist of all Intercompany Claims against the Pulitzer Debtors.
- (b) Treatment: On the Effective Date each Holder of an Allowed Intercompany Claim against any of the Pulitzer Debtors shall have such Claim Reinstated; <u>provided</u>, <u>however</u>, that any Intercompany Claims owed by any of the Pulitzer Debtors to any others of the Pulitzer Debtors shall be capable of being cancelled notwithstanding such Reinstatement with the agreement of the Pulitzer Debtors in question.
- (c) Voting: Claims in Classes 10E through 35E are Unimpaired. Each Holder of an Allowed Claim in Classes 10E through 35E shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.3.6 Classes 12F and 14F: Herald Claim against PD LLC and DS LLC.

- (a) Classification: Classes 12F and 14F consist of the Herald Claim against PD LLC and DS LLC.
- (b) Treatment: On the Effective Date, the Herald Claim shall be Reinstated, and shall, in accordance with the treatment afforded to such Claim under that certain Redemption Agreement dated February 18, 2009, (i) not be redeemed for Cash unless all obligations under the New First Lien Term Loan, the New Second Lien Term Loan, and the New First Lien Revolving Credit Facility have been paid in full in Cash and all obligations of PD LLC and DS LLC under the New PD LLC Notes Documents have been paid in full in Cash, and (ii) be junior and subordinate in all respects to the obligations of Reorganized PD LLC and Reorganized DS LLC under the New PD LLC Notes Documents in the same manner as the Herald Claim was junior and subordinate in all respects to the obligations of PD LLC and DS LLC under the PD LLC Notes Documents.
- (c) Voting: The Herald Claim in Classes 12F and 14F is Unimpaired. The Holder of the Herald Claim in Classes 12F and 14F shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

- 3.3.7 <u>Classes 10G through 35G: Interests in the Pulitzer Debtors.</u>
 - (a) Classification: Classes 10G through 35G consist of all Interests in the Pulitzer Debtors.
- (b) Treatment: On the Effective Date, each Holder of an Interest in the Pulitzer Debtors shall have such Interest Reinstated.
- (c) Voting: Interests in Classes 10G through 35G are Unimpaired. Each Holder of an Interest in Classes 10G through 35G shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.
- 3.4. <u>Unimpaired Claims and Interests</u>. Except as otherwise explicitly provided in this Plan, nothing herein shall affect the Debtors' or the Reorganized Debtors' rights and defenses, both legal and equitable, with respect to any Unimpaired Claims or Interests, including, but not limited to, the legal and equitable defenses of setoff or recoupment with respect to the Unimpaired Claims.

ARTICLE IV: ACCEPTANCE OR REJECTION OF THE PLAN

- 4.1. <u>Impaired Classes of Claims Entitled to Vote</u>. Holders of Claims in Classes 1C through 9C (Prepetition Credit Agreement Claims against the Lee Debtors) and 10C through 35C (PD LLC Notes Claims against the Pulitzer Debtors) are Impaired and entitled to vote to accept or reject this Plan.
- 4.2. <u>Acceptance by an Impaired Class</u>. In accordance with section 1126(c) of the Bankruptcy Code and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class of Claims shall have accepted this Plan if this Plan is accepted by the Holders of at least two-thirds (¾) in dollar amount and more than one-half (½) in number of the Allowed Claims in such Class that have timely and properly voted to accept or reject this Plan.
- 4.3. Presumed Acceptances by Unimpaired Classes. Classes 1A through 35A (Priority Non-Tax Claims against the Debtors), 1B through 35B (Other Secured Claims against the Debtors), 1D through 35D (General Unsecured Claims against the Debtors), 1E through 35E (Intercompany Claims against the Debtors), 12F and 14F (Herald Claim against PD LLC and DS LLC), and 1G through 35G (Interests in the Debtors) are Unimpaired by this Plan. Pursuant to section 1126(f) of the Bankruptcy Code, the Holders of Claims and Interests in such Classes are conclusively presumed to have accepted this Plan and therefore shall not be entitled to vote to accept or reject this Plan.
 - 4.4. <u>Confirmability and Severability of this Plan</u>.
- 4.4.1 <u>Consensual Confirmation</u>. The confirmation requirements of section 1129(a) of the Bankruptcy Code must be satisfied separately with respect to each Debtor. Therefore, notwithstanding the combination of the separate plans of reorganization for all Debtors in this Plan for purposes of, among other things, economy and efficiency, this Plan shall be deemed a separate chapter 11 plan for each such Debtor.

4.4.2 <u>Reservation of Rights</u>. The Debtors reserve the right to amend, modify, or supplement this Plan for any reason, including, without limitation, in the event that any separate plan for a particular Debtor is not confirmed.

ARTICLE V: MEANS FOR IMPLEMENTATION OF THE PLAN

- 5.1. Non-Substantive Consolidation. This Plan is a joint plan that does not provide for substantive consolidation of the Estates and, on the Effective Date, the Estates shall not be deemed to be substantively consolidated for purposes hereof. Except as specifically set forth herein, nothing in this Plan or the Disclosure Statement shall constitute or be deemed to constitute an admission that any one of the Debtors is subject to or liable for any claim against any other Debtor. Additionally, claimants holding Claims against multiple Debtors, to the extent such Claims are Allowed in each Debtor's case, will be treated as Holders of separate Claims against each applicable Estate for all purposes (including, but not limited to, voting and distributions); provided, however, that no Holder shall be entitled to receive more than payment in full of its Allowed Claim (plus post-petition interest, if and to the extent provided in this Plan), and such Claims will be administered and treated in the manner provided in this Plan. Unless otherwise provided by this Plan or the Confirmation Order, Allowed Claims held against any Debtor shall be satisfied solely from the Cash and other assets of such Debtor and its Estate, provided that, to the extent of any insufficiency, funds or other property may be advanced to the relevant Debtor(s) by (x) with respect to a Lee Debtor, any of the other Lee Debtors and (y) with respect to a Pulitzer Debtor, any of the other Pulitzer Debtors.
 - 5.2. <u>Corporate Governance, Directors, Officers and Corporate Action.</u>
- 5.2.1 <u>Certificate of Incorporation; By-Laws; Limited Liability Company Agreements</u>. On the Effective Date, the Certificate of Incorporation and the By-Laws shall be substantially in the form of <u>Exhibit 1.18</u> and <u>Exhibit 1.16</u>, respectively. Consistent with, but only to the extent required by, section 1123(a)(6) of the Bankruptcy Code, on the Effective Date, the Certificate of Incorporation shall be amended to prohibit the issuance of non-voting equity securities. The certificates or articles of incorporation, by-laws, certificates of formation, limited liability company agreements, or similar governing documents, as applicable, of the other Debtors or Reorganized Debtors shall be amended as necessary to satisfy the provisions of this Plan and the Bankruptcy Code. After the Effective Date, the Reorganized Debtors may amend and restate their certificates or articles of incorporation, by-laws, certificates of formation, limited liability company agreements, partnership agreements or similar governing documents, as applicable, as permitted by applicable law.
- 5.2.2 <u>Directors and Officers of Reorganized Lee Enterprises</u>. Subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, on the Effective Date, the initial directors and officers of Reorganized Lee Enterprises shall be the persons identified in <u>Exhibit 5.2.2</u>. After the Effective Date, the Certificate of Incorporation and the By-Laws, as each may be amended thereafter from time to time, shall govern the designation and election of directors, subject to the applicable restrictions, if any, in the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents.
 - 5.2.3 <u>Directors and Officers of the Reorganized Debtors Other than</u>

Reorganized Lee Enterprises. Subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, on the Effective Date, the initial directors and officers of the Reorganized Debtors other than Reorganized Lee Enterprises shall be the persons identified in Exhibit 5.2.3. After the Effective Date, the certificates or articles of incorporation, bylaws, certificates of formation, limited liability company agreements, or similar governing documents, as applicable, of the Reorganized Debtors other than Reorganized Lee Enterprises, as each may be amended thereafter from time to time, shall govern the designation and election of directors, subject to the applicable restrictions, if any, in the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents.

5.2.4 <u>Corporate Action</u>. On the Effective Date, the amendment of the Certificate of Incorporation, the selection of directors and officers for Reorganized Lee Enterprises and each other Reorganized Debtor, and all other actions contemplated by this Plan shall be deemed authorized and approved in all respects (subject to the provisions of this Plan). All matters provided for in this Plan involving the corporate structure of the Debtors or the Reorganized Debtors, and any corporate action required by the Debtors or the Reorganized Debtors in connection with this Plan, shall be deemed to have timely occurred in accordance with applicable law and shall be in effect, without any requirement of further action by the security holders or directors of the Debtors or the Reorganized Debtors. On and after the Effective Date, the appropriate officers of Reorganized Lee Enterprises and/or the other Reorganized Debtors and members of the boards of directors of Reorganized Lee Enterprises and/or the other Reorganized Debtors shall be authorized and directed to issue, execute and deliver the agreements, documents, securities and instruments contemplated by this Plan in the name of and on behalf of Reorganized Lee Enterprises and/or the other Reorganized Debtors.

5.3. <u>Issuance and Distribution of New Securities</u>.

5.3.1 <u>Issuance of New Common Stock and New PD LLC Notes</u>. On the Effective Date, (i) as authorized by the Certificate of Incorporation, Reorganized Lee Enterprises shall issue shares of New Common Stock equal to fifteen percent (15%) of the total number of issued and outstanding shares of common stock of Reorganized Lee Enterprises, computed on a pre-issuance basis, together with any and all instruments, certificates and other documents required to be issued or distributed pursuant to this Plan in order to effect such issuance of New Common Stock without further act or action under applicable law, regulation, order or rule, and (ii) Reorganized PD LLC shall issue the New PD LLC Notes for distribution in accordance with the terms of the New PD LLC Notes Documents and this Plan. The issuance and distribution of the New Common Stock and the New PD LLC Notes under or in connection with this Plan shall be, and shall be deemed to be, exempt from registration under any applicable federal or state securities laws to the fullest extent permissible under applicable bankruptcy law and non-bankruptcy law, including, without limitation, section 1145(a) of the Bankruptcy Code. Without limiting the effect of section 1145 of the Bankruptcy Code, all documents, agreements and instruments entered into on or as of the Effective Date contemplated by or in furtherance of this Plan shall become effective and binding in accordance with their respective terms and conditions upon the parties thereto. In addition, all of the shares of New Common Stock and the New PD LLC Notes issued under or in connection with this Plan shall be fully paid and non-assessable and freely tradable under section 1145 of the Bankruptcy Code.

5.3.2 <u>Distribution of New Common Stock and New PD LLC Notes</u>. On the

Effective Date, (i) all of the shares of the New Common Stock shall be distributed to the Converting Lenders as provided in Section 3.2.3(c) of this Plan and (ii) the New PD LLC Notes shall be distributed to the Holders of Allowed PD LLC Notes Claims as provided in Section 3.3.3(c) of this Plan and pursuant to the terms of the New PD LLC Notes Documents. Distribution of the New Common Stock may be made by means of book-entry exchange through the facilities of the DTC in accordance with the customary practices of the DTC, as and to the extent practicable. In connection with such book-entry exchange, the Disbursing Agent(s) shall deliver instructions to the DTC instructing the DTC to effect distributions of New Common Stock as provided under this Plan. In the period pending distribution of the New Common Stock to any Converting Lender, such Converting Lender shall be entitled to exercise any voting rights and receive any dividends or other distributions payable in respect of such Converting Lender's New Common Stock and to exercise all other rights in respect of the New Common Stock (so that such Converting Lender shall be deemed for tax and all other purposes to be the owner of the New Common Stock).

5.4. <u>Reporting Requirements Under Securities Exchange Act of 1934</u>. Reorganized Lee Enterprises shall continue to be a mandatory reporting company under section 12 of the Securities Exchange Act of 1934, as amended.

5.5. New Debt Documents.

- 5.5.1 New First Lien Credit Facility Documents. On the Effective Date, the parties to each New First Lien Credit Facility Document shall be deemed to have executed such New First Lien Credit Facility Document and shall become bound thereto, regardless of whether any party actually executes such New First Lien Credit Facility Document.
- 5.5.2 <u>New Second Lien Term Loan Documents</u>. On the Effective Date, the parties to each New Second Lien Term Loan Document shall be deemed to have executed such New Second Lien Term Loan Document and shall become bound thereto, regardless of whether any party actually executes such New Second Lien Term Loan Document.
- 5.5.3 New PD LLC Notes Documents. On the Effective Date, the parties to each New PD LLC Notes Document shall be deemed to have executed such New PD LLC Notes Document and shall become bound thereto, regardless of whether any party actually executes such New PD LLC Notes Document.

- 5.6. <u>Conversion or Repayment of DIP Revolving Facility Claims</u>. On the Effective Date, in accordance with and on the terms set forth in the DIP Revolving Facility Agreement, the DIP Revolving Facility Claims shall either be (x) indefeasibly paid in full in Cash from the Lee Debtors (and the "Commitments" (as defined in the DIP Revolving Facility Agreement) shall be irrevocably cancelled and the "Letters of Credit" (as defined in the DIP Revolving Credit Facility Agreement) shall be returned undrawn) in accordance with Section 2.2 of this Plan or (y) converted into revolving loans and commitments outstanding under the New First Lien Revolving Credit Facility pursuant to the New First Lien Credit Facility Documents.
- 5.7. New Second Lien Term Loan Election. Consenting Lenders that made the New Second Lien Term Loan Election at the time of their execution and delivery of the Lee Support Agreement or Lender Joinder thereto shall receive a percentage interest in the New Second Lien Term Loan and the New Common Stock, in each case computed in accordance with the Lee Support Agreement, as provided in Section 3.2.3 of this Plan. The Ballot for such Consenting Lenders to accept or reject this Plan shall have referenced the applicable amount of the Prepetition Credit Agreement Claims that each such Consenting Lender is anticipated to convert under this Plan into a percentage interest in the New Second Lien Term Loan and the New Common Stock. Holders of Prepetition Credit Agreement Claims that are not Consenting Lenders as of the date on which votes to accept or reject this Plan are solicited shall have been afforded the option on their Ballots to make the New Second Lien Term Loan Election regardless of how they vote on this Plan, provided, however, that if any such Holder makes the New Second Lien Term Loan Election, such election shall be subject to the limitations set forth in Section 3.2.3 of this Plan.
- 5.8. New First Lien Credit Facility. On the Effective Date, the Reorganized Lee Debtors shall be authorized to enter into the New First Lien Credit Agreement as well as any notes, documents or agreements delivered in connection therewith, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith. The New First Lien Lenders shall have valid, binding and enforceable first priority liens on the collateral specified in the New First Lien Credit Facility Documents and, other than as expressly set forth in the New First Lien Credit Facility Documents, such liens shall be senior to any other liens with respect to the Lee Debtors. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New First Lien Credit Facility Documents are granted in good faith as an inducement to the New First Lien Lenders to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New First Lien Credit Facility Documents. The New First Lien Revolving Credit Facility shall be a superpriority revolving credit facility with payment priority over the New First Lien Term Loan, and the provisions of the New First Lien Credit Facility Documents setting forth the payment priority of the New First Lien Revolving Credit Facility shall be fully enforceable in accordance with their terms and the parties to the New First Lien Credit Facility Documents shall be enjoined from avoiding, circumventing, recharacterizing or otherwise frustrating the enforcement of the priority provisions thereof.
- 5.9. New Second Lien Term Loan. On the Effective Date, the Reorganized Lee Debtors shall be authorized to enter into the New Second Lien Term Loan Agreement and the Reorganized Lee Debtors and the Reorganized Pulitzer Debtors shall be authorized to enter into any notes, guarantees, documents or agreements delivered, executed, or entered in connection therewith,

including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith including, without limitation, the New Second Lien Term Loan Documents. The New Second Lien Lenders shall have valid, binding and enforceable liens on the collateral specified in the New Second Lien Term Loan Documents. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New Second Lien Term Loan Documents are granted in good faith as an inducement to the New Second Lien Lenders to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New Second Lien Term Loan Documents.

- 5.10. New PD LLC Notes. On the Effective Date, Reorganized PD LLC shall be authorized to enter into the New PD LLC Notes Agreement and issue the New PD LLC Notes, and the Reorganized Pulitzer Debtors shall be authorized to enter into any documents or agreements delivered in connection therewith, including, without limitation, any documents required in connection with the creation or perfection of liens in connection therewith. Other than as expressly set forth in the New PD LLC Notes Documents, the New PD LLC Notes Documents, which shall be senior to any other liens with respect to the Pulitzer Debtors. The guarantees, mortgages, pledges, liens and other security interests granted pursuant to the New PD LLC Notes Documents and payments made (including, but not limited to, payments of principal of the PD LLC Notes and the Lee Closing Date Payment) with respect to the restructuring of the PD LLC Notes pursuant to, and identified in, the Pulitzer Support Agreement are granted and made in good faith as an inducement to the Holders of New PD LLC Notes to extend credit thereunder and shall be deemed not to constitute a fraudulent conveyance or fraudulent transfer, shall not otherwise be subject to avoidance, and the priorities of such liens and security interests shall be as set forth in the New PD LLC Notes Documents.
- 5.11. Binding Effect. On the Effective Date, except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, all provisions of this Plan, including all agreements, instruments and other documents filed in connection with this Plan and executed by the Debtors or the Reorganized Debtors in connection with this Plan, shall be binding upon the Debtors, the Reorganized Debtors, all Holders of Claims against and Interests in each of the Debtors and such Holder's respective successors and assigns, whether or not the Claim or Interest of such Holder is Impaired under this Plan and whether or not such Holder has accepted this Plan, and all other parties that are affected in any manner by this Plan. All agreements, instruments and other documents filed in connection with this Plan shall be given full force and effect, and shall bind all parties referred to therein as of the Effective Date, whether or not such agreements are actually issued, delivered or recorded on the Effective Date or thereafter and whether or not a party has actually executed such agreement.
- 5.12. Continued Corporate Existence and Vesting of Assets in the Reorganized Debtors. On and after the Effective Date, each of the Reorganized Debtors shall continue to exist as separate entities in accordance with the applicable law in the respective jurisdiction in which they are formed and pursuant to their respective certificates or articles of incorporation (or similar organizational documents) and by-laws in effect prior to the Effective Date, except to the extent such certificates or articles of incorporation (or similar organizational documents) and by-laws are to be amended and/or restated pursuant to the terms of this Plan. Notwithstanding anything to the contrary in this Plan, the

Reinstated Claims against and Interests in a particular Debtor or Reorganized Debtor shall remain the obligations solely of such Debtor or Reorganized Debtor following the Effective Date and shall not become obligations of any other Debtor or Reorganized Debtor by virtue of this Plan, the Chapter 11 Cases, or otherwise. Pursuant to section 1141(b) of the Bankruptcy Code, except as otherwise provided under this Plan, all property of the respective Estate of each Debtor, including all claims, rights and causes of action and any property acquired by the Debtors or the Reorganized Debtors under or in connection with this Plan, together with any property of each Debtor that is not property of its Estate and that is not specifically disposed of pursuant to this Plan, shall revest in the applicable Reorganized Debtor on the Effective Date free and clear of all Claims, Liens, charges, other encumbrances and Interests, provided that such revesting shall not relieve any Reorganized Debtor from such Reorganized Debtor's obligations under and the liens created pursuant to the applicable terms and provisions of the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents. Thereafter, the Reorganized Debtors may operate their businesses and may use, acquire and dispose of property free of any restrictions of the Bankruptcy Code and the Bankruptcy Rules. As of the Effective Date, all property of each Reorganized Debtor shall be free and clear of all Liens and non-Reinstated Claims and Interests, except as specifically provided in this Plan or the Confirmation Order. For the avoidance of doubt, on the Effective Date, the property of each Reorganized Debtor shall be subject to new liens pursuant to the terms of the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents.

- 5.13. <u>Cancellation of Certain Credit and Debt Documents</u>. On the Effective Date, in consideration for the distributions to be made on the Effective Date pursuant to this Plan and except as otherwise provided herein, (a) all Prepetition Credit Agreement Documents, PD LLC Notes, PD LLC Notes Documents, and any other instruments, documents, plans or agreements evidencing or creating any indebtedness or obligations of a Debtor related thereto shall be cancelled, (b) the obligations of any of the Debtors under any Prepetition Credit Agreement Documents, PD LLC Notes, PD LLC Notes Documents, or any other agreements evidencing or creating any indebtedness or obligations of a Debtor that relate to Claims or Interests related thereto shall be discharged pursuant to Section 10.2 of this Plan, and (c) the DIP Revolving Facility shall convert into the New First Lien Revolving Credit Facility; <u>provided</u>, <u>however</u>, that this Section 5.13 shall not apply to Reinstated Claims and Interests, including, but not limited to, Reinstated Intercompany Claims.
- 5.14. Cancellation of Liens. Except as otherwise provided in this Plan, on the Effective Date, in consideration for the distributions to be made on the Effective Date pursuant to this Plan, (a) all Liens, charges, encumbrances and rights related to any Claim or Interest, including, without limitation, those existing under the Prepetition Credit Agreement Documents and the PD LLC Notes Documents, but excluding any Lien securing an Other Secured Claim that is Reinstated pursuant to this Plan, shall be terminated, null and void and of no effect and (b) the Liens, charges, encumbrances and rights related to any Claim or Interest under the DIP Revolving Facility shall secure the obligations under the New First Lien Revolving Credit Facility. The Holders of Secured Claims (other than Other Secured Claims that are Reinstated pursuant to this Plan) shall be authorized and directed to release any collateral or other property of any Debtor (including any Cash collateral) held by such Holder and to take such actions as may be requested by the Debtors (or the Reorganized Debtors, as the case may be) to evidence the release of any Liens, including the execution, delivery, and filing or recording of such release documents as may be requested by the Debtors (or the Reorganized Debtors, as the case may be).

5.15. <u>Additional Transactions Authorized Under this Plan</u>. On or prior to the Effective Date, the Debtors shall be authorized to take any such actions as may be necessary or appropriate to Reinstate Claims or Interests or render Claims or Interests not Impaired, as provided for under this Plan.

ARTICLE VI: TREATMENT OF EXECUTORY CONTRACTS, UNEXPIRED LEASES, INSURANCE POLICIES AND EMPLOYEE BENEFIT PLANS

- 6.1. Assumption of Executory Contracts and Unexpired Leases. On the Effective Date, all executory contracts or unexpired leases of the Debtors will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, unless such executory contract or unexpired lease (i) was previously assumed or rejected by the relevant Debtor(s) or (ii) previously expired or terminated pursuant to its own terms. Entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each executory contract and unexpired lease assumed pursuant to this Article VI shall revest in and be fully enforceable by the respective Reorganized Debtor in accordance with its terms, except as modified by the provisions of this Plan, or any order of the Bankruptcy Court authorizing and providing for its assumption.
- 6.2. <u>Cure of Defaults Under Assumed Executory Contracts and Unexpired Leases</u>. Any monetary amounts by which each executory contract and unexpired lease to be assumed is in default shall be satisfied, pursuant to section 365(b)(l) of the Bankruptcy Code, by payment of the default amount in Cash on the Effective Date or on such other terms as the parties to each such executory contract or unexpired lease may otherwise agree. In the event of a dispute regarding (a) the amount of any cure payments, (b) the ability of the Reorganized Debtors or any assignee to provide "adequate assurance of future performance" (within the meaning of section 365 of the Bankruptcy Code) under the contract or lease to be assumed or (c) any other matter pertaining to assumption, the cure payments required by section 365(b)(l) of the Bankruptcy Code shall be made following the entry of a Final Order resolving the dispute and approving the assumption. Pending the Bankruptcy Court's ruling on such motion, the executory contract or unexpired lease at issue shall be deemed assumed by the relevant Debtor unless otherwise ordered by the Bankruptcy Court.
- 6.3. <u>Assumption of Collective Bargaining Agreements</u>. All Collective Bargaining Agreements shall be deemed to have been assumed by the applicable Debtor(s) party thereto upon the occurrence of the Effective Date. Entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the pertinent Reorganized Debtor's assumption of each Collective Bargaining Agreement to which it is a party for the remaining term of agreement of each such Collective Bargaining Agreement as in effect on the Effective Date, except to the extent that such agreements have already been assumed prior to the Effective Date.

- 6.4. <u>Insurance Policies and Agreements</u>. Insurance policies issued to, or insurance agreements entered into by, the Debtors prior to the Petition Date (including, without limitation, any policies covering directors' or officers' conduct) shall continue in effect after the Effective Date. To the extent that such insurance policies or agreements are considered to be executory contracts, this Plan shall constitute a motion to assume or ratify such insurance policies and agreements, and, subject to the occurrence of the Effective Date, the entry of the Confirmation Order shall constitute approval of such assumption pursuant to section 365(a) of the Bankruptcy Code and a finding by the Bankruptcy Court that each such assumption is in the best interest of each Debtor and its Estate. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments shall be required to cure any defaults of the Debtors existing as of the Confirmation Date with respect to each such insurance policy.
- 6.5. Employee Compensation and Benefit Plans. From and after the Effective Date, each of the Reorganized Debtors shall continue to perform its obligations (whether statutory or contractual) under all employment and severance contracts and all Employee Benefit Plans applicable to its employees, retirees and non-employee directors. In addition, notwithstanding anything in this Plan or the Confirmation Order to the contrary, on the Effective Date, the Pension Plans shall be sponsored by and become obligations of the Reorganized Debtors that sponsored such Pension Plans prior to the Petition Date, with such Reorganized Debtors assuming all statutory obligations with respect to the Pension Plans, and shall otherwise be unaffected by Confirmation of this Plan.
- 6.6. Retiree Benefits and Pension Plans. In furtherance of, and without in any way limiting Section 6.5 of this Plan, from and after the Effective Date, pursuant to section 1129(a)(13) of the Bankruptcy Code, the Debtors shall assume and pay all retiree benefits (within the meaning of section 1114 of the Bankruptcy Code) and contribute to the Pension Plans the amount necessary to satisfy the minimum funding standards under sections 302 and 303 of ERISA, 29 U.S.C. §§ 1082 and 1083, sections 412 and 430 of the Internal Revenue Code, and 26 U.S.C. §§ 412 and 430, if any, relating to the Pension Plans; provided, however, that the Reorganized Debtors may modify such benefits to the extent permitted by applicable law. Nothing in this Plan shall be construed as discharging, releasing, or relieving the Debtors, or their successors, including the Reorganized Debtors, or any party, in any capacity, from any liability imposed under any law or regulatory provision with respect to the Pension Plans or the Pension Benefit Guaranty Corporation ("PBGC"). The PBGC and the Pension Plans will not be enjoined or precluded from enforcing such liability as a result of any provision of this Plan or the Confirmation Order.
- 6.7. <u>Post-Petition Contracts and Leases</u>. All contracts, agreements and leases that were entered into by the Debtors or assumed by the Debtors after the Petition Date shall be deemed assigned by the Debtors to the Reorganized Debtors on the Effective Date.

ARTICLE VII: PROVISIONS GOVERNING DISTRIBUTIONS

- 7.1. <u>Distributions on Account of Claims Allowed as of the Effective Date</u>. Unless the Holder of an Allowed Claim and the Debtors or the Reorganized Debtors agree to a different Distribution Date and except as otherwise provided herein or as ordered by the Bankruptcy Court, distributions to be made on account of Claims that are Allowed as of the Effective Date shall be made on the Effective Date. Any payment or distribution required to be made under this Plan on a day other than a Business Day shall be made on the next succeeding Business Day.
- 7.2. <u>Distributions on Account of Claims that Become Allowed after the Effective Date</u>. Distributions on account of Claims that first become Allowed Claims after the Effective Date shall be made as soon as practicable after the date that such Claim becomes Allowed.
- 7.3. <u>Distribution Date</u>. Distributions to be made on the Effective Date shall be deemed actually made on the Effective Date if made either (a) on the Effective Date or (b) as soon as practicable thereafter, but in no event later than ten (10) Business Days after the Effective Date, except as otherwise provided for herein; <u>provided</u>, <u>however</u>, that, notwithstanding anything to the contrary herein, distributions to (i) Holders of Allowed Prepetition Credit Agreement Claims in Classes 1C through 9C and (ii) Holders of Allowed PD LLC Notes Claims in Classes 10C through 35C shall be made on the Effective Date pursuant to Section 3.2.3(c) and Section 3.3.3(c), respectively, of this Plan.
- 7.4. Interest on Claims. Except as otherwise specifically provided for in this Plan (including, for the avoidance of doubt, with respect to Prepetition Credit Agreement Claims, PD LLC Notes Claims, and General Unsecured Claims), the Confirmation Order or other order of the Bankruptcy Court, or required by applicable bankruptcy or non-bankruptcy law, post-petition interest shall not accrue or be paid on any Claims (other than Secured Claims), and no Holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim.
- 7.5. <u>Distributions by Disbursing Agent(s)</u>. Other than as specifically set forth in this Plan, the Disbursing Agent(s) shall make all distributions required to be made under this Plan. Distributions on account of the PD LLC Notes Claims shall be made in accordance with the New PD LLC Notes Agreement or in accordance with this Plan where the New PD LLC Notes Agreement is silent. The Reorganized Debtors may act as Disbursing Agent or may employ or contract with other Entities to assist in or make the distributions required by this Plan.
- 7.6. <u>Delivery of Distributions and Undeliverable or Unclaimed Distributions</u>. The following terms shall govern the delivery of distributions and undeliverable or unclaimed distributions with respect to Claims.
- 7.6.1 <u>Delivery of Distributions in General</u>. Distributions to Holders of Allowed Claims shall be made at the addresses set forth in the Debtors' records.
 - 7.6.2 <u>Undeliverable and Unclaimed Distributions</u>.
- (a) <u>Holding and Investment of Undeliverable and Unclaimed Distributions</u>. If the distribution to any Holder of an Allowed Claim is returned to the Reorganized Debtors or the Disbursing Agent(s) as undeliverable or is otherwise unclaimed, no further

distributions shall be made to such Holder unless and until the Reorganized Debtors or the Disbursing Agent(s) are notified in writing of such Holder's then-current address.

- (b) Failure to Claim Undeliverable Distributions. Any Holder of an Allowed Claim that does not assert a claim pursuant to this Plan for an undeliverable or unclaimed distribution within one (1) year after the Effective Date shall be deemed to have forfeited its claim for such undeliverable or unclaimed distribution and shall be forever barred and enjoined from asserting any such claim for an undeliverable or unclaimed distribution against the Debtors or their Estates or the Reorganized Debtors or their property. In such cases, any Cash for distribution on account of such claims for undeliverable or unclaimed distributions shall become the property of the Estates and the Reorganized Debtors free of any restrictions thereon and notwithstanding any federal or state escheat laws to the contrary. Any New Common Stock or New PD LLC Notes held for distribution on account of such Claim shall be canceled and of no further force or effect. Nothing contained in this Plan shall require any Disbursing Agent, including, but not limited to, any of the Reorganized Debtors, to attempt to locate any Holder of an Allowed Claim.
- 7.7. Record Date for Distributions. The Reorganized Debtors and the Disbursing Agent(s) will have no obligation to but may, in their sole and absolute discretion, recognize the transfer of, or the sale of any participation in, any Allowed Claim that occurs after the close of business on the Distribution Record Date, and will be entitled for all purposes herein to recognize and distribute only to those Holders of Allowed Claims that are Holders of such Claims, or participants therein, as of the close of business on the Distribution Record Date.
- 7.8. <u>Allocation of Plan Distributions Between Principal and Interest</u>. To the extent that any Allowed Claim entitled to a distribution under this Plan is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, for all income tax purposes, be allocated to the principal amount of the Claim first and then, to the extent that the consideration exceeds the principal amount of the Claim, to the portion of such Claim representing accrued but unpaid interest.
- 7.9. <u>Means of Cash Payment</u>. Payments of Cash made pursuant to this Plan shall be in U.S. dollars and shall be made, at the option and in the sole discretion of the Reorganized Debtors, by (a) checks drawn on or (b) wire transfers from a bank selected by the Reorganized Debtors. Cash payments to foreign creditors may be made, at the option of the Reorganized Debtors, in such funds and by such means as are necessary or customary in a particular foreign jurisdiction.
- 7.10. Withholding and Reporting Requirements. In connection with this Plan and all distributions thereunder, the Reorganized Debtors shall comply with all withholding and reporting requirements imposed by any federal, state, local or foreign taxing authority, and all distributions hereunder shall be subject to any such withholding and reporting requirements. The Reorganized Debtors shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements. All persons holding Claims or Interests shall be required to provide any information necessary to effect information reporting and the withholding of such taxes. Each Holder of an Allowed Claim that is to receive a distribution pursuant to this Plan shall have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding and other tax obligations, on account of such distribution. No distribution shall be made to or on behalf of such Holder pursuant to this Plan unless and until such Holder has made arrangements satisfactory to the

Reorganized Debtors for the payment and satisfaction of such tax obligations.

- 7.11. Setoff and Recoupment. The Reorganized Debtors may, pursuant to sections 553 and/or 558 of the Bankruptcy Code or applicable non-bankruptcy laws, but shall not be required to, set off and/or recoup against any Claim the payments or other distributions to be made pursuant to this Plan in respect of such Claim, or claims of any nature whatsoever that any of the Debtors or the Reorganized Debtors may have against the Holder of such Claim; provided, however, that neither the failure to assert such rights of setoff and/or recoupment nor the allowance of any Claim hereunder shall constitute a waiver or release by any of the Reorganized Debtors of any claim that any of the Debtors or the Reorganized Debtors may assert against any Holder of an Allowed Claim.
- 7.12. <u>Fractional Shares</u>. No fractional shares of New Common Stock shall be distributed. Where a fractional share of New Common Stock would otherwise be called for, the actual issuance shall reflect a rounding up (in the case of more than .50) of such fraction to the nearest whole share of New Common Stock, or a rounding down of such fraction (in the case of .50 or less than .50) to the nearest whole share of New Common Stock. The total number of shares of New Common Stock to be distributed pursuant to this Plan shall be adjusted as necessary to account for the rounding provided for herein.

ARTICLE VIII: PROCEDURES FOR RESOLVING DISPUTED CLAIMS

- 8.1. Objection to and Estimation of Claims. After the Effective Date, only the Reorganized Debtors may object to the allowance of any Claim or Administrative Expense Claim. After the Effective Date, the Reorganized Debtors shall be accorded the power and authority to allow or settle and compromise any Claim without notice to any other party, or approval of, or notice to the Bankruptcy Court. In addition, the Debtors or the Reorganized Debtors may, at any time, request that the Bankruptcy Court estimate any contingent or unliquidated Claim pursuant to section 502(c) of the Bankruptcy Code regardless of whether the Debtors or the Reorganized Debtors have previously objected to such Claim.
- 8.2. <u>No Distributions Pending Allowance</u>. No payments or distributions shall be made with respect to all or any portion of a Disputed Claim unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by Final Order and the Disputed Claim, or some portion thereof, has become an Allowed Claim.
- 8.3. <u>Distributions on Account of Disputed Claims Once They Are Allowed</u>. To the extent that a Disputed Claim ultimately becomes an Allowed Claim, distributions (if any) shall be made to the Holder of such Allowed Claim in accordance with the provisions of this Plan. As soon as practicable after the date that the order or judgment of the Bankruptcy Court allowing any Disputed Claim becomes a Final Order, the Disbursing Agent(s) shall provide to the Holder of such Claim the distribution (if any) to which such Holder is entitled under this Plan as of the Effective Date, without any post-Effective Date interest to be paid on account of such Claim.

8.4. Reinstated Claims and Interests. Notwithstanding anything contained herein to the contrary, nothing shall affect, diminish or impair the Debtors' or the Reorganized Debtors' rights and defenses, both legal and equitable, with respect to any Reinstated Claim or Interest, including, but not limited to, legal and equitable rights of setoff and/or recoupment against the Holders of any Reinstated Claims.

ARTICLE IX: CONFIRMATION AND CONSUMMATION OF THE PLAN

- 9.1. <u>Condition to Confirmation</u>. This Plan shall not be confirmed unless and until the Confirmation Order, in form and substance reasonably satisfactory to the Debtors, the Prepetition Administrative Agent, the Initial Backstop Lenders, the Required Consenting Lenders, and the Required Consenting Noteholders shall have been entered not later than forty-five (45) calendar days after the Petition Date, unless (i) otherwise agreed by the Prepetition Administrative Agent, the Initial Backstop Lenders, the Required Consenting Lenders, and the Required Consenting Noteholders or (ii) waived in accordance with Section 9.3 of this Plan.
- 9.2. <u>Conditions to Effective Date</u>. This Plan shall not become effective and the Effective Date shall not occur unless and until the following conditions shall have been satisfied or waived in accordance with Section 9.3 of this Plan:
- 9.2.1 The condition to confirmation in Section 9.1 of this Plan shall have been either satisfied or waived in accordance with Section 9.3 of this Plan.
- 9.2.2 The Confirmation Order confirming this Plan shall have been entered by the Bankruptcy Court and there shall not be a stay or injunction (or similar prohibition) in effect with respect thereto.
- 9.2.3 The Backstop Commitment Letters shall have become effective (and shall not have thereafter terminated) in accordance with the terms set forth therein and the conditions precedent to the obligations of the Backstop Lenders thereunder shall have been satisfied (including, but not limited to, the payment of fees and expenses).
- 9.2.4 Either (a) the DIP Revolving Facility Claims shall have been indefeasibly paid in full in Cash by the Lee Debtors, the "Commitments" (as defined in the DIP Revolving Facility Agreement) irrevocably cancelled, and the "Letters of Credit" (as defined in the DIP Revolving Facility Agreement) returned undrawn or (b) all conditions for converting the DIP Revolving Facility Claims into revolving loans and commitments outstanding under the New First Lien Revolving Credit Facility shall have been satisfied.
- 9.2.5 The New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, and the New PD LLC Notes Documents shall have become effective in accordance with their respective terms (and all conditions precedent thereunder have been met, including, but not limited to, the payment of fees and expenses) on the Effective Date or shall be deemed to become effective on the Effective Date as a result of the coming into effect of this Plan.
 - 9.2.6 All authorizations, consents, certifications, approvals, rulings, no-

action letters, opinions or other documents or actions required by any law, regulation or order to be received or to occur in order to implement this Plan on the Effective Date shall have been obtained or shall have occurred unless failure to do so will not have a material adverse effect on the Reorganized Debtors.

- 9.3. <u>Waiver of Conditions</u>. Each of the conditions set forth in Sections 9.1 and 9.2 of this Plan may be waived in whole or in part by the Debtors with the written consent of the Prepetition Administrative Agent, the Initial Backstop Lenders, the Required Consenting Lenders, and the Required Consenting Noteholders after notice to the Bankruptcy Court and parties in interest but without the need for a hearing.
- 9.4. <u>Vacation of Confirmation Order</u>. If the Confirmation Order is vacated, (a) this Plan shall be null and void in all respects; (b) any settlement of Claims or Interests provided for hereby shall be null and void without further order of the Bankruptcy Court; and (c) the time within which the Debtors may assume and assign or reject all executory contracts and unexpired leases shall be extended for a period of one hundred twenty (120) days after the date the Confirmation Order is vacated.
- 9.5. <u>Notice of Effective Date</u>. The Debtors shall file with the Bankruptcy Court a notice of the occurrence of the Effective Date within a reasonable period of time after the conditions in Section 9.2 of this Plan have been satisfied or waived pursuant to Section 9.3 of this Plan.

ARTICLE X: EFFECT OF PLAN CONFIRMATION

10.1. Discharge.

- 10.1.1 <u>Discharge of Non-Reinstated Claims</u>. On the Effective Date, the Debtors shall be deemed discharged and released under section 1141(d)(l)(A) of the Bankruptcy Code from any and all Claims not Reinstated under this Plan, including, but not limited to, demands and liabilities that arose before the Effective Date, and all debts of the kind specified in sections 502(g), 502(h) or 502(i) of the Bankruptcy Code. Except as otherwise provided herein (including with respect to all Claims that are Reinstated hereunder) or in the Confirmation Order, all consideration distributed under this Plan shall be in exchange for, and in complete satisfaction, settlement, discharge and release of, all Claims of any nature whatsoever against the Debtors or any of their Estates, assets, properties or interests in property. Notwithstanding anything in this Section 10.1.1 to the contrary, the DIP Revolving Facility shall be converted into the New First Lien Revolving Credit Facility, subject to the terms thereof, including the satisfaction of the conversion conditions set forth therein.
- 10.1.2 <u>Non-Discharge of Interests</u>. All Interests in the Debtors are Reinstated under this Plan. Accordingly, Interests shall not be discharged by this Plan.
- 10.1.3 <u>Discharge Injunction</u>. As of the Effective Date, except as otherwise expressly provided in this Plan or the Confirmation Order, all Entities shall be precluded from asserting against the Debtors or the Reorganized Debtors and their respective assets, property and

Estates, any other or further Claims (other than those Reinstated under this Plan), or any other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities of any nature whatsoever, relating to any of the Debtors or Reorganized Debtors or any of their respective assets, property and Estates, based upon any act, omission, transaction or other activity of any nature that occurred prior to the Effective Date. In accordance with the foregoing, except as expressly provided in this Plan or the Confirmation Order, the Confirmation Order shall constitute a judicial determination, as of the Effective Date, of the discharge of all non-Reinstated Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities, pursuant to sections 524 and 1141 of the Bankruptcy Code, and such discharge shall void and extinguish any judgment obtained against the Debtors, the Reorganized Debtors, or any of their respective assets, property and Estates at any time, to the extent such judgment is related to a discharged Claim, debt or liability.

- 10.2. Releases by the Debtors. Except as otherwise expressly provided in this Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each of the Debtors and Reorganized Debtors on its own behalf and as a representative of its respective Estate, shall, and shall be deemed to, completely and forever release, waive, void, extinguish and discharge unconditionally, each and all of the Released Parties of and from any and all Claims and causes of action (including, without limitation, Avoidance Actions), any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to any of the Debtors, the Reorganized Debtors, their respective assets, property and Estates, the Chapter 11 Cases, the Lee Support Agreement, the Pulitzer Support Agreement, this Plan, or the Disclosure Statement that may be asserted by or on behalf of any of the Debtors, the Reorganized Debtors or their respective Estates against any of the Released Parties; provided, however, that nothing in this Section 10.2 shall be construed to release any Released Party from willful misconduct or gross negligence as determined by a Final Order.
- 10.3. Releases by Certain Holders of Claims. Except as otherwise expressly provided in this Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each Holder of a Claim entitled to vote on this Plan shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the Debtors and their respective Related Persons of and from any and all Claims, any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to any of the Debtors, the Reorganized Debtors or their respective assets, property and Estates, the Chapter 11 Cases, this Plan, and/or the Disclosure Statement; provided, however,

that each Holder of a Claim that has submitted a Ballot may elect, by checking the appropriate box on its Ballot, not to grant the releases set forth in this Section 10.3, unless otherwise bound by the Lee Support Agreement or the Pulitzer Support Agreement to do so, with respect to the Debtors and their respective Related Persons; and <u>provided further</u>, <u>however</u>, that nothing in this Section 10.3 shall be construed to release the Debtors or their respective Related Persons, if any, from willful misconduct or gross negligence as determined by a Final Order. For the avoidance of doubt, nothing in this Section 10.3 shall release the Reorganized Debtors and their respective Related Persons from their obligations under the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, or the New PD LLC Notes Documents.

- 10.4. Exculpation. From and after the Effective Date, the Released Parties shall neither have nor incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, formulating, negotiating or implementing the Lee Support Agreement, the Pulitzer Support Agreement, this Plan, and/or the Disclosure Statement, the solicitation of acceptances of this Plan, the pursuit of Confirmation of this Plan, the Confirmation of this Plan, the consummation of this Plan, the administration of this Plan, the property to be distributed under this Plan, or any other act taken or omitted to be taken in connection with or in contemplation of the Chapter 11 Cases or implementation of this Plan; provided, however, that this Section 10.4 shall not apply to release (x) obligations under this Plan, and obligations under the contracts, instruments, releases, agreements, and documents delivered, Reinstated or assumed under this Plan, and (y) any claims or causes of action arising out of willful misconduct or gross negligence as determined by a Final Order. Any of the Released Parties shall be entitled to rely, in all respects, upon the advice of counsel with respect to their duties and responsibilities under this Plan.
- 10.5. <u>Injunction Related to Exculpation</u>. Except as expressly provided in this Plan or the Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, cause of action or liability of any nature whatsoever, of the types described in Section 10.4 of this Plan and relating to any of the Debtors or the Reorganized Debtors or any of their respective assets, property and/or Estates, are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any Released Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit, action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 10.1 of this Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is

inconsistent with the provisions of this Plan or the Confirmation Order.

- 10.6. <u>Survival of Indemnification Obligations</u>. The obligations of the Debtors to indemnify any past and present directors, officers, agents, employees and representatives, pursuant to certificates or articles of incorporation, by-laws, contracts and/or applicable statutes, in respect of all actions, suits and proceedings against any of such officers, directors, agents, employees and representatives, based upon any act or omission related to service with, for or on behalf of the Debtors, shall not be discharged or impaired by Confirmation or consummation of this Plan and shall be assumed by the Reorganized Debtors.
- 10.7. <u>Term of Bankruptcy Injunction or Stays</u>. All injunctions or stays provided for in the Chapter 11 Cases under section 105 or section 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

ARTICLE XI: RETENTION OF JURISDICTION

- 11.1. Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court shall retain exclusive jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and this Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:
- 11.1.1 allow, disallow, determine, liquidate, classify, estimate or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of any request for payment of any Administrative Expense Claim, Priority Tax Claim, DIP Revolving Facility Claim, or Section 507(b) Claim and the resolution of any objections to the allowance or priority of Claims or Interests;
- 11.1.2 resolve any matters related to the assumption or assumption and assignment of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any Debtor or Reorganized Debtor may be liable and to hear, determine, and, if necessary, liquidate any Claims arising therefrom;
 - 11.1.3 ensure that distributions to Holders of Allowed Claims are accomplished pursuant to the provisions of this Plan;
- 11.1.4 decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving the Debtors that may be pending on the Effective Date;
- 11.1.5 enter such orders as may be necessary or appropriate to implement or consummate the provisions of this Plan and all contracts, instruments, releases and other agreements or documents created in connection with this Plan, the Disclosure Statement or the Confirmation Order, and issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any entity with consummation, implementation or enforcement of this Plan or the Confirmation Order;

- 11.1.6 resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation, or enforcement of this Plan or any contract, instrument, release or other agreement or document that is executed or created pursuant to this Plan, or any entity's rights arising from or obligations incurred in connection with this Plan or such documents;
- 11.1.7 approve any modification of this Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code or approve any modification of the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order, or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, this Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order, in such manner as may be necessary or appropriate to consummate this Plan;
- 11.1.8 subject to Section 12.2 of this Plan, hear and determine all applications for compensation and reimbursement of expenses of Professionals under this Plan or under sections 330, 331, 363, 503(b), 1103 and 1129(c)(9) of the Bankruptcy Code, which shall be payable by the Debtors only upon allowance thereof pursuant to an order of the Bankruptcy Court;
 - 11.1.9 hear and determine causes of action by or on behalf of the Debtors or the Reorganized Debtors;
- 11.1.10 hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code;
- 11.1.11 enter and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked or vacated, or if distributions pursuant to this Plan are enjoined or stayed;
- 11.1.12 determine any other matters that may arise in connection with or relate to this Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release, or other agreement, or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order;
- 11.1.13 enforce all orders, judgments, injunctions, releases, exculpations, indemnifications and rulings entered in connection with the Chapter 11 Cases;
- 11.1.14 hear and determine all matters related to (i) the property of the Estates from and after the Confirmation Date and (ii) the activities of the Reorganized Debtors;
- 11.1.15 hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code; and
 - 11.1.16 enter an order closing the Chapter 11 Cases.
 - 11.2. Notwithstanding anything in this Article XI to the contrary, any disputes arising

under the New First Lien Credit Facility Documents, the New Second Lien Term Loan Documents, any Backstop Commitment Letter, or the New PD LLC Notes Documents will be governed by the jurisdictional provisions thereof.

ARTICLE XII: MISCELLANEOUS PROVISIONS

- 12.1. Surrender of Instruments. As a condition to participation in distributions under this Plan, unless the Debtors shall otherwise agree, each PD LLC Noteholder and the Holder(s) of any evidence of indebtedness of the Debtors relating to a non-Reinstated Claim that desires to receive the property to be distributed on account of an Allowed non-Reinstated Claim based on such PD LLC Note or evidence of indebtedness to the Debtors, or their designee, and shall execute and deliver such other documents as are necessary to effectuate this Plan. Except as otherwise provided in this Section 12.1 (or except as the Debtors otherwise agree), if no surrender of a PD LLC Note or evidence of indebtedness relating to a non-Reinstated Claim occurs and a claimant does not provide an affidavit and indemnification agreement, in form and substance reasonably satisfactory to the Debtors, that such PD LLC Note or evidence of indebtedness was lost, then no distribution may be made to any claimant whose Claim is based on such security, note, debenture or evidence of indebtedness thereof. The Debtors shall make subsequent distributions only to the persons who surrender PD LLC Notes or evidence of indebtedness, as applicable, for exchange (or their assignees) and the record holders of such PD LLC Notes or other indebtedness shall be those holders of record as of the Distribution Record Date.
- 12.2. <u>Post-Confirmation Date Retention of Professionals</u>. On the Effective Date, any requirement that Professionals employed by the Reorganized Debtors comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date will terminate, and the Reorganized Debtors will be authorized to employ and compensate professionals in the ordinary course of business and without the need for application to or approval by the Bankruptcy Court.
- 12.3. Bar Date for Certain Administrative Expense Claims. All applications for final allowance of compensation or reimbursement of expenses incurred by any Professional, and all other requests for the payment of Administrative Expense Claims, including all requests for the allowance of any Administrative Expense Claim pursuant to section 503(b)(3)(D) of the Bankruptcy Code for substantial contributions made in the Chapter 11 Cases (but excluding all requests for the payment of obligations incurred by the Debtors in the ordinary course of their business operations after the Petition Date), must be filed with the Bankruptcy Court and served on the Reorganized Debtors and their counsel at the addresses set forth in Section 12.15 of this Plan not later than thirty (30) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court. Any request for the payment of an Administrative Expense Claim that is not timely filed and served shall be discharged and forever barred and the Holder of such Administrative Expense Claim shall be enjoined from commencing or continuing any action, process, or act to collect, offset or recover such Claim. The Debtors and the Reorganized Debtors shall have sole responsibility for filing objections to and resolving all requests for the allowance of Administrative Expense Claims.

- 12.4. <u>Effectuating Documents and Further Transactions</u>. Each of the Debtors and the Reorganized Debtors is authorized to execute, deliver, file or record such contracts, instruments, certificates, notes, releases and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement and further evidence the terms and conditions of this Plan and the New Common Stock and New PD LLC Notes issued under or in connection with to this Plan.
- 12.5. <u>Corporate Action</u>. Prior to, on, or after the Effective Date (as appropriate), all matters expressly provided for under this Plan that would otherwise require approval of the stockholders or directors of one or more of the Debtors or the Reorganized Debtors shall be deemed to have occurred and shall be in effect prior to, on, or after the Effective Date (as appropriate) pursuant to the applicable general corporation law of the states in which the Debtors or the Reorganized Debtors are incorporated or organized without any requirement of further action by the stockholders or directors of the Debtors or the Reorganized Debtors.
- 12.6. Exemption from Transfer Taxes. Pursuant to section 1146(a) of the Bankruptcy Code, (a) the issuance, transfer or exchange of notes or equity securities under this Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; (c) the making or assignment of any lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under this Plan, including, without limitation, merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any stamp tax or other similar tax.
- 12.7. <u>Payment of Statutory Fees</u>. All fees payable pursuant to section 1930 of title 28 of the United States Code shall be paid as and when due or as otherwise agreed by the United States Department of Justice, Office of the United States Trustee.
- 12.8. Payment of Consenting Lender Fees, Consenting Noteholder Fees, and Backstop Lender Fees. Notwithstanding anything to the contrary in this Plan, the Consenting Lender Fees, the Consenting Noteholder Fees, and the Backstop Lender Fees incurred in connection with the Lee Support Agreement, the Pulitzer Support Agreement, and Backstop Commitment Letters, respectively, shall be paid in full, in Cash, on the Effective Date to the Consenting Lenders, the Consenting Noteholders, and the Backstop Lenders, as applicable, and the Debtors shall pay such other fees and expenses on the Effective Date as may have been agreed to by the Debtors prior to the Effective Date, including, without limitation, the PD LLC Noteholder Fees and any fees or expenses required to be paid under any Backstop Commitment Letter.

- 12.9. <u>Support Agreements</u>. Until the Effective Date, notwithstanding any provision of this Plan to the contrary, nothing in this Plan shall relieve the parties to the Lee Support Agreement and the Pulitzer Support Agreement from any of their obligations thereunder.
- 12.10. Amendment or Modification of this Plan. Subject to section 1127 of the Bankruptcy Code, the Debtors may alter, amend or modify this Plan or the Exhibits at any time prior to or after the Confirmation Date but prior to the substantial consummation of this Plan provided that this Plan and the Exhibits remain in form and substance satisfactory to the Prepetition Administrative Agent, the Initial Backstop Lenders, and the Required Consenting Noteholders. Any Holder of a Claim or Interest that has accepted this Plan shall be deemed to have accepted this Plan, as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim or Interest of such Holder.
- 12.11. Severability of Plan Provisions. If, prior to the Confirmation Date, any term or provision of this Plan is determined by the Bankruptcy Court to be invalid, void or unenforceable, the Bankruptcy Court will have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision will then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of this Plan will remain in full force and effect and will in no way be affected, impaired or invalidated by such holding, alteration, or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of this Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 12.12. <u>Successors and Assigns</u>. This Plan shall be binding upon and inure to the benefit of the Debtors, and their respective successors and assigns, including, without limitation, the Reorganized Debtors. The rights, benefits and obligations of any entity named or referred to in this Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such entity.
- 12.13. Revocation, Withdrawal or Non-Consummation. Subject to certain restrictions and requirements set forth herein (including, without limitation, Section 12.9 of this Plan), in section 1127 of the Bankruptcy Code and in Bankruptcy Rule 3019, the Debtors reserve the right to revoke or withdraw this Plan as to any or all of the Debtors prior to the Confirmation Date and to file one or more subsequent chapter 11 plans. If the Debtors revoke or withdraw this Plan as to any or all of the Debtors or if confirmation or consummation of this Plan as to any or all of the Debtors does not occur, then, with respect to such Debtors, (a) this Plan shall be null and void in all respects, (b) any settlement or compromise embodied in this Plan (including the fixing or limiting to an amount certain any Claim or Interest or Class of Claims or Interests), assumption or rejection of executory contracts or leases affected by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null and void and (c) nothing contained in this Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or any other Person, (ii) prejudice in any manner the rights of such Debtors or any other Person or (iii) constitute an admission of any sort by the Debtors or any other Person.

12.14. <u>Notice</u>. All notices, requests and demands to or upon the Debtors or the Reorganized Debtors to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

LEE ENTERPRISES, INCORPORATED

201 N. Harrison Street, Suite 600

Telephone: 563-383-2179 Facsimile: 563-327-2600 Attn: Carl G. Schmidt

Davenport, IA 52801

with a copy to:

SIDLEY AUSTIN LLP

One South Dearborn Street

Chicago, Illinois 60603

Telephone: (312) 853-7000 Facsimile: (312) 853-7036

Attn: Larry J. Nyhan Attn: Kenneth P. Kansa

-and-

YOUNG CONAWAY STARGATT & TAYLOR, LLP

The Brandywine Building 1000 West Street, 17th Floor

P.O. Box 391

Wilmington, Delaware 19899-0391

Telephone: (302) 571-6600 Facsimile: (302) 571-1253 Attn: Robert S. Brady

Counsel to Debtors and Debtors-in-Possession

- 12.15. <u>Governing Law</u>. Subject to the provisions of any contract, certificates or articles of incorporation, by-laws, instruments, releases, or other agreements or documents entered into in connection with this Plan, and subject further to Section 11.2 of this Plan, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with (i) the Bankruptcy Code, the Bankruptcy Rules or other federal law to the extent applicable and (ii) if none of such law is applicable, the laws of the State of Delaware, without giving effect to the principles of conflicts of law of such jurisdiction.
- 12.16. <u>Tax Reporting and Compliance</u>. The Reorganized Debtors are hereby authorized, on behalf of each of the Debtors, to request an expedited determination under section 505 of the Bankruptcy Code of the tax liability of the Debtors for all taxable periods ending after the Petition Date through, and including, the Effective Date.

- 12.17. Exhibits. All Exhibits to this Plan are incorporated into and are a part of this Plan as if set forth in full herein.
- 12.18. <u>Filing of Additional Documents</u>. On or before substantial consummation of this Plan, the Reorganized Debtors and the Debtors shall, as applicable, file such agreements and other documents as may be necessary or appropriate to effectuate and evidence further the terms and conditions of this Plan.
- 12.19. Reservation of Rights. Except as expressly set forth herein, this Plan shall have no force and effect unless the Bankruptcy Court has entered the Confirmation Order. The filing of this Plan, any statement or provision contained in this Plan, or the taking of any action by the Debtors with respect to this Plan shall not be and shall not be deemed to be an admission or waiver of any rights of the Debtors or any other Person with respect to Claims against and Interests in the Debtors.

Dated: January 19, 2012

Respectfully submitted,

LEE ENTERPRISES, INCORPORATED (for itself and on behalf of the other Debtors)

/s/ Carl G. Schmidt

Name: Carl G. Schmidt

Title: Vice President, Chief Financial Officer and Treasurer

SIDLEY AUSTIN LLP

Larry J. Nyhan Kenneth P. Kansa Bojan Guzina Kerriann S. Mills One South Dearborn Street Chicago, Illinois 60603 Telephone: (312) 853-7000

Facsimile: (312) 853-7036

- and -

YOUNG CONAWAY STARGATT & TAYLOR, LLP

Robert S. Brady (No. 2847) Edwin J. Harron (No. 3396) Edmon L. Morton (No. 3856) The Brandywine Building 1000 West Street, 17th Floor P.O. Box 391

Wilmington, Delaware 19899-0391

Telephone: (302) 571-6600 Facsimile: (302) 571-1253

Counsel to the Debtors and Debtors-in-Possession

<u>Appendix A</u>

Lee Debtors

Debtor Number	Debtor Name
1.	Lee Enterprises
2.	Journal-Star Printing Co.
3.	Accudata, Inc.
4.	INN Partners, L.C.
5.	K. Falls Basin Publishing, Inc.
6.	Lee Consolidated Holdings Co.
7.	Lee Publications, Inc.
8.	Lee Procurement Solutions Co.
9.	Sioux City Newspapers, Inc.

<u>Appendix B</u>

Pulitzer Debtors

Debtor Number	Debtor Name
10.	Pulitzer Inc.
11.	Pulitzer Technologies, Inc.
12.	St. Louis Post-Dispatch LLC
13.	Fairgrove LLC
14.	STL Distribution Services LLC
15.	Suburban Journals of Greater St. Louis LLC
16.	Pulitzer Network Systems LLC
17.	Pulitzer Newspapers, Inc.
18.	Flagstaff Publishing Co.
19.	Hanford Sentinel Inc.
20.	HomeChoice, LLC
21.	Kauai Publishing Co.
22.	Napa Valley Publishing Co.
23.	NIPC, Inc. f/k/a Northern Illinois Publishing Co., Inc.
24.	NVPC LLC
25.	Northern Lakes Publishing Co.
26.	NLPC LLC
27.	Pantagraph Publishing Co.
28.	HSTAR LLC
29.	Pulitzer Missouri Newspapers, Inc.
30.	Pulitzer Utah Newspapers, Inc.
31.	Santa Maria Times, Inc.
32.	SHTP LLC
33.	Southwestern Oregon Publishing Co.
34.	SOPC LLC
35.	Ynez Corporation



201 N. Harrison St. Davenport, IA 52801 www.lee.net

NEWS RELEASE

Lee Enterprises refinancing takes effect January 30

DAVENPORT, Iowa (January 23, 2012) – Lee Enterprises, Incorporated (NYSE: LEE), a leading provider of local news, information and advertising in 52 markets, has successfully completed the remaining steps to enable its refinancing agreements to go into effect on January 30, 2012.

The agreements extend the maturities of Lee's borrowings to December 2015 and April 2017. Implementation required a voluntary, prepackaged Chapter 11 process to bind a small minority of non-consenting lenders to the terms. The plan of reorganization was confirmed today by Chief U.S. Bankruptcy Judge Kevin Gross in the District of Delaware, where Lee is incorporated. Lee is scheduled to emerge from Chapter 11 on January 30, 2012, when the plan of reorganization becomes effective.

"This is the favorable outcome we fully expected, and it provides Lee with a nearly four-year runway to continue improving our balance sheet," said Mary Junck, chairman and chief executive officer. "The completion of our Chapter 11 process, which began December 12, 2011, did not affect employees, vendors, contractors, customers or any aspect of company operations. Stockholders retain their interest in the company with only modest dilution. The refinancing, along with our ongoing strong cash flow, will keep Lee on solid financial footing as we continue reshaping our company and building on our unique strengths. Because of our intensive sales culture and evolving array of products, Lee has outpaced the industry in advertising revenue performance for 33 quarters in a row. Even in a challenging economy, more than 80% percent of adults in our larger markets read or use our print and digital products each week, including two-thirds of 18- to 29-year-olds. All of this reinforces our optimism for the future."

Lee announced in September that its credit facility would be amended and extended beyond its current maturity of April 2012 in a structure of first and second lien debt. The first lien debt consists of a term loan of \$689.5 million, as well as a new \$40 million revolving credit facility that is not expected to be drawn at closing, both of which mature in December 2015. The first lien debt carries interest at LIBOR plus 6.25%, with a LIBOR floor of 1.25%. Interest on the \$40 million revolver is at LIBOR plus 5.5%, with a LIBOR floor of 1.25%. Mandatory amortization payments for the first lien debt total \$5 million for the remainder of our 2012 fiscal year, \$11 million in 2013, \$12.75 million in 2014, and \$13.5 million prior to the final maturity in 2015. The second lien debt consists of a \$175 million term loan with an interest rate of 15% maturing in April 2017. There are no mandatory amortization payments required. Second lien creditors will share in issuance of approximately 6,744,000 shares of Lee Common Stock, amounting to approximately 13% of outstanding shares on a pro forma basis as of the closing date.

Agreement on extending Lee's remaining debt, the Pulitzer Notes, was reached in December. The debt will carry an interest rate of 10.55%, increasing 0.75% in January 2013 and each year thereafter. Adjusted for principal payments and non-cash fees to be paid to noteholders, the amended Pulitzer Notes will have a balance of \$126.4 million. Mandatory amortization payments total \$1.4 million in

2012 and \$6.4 million annually thereafter prior to the final maturity in December 2015. Both the first lien debt and Pulitzer Notes also require principal payments based on calculated excess cash flow and allow for optional repayments.

Carl Schmidt, vice president, chief financial officer and treasurer, said that although the refinancing agreements ultimately received support from approximately 97% of lenders under the credit facility and all Pulitzer Notes lenders, Lee needed the court process to complete the transactions due to the requirement for unanimous approval of certain of the terms of the amended facilities.

Lee Enterprises is a leading provider of local news and information, and a major platform for advertising, in its markets, with 48 daily newspapers and a joint interest in four others, rapidly growing digital products and nearly 300 specialty publications in 23 states. Lee's newspapers have circulation of 1.3 million daily and 1.7 million Sunday, reaching nearly four million readers in print alone. Lee's websites and mobile and tablet products attracted 21.8 million unique visitors in December 2011. Lee's markets include St. Louis, MO; Lincoln, NE; Madison, WI; Davenport, IA; Billings, MT; Bloomington, IL; and Tucson, AZ. Lee Common Stock is traded on the New York Stock Exchange under the symbol LEE. For more information about Lee, please visit www.lee.net.

FORWARD-LOOKING STATEMENTS – The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" for forward-looking statements. This news release contains information that may be deemed forward-looking that is based largely on our current expectations, and is subject to certain risks, trends and uncertainties that could cause actual results to differ materially from those anticipated. Among such risks, trends and other uncertainties, which in some instances are beyond our control, are our ability to generate cash flows and maintain liquidity sufficient to service our debt, comply with or obtain amendments or waivers of the financial covenants contained in our credit facilities, if necessary, and to refinance our debt as it comes due. Other risks and uncertainties include the impact and duration of continuing adverse economic conditions, changes in advertising demand, potential changes in newsprint and other commodity prices, energy costs, interest rates, availability of credit, labor costs, legislative and regulatory rulings, difficulties in achieving planned expense reductions, maintaining employee and customer relationships, increased capital costs, maintaining our listing status on the NYSE, competition and other risks detailed from time to time in our publicly filed documents. Any statements that are not statements of historical fact (including statements containing the words "may", "will", "would", "could", "believe", "expect", "anticipate", "intend", "plan", "project", "consider" and similar expressions) generally should be considered forward-looking statements, which are made as of the date of this release. We do not undertake to publicly update or revise our forward-looking statements.

Contact: dan.hayes@lee.net, (563) 383-2100



201 N. Harrison St. Davenport, IA 52801 www.lee.net **Mary Junck**

Chairman, President and Chief Executive Officer (563) 383-2100

January 23, 2012

Dear Lee Stockholders and Employees:

The comprehensive refinancing of Lee's debt will go into effect on January 30, 2012.

The achievement will mark the successful conclusion of a process that began nearly a year ago in a difficult credit environment. The refinancing, along with our rapid digital growth, huge audiences, intensive sales culture and ongoing strong cash flow, will keep Lee on solid financial footing as we continue reshaping our company and improving our balance sheet.

The agreements extend our existing loan agreements to 2015 and 2017 on reasonable terms that preserve stockholders' ownership interests with only 13% dilution. I am enclosing our news release with additional details.

As I noted in my letter of December 2, 2011, completion of the refinancing required a voluntary, prepackaged Chapter 11 process to bind a small minority of non-consenting creditors to agreements that had been reached with an overwhelming majority of lenders. The process did not affect employees, vendors, contractors, customers or any aspect of company operations.

Lee continues to be listed on the New York Stock Exchange, although the stock price currently remains below the minimum average closing price of \$1 per share. We expect investor sentiment to improve with the conclusion of the refinancing. Regardless, NYSE rules allow the cure period to extend through Lee's annual meeting in March 2012 because, if necessary, Lee could receive shareholder approval to enact a reverse stock split. If Lee stock were to be removed from the NYSE, it would be traded over the counter.

As you may have seen in our earnings report for the first quarter of our fiscal 2012, Lee continues to gain headway in a still-uneven economy. We expect revenue trends to improve slowly in 2012, as we press forward with more digital and print initiatives. Because of our intensive sales culture and evolving array of products, Lee has outpaced the industry in advertising revenue performance for 33 quarters in a row, and we expect to continue extending that string.

Even in a challenging economy, our print and digital media attract huge audiences. Continual, extensive research shows that more than 80% percent of adults in our larger markets read or use our products each week, including two-thirds of 18- to 29-year-olds. We believe the statistics are even stronger in our smaller markets. That strength reflects the unique and indispensable role we have earned in our communities and reinforces our optimism for the future.

With deep appreciation for your continuing support,

losey june

Mary Junck

Chairman, President and Chief Executive Officer

Enc.: News release